Metabolic effects of glycine supplementation in captive cheetahs (*Acinonyx jubatus*)

Kathryn van Boom

u19409542

Supervisor: Prof Tertius Kohn Co-supervisor: Prof Adrian Tordiffe

Thesis presented for the degree: Doctor of Philosophy in Veterinary Science (Paraclinical Sciences) University of Pretoria July 2023

> Department of Paraclinical Sciences Faculty of Veterinary Science University of Pretoria Onderstepoort, Pretoria





The copyright of this thesis vests in the author. No quotation from it or information derived from it is to be published without full acknowledgement of the source. The thesis is to be used for private study or non-commercial research purposes only.

Published by the University of Pretoria (UP) in terms of the non-exclusive licence granted to UP by the author.

TABLE OF CONTENTS

DECLA	ARATION	
DEDIC	ATION6	
PREFA	ACE7	
LIST O	F FIGURES12	
LIST O	F TABLES14	
ABSTR	RACT	
CHAP	rer one	
1. LIT	TERATURE REVIEW	
1.1.	The cheetah (Acinonyx jubatus)17	
1.2.	Cheetahs in captivity	
1.3.	Nutrition and metabolism	
1.4.	Dietary studies in cheetahs	
1.5.	The role of glycine – a simple amino acid27	
1.6.	The metabolomics approach	
1.7.	Aims of the study 34	
CHAPTER TWO		
2. ST	UDY DESIGN AND METHODOLOGY35	
2.1.	Introduction	
2.2.	Ethical approval	
2.3.	Study design and feeding intervention	
2.4.	Sample collection procedure	
2.5.	Blood analyses	
2.6.	NMR analysis 42	
2.7.	Data analysis43	
2.8.	Statistical analysis 44	
CHAP	TER THREE	
3. BC	DDY MEASUREMENTS AND BLOOD DATA48	
3.1.	Introduction	

3.2.	Results			
3.3.	Discussion	52		
CHAPTER FOUR				
4. UR	INE METABOLITES	57		
4.1.	Introduction	57		
4.2.	Results	58		
4.3.	Discussion	67		
4.4.	Summary			
CHAPTER FIVE				
5. SE	RUM METABOLITES	78		
5.1.	Introduction			
5.2.	Results			
5.3.	Discussion	85		
5.4.	Summary			
CHAPTER SIX				
6. GE	INERAL DISCUSSION AND CONCLUSION	95		
6.1.	Methionine associated pathways			
6.2.	Threonine associated pathways	101		
6.3.	Carbohydrate and fat metabolism	102		
6.4.	Glutamic acid associated pathways	104		
6.5.	Study limitations and future directions	106		
6.6.	Conclusion	107		
REFERENCES				
APPENDIX 1				
APPENDIX 2				

DECLARATION

I, *Kathryn van Boom*, hereby declare that the work on which this dissertation/thesis is based is my original work (except where acknowledgements indicate otherwise) and that neither the whole work nor any part of it has been, is being, or is to be submitted for another degree in this or any other university.

I empower the university to reproduce for the purpose of research either the whole or any portion of the contents in any manner whatsoever.

Signature:

Date: 5 July 2023

DEDICATION

This is dedicated to my mom, who saw no part of this journey, but who showed me how to dream and how to work hard to achieve them.

PREFACE

Study motivation

Cheetahs are a vulnerable species and their management and conservation is of global importance. In captivity, cheetahs are prone to a wide range of diseases (such as gastritis and renal disease) that are not seen in their free-roaming counterparts (Munson et al., 1999). It is believed that certain factors linked to captivity, like diet, may play a role in the high prevalence of these diseases. In the wild, cheetahs would eat whole carcass diets rich in collagen due to the bones, skin and connective tissue (Depauw et al., 2013). However, these diets are not always feasible in captive settings which results in most captive facilities predominantly feeding their cheetahs a variety of raw muscle meat, such as chicken, horse and beef (Whitehouse-Tedd et al., 2015). Glycine is a simple, inexpensive, highly palatable amino acid with a range of physiological functions in mammals, most notably its role on collagen biosynthesis (Wang et al., 2013). Anecdotal evidence suggests that dietary glycine supplementation may be beneficial for cheetahs in captivity - however, the metabolic effect of glycine supplementation in cheetahs has not yet been investigated. Therefore, the primary focus of this study was to determine the effect of glycine supplementation on the metabolism of the cheetah through NMR techniques which would provide a holistic and clear picture of the many metabolic pathways involved in their metabolism. This study would provide important information on the effects of glycine which may also have implications in other felid species and wildlife.

Summary of samples collected

A total of 30 blood, urine, *Vastus lateralis* skeletal muscle biopsies and 270 gastric biopsies (9 per cheetah per sample collection) were collected from 10 captive cheetahs during 3 different sample collection periods in 2020. Samples were collected before, during and at completion of the diet intervention. Cheetahs were housed at Cango Wildlife Ranch in Oudtshoorn, Western Cape. Only the results obtained from the urine and blood are reported in this thesis. Ethical approval for this study was obtained from the Animal Research Ethics Committee of the Faculty of Veterinary Science, University of Pretoria and the Animal Research Ethics Committee of the University of the Western Cape.

Structure of the thesis

Chapter one provides a literature review discussing various aspects including the diseases and diets of captive cheetahs, the physiological effects of glycine and the benefits of a metabolomics approach. Chapter two describes the methodology of the study from the study design and sample collection to the data and statistical analysis. Chapter three discusses the body measurement, serum biochemistry and haematology results. Chapter four and five discuss the urine and serum

metabolites, respectively, that were identified with NMR. Lastly, chapter six provides a general discussion of the findings and the conclusion.

Planned outcomes of the study

One article entitled "A cross-over dietary intervention in captive cheetahs (*Acinonyx jubatus*) – investigating the effects of glycine supplementation on blood parameters" has been submitted and is under review by Zoo Bioogy (submitted February 2023 and re-submitted May 2023).

There are two planned publications entitled "The effect of glycine on urine and serum metabolomics in captive cheetahs" and "The effect of glycine on skeletal muscle properties in captive cheetahs" for 2023 with the latter publication utilising additional data not included in this thesis as it was out of the scope.

Author and study leader contributions

The study was conceived by Prof Adrian Tordiffe (co-supervisor) and designed by myself, Prof Tordiffe and Prof Tertius Kohn (supervisor). Funding for the study was obtained by Prof Kohn (NRF) and myself (SAVF). Prof Tordiffe immobilised all of the cheetahs and performed sample collection during immobilisation at Cango Wildlife Ranch, Oudtshoorn. Sample analysis was performed by technicians in the Clinical Pathology laboratory, Department of Companion Animal Clinical Studies, University of Pretoria and the Centre for Human Metabolomics, North West University. I planned the logistics of the study, obtained ethical approval, managed the feeding intervention, assisted in the sample and data collection and performed all data and statistical analyses pertaining to the study. I also wrote this thesis, the submitted journal manuscript and I will write the planned future manuscripts.

Acknowledgements

There are many people I need to thank for their important roles in the completion of this study. Firstly, I would like to thank my supervisor, Prof Tertius Kohn, who has been my mentor since we started working together in 2016. Thank you for all your support and guidance throughout this journey and for showing me what it means to be a thoughtful and passionate researcher. To think that we first met when I was in my second year of my undergraduate degree, and to come full circle, nearly 10 years later, is an absolute privilege. It is a testament to your character as a leader and mentor – I am exceedingly grateful that you have travelled this journey alongside me.

Thank you to my co-supervisor, Prof Adrian Tordiffe, who was instrumental in getting this project off the ground and whose passion for cheetahs and their wellbeing has been a driving force in this field of research. I am grateful for the opportunity to have worked with you and I have learnt so much in the process. Thank you for your expertise and for your assistance with the ethics process, sample collection, data analysis and interpretation and reviewing of this thesis.

Thank you to the staff at Cango Wildlife Ranch in Oudtshoorn, Western Cape for their involvement in this study particularly during the feeding intervention and sample collection phase. I would like to specifically mention Narinda Beukes, Craig Gouws, Tamblyn Williams and Riëtte Koortzen who went above and beyond to ensure the success of this project. It was a pleasure to work with you all.

I would like to express my gratitude to the National Research Foundation and the South African Veterinary Foundation for providing the research funding for this study. I would also like to thank WildCat Nutrition for donating the supplements used in this study. Additionally, I am extremely grateful to the NRF (NRF-DAAD) and the University of Pretoria for providing me with a doctoral scholarship which has enabled me to focus wholeheartedly on this research. I would not have been able to complete this degree without their financial assistance.

Thank you to my fellow students, Dean Herbig and Channen Long, for providing assistance in the data and statistical analysis. As well as Luqmaan Adamson for assisting during the sample collection period. Thank you to Kerryn Mac Dermott for reviewing parts of this thesis and for your encouragement throughout. I would also like to extend my thanks to Madelyn de Wet and Leonie Johnson at the University of Pretoria for all their administrative assistance.

My deepest thanks and appreciation goes to my dad, Garth van Boom. You have been my biggest supporter and fan since day one. Thank you for always being there for me on the good days and bad days and for always taking an interest in my work. Your heart for people has shown me what it means to be a kind, compassionate and loving human being which I hope to emulate my entire life. I love you.

Lastly, I would like to acknowledge and thank God for creating this beautiful world and all the majestic animals that roam it. By your grace, you have guided me along this path and given me the opportunity to explore and understand your creation.





LIST OF KEY ABBREVIATIONS

1,3-DAP	1,3-diaminopropane
a.u	arbitrary unit
AABA	α-aminobutyric acid
Alb	albumin
ALP	alkaline phosphatase
ALT	alanine aminotransferase
ANOVA	analysis of variance
ASICS	Automatic Statistical Identification in Complex Spectra
ATP	adenosine triphosphate
BCFA	branched chain fatty acid
BMI	body mass index
СМР	cytidine monophosphate
CoA	coenzyme A
CSF	cerebrospinal fluid
dH20	distilled water
DHA	dehydroascorbic acid
DHP	dihydropyrimidinase
DMA	dimethylamine
DMSO ₂	dimethyl sulphone
DNA	deoxyribonucleic acid
FDR	false discovery rate
g	gram
g	gravity
G-6-P	glucose-6-Phosphate
GABA	γ-aminobutyric acid
GC-MS	gas chromatography mass spectrometry
GCS	glycine cleavage system
GMP	guanosine monophosphate
HPLC	high performance liquid chromatography
Hz	Hertz
IQR	interquartile range
IU	International unit
IUCN	International Union of Conservation of Nature
К	Kelvin
kDa	kilodalton
kg	kilogram
L	litre
LC-MS	liquid chromatography mass spectrometry
m	metre
Μ	molar
МСН	mean corpuscular haemoglobin
MCHC	mean corpuscular haemoglobin concentration

MCM	methylmalonyl-CoA mutase
MCV	mean corpuscular volume
mg	miligram
MHz	megahertz
min	minutes
ml	mililitre
mm	milimetre
MMA	methylmalonic acid
MPV	mean platelet volume
NH ₃	ammonia
NMR or H-NMR	nuclear magnetic resonance
р	p-value
PC	phosphatidylcholine
PCA	principle component analysis
PDW	platelet distribution width
PE	phosphatidylethanolamine
PLS-DA	partial least squares disciminant analysis
ppm	parts per million
PQN	probabilistic quotient normalisation
PS	phosphatidylserine
Q2	cross-validated sum of squares R2
RBC	red blood cell
RDW	red blood cell distribution width
RNA	ribonucleic acid
rpm	revolutions per minute
S	seconds
SAM	S-adenosylmethionine
SCFA	short chain fatty acids
SD	standard deviation
SHMT	serine hydroxymethyltransferase
THF	tetrahydrofolate
ТМА	trimethylamine
ΤΜΑΟ	trimethylamine oxide
TSP	total serum protein
UDPG	uridine 5'-diphosphate glucose
μg	microgram
μΙ	microlitre
μΜ	micromolar
µmol	micromole
UMP	uridine monophosphate
VIP	variable importance in projection
WBC	white blood cell

LIST OF FIGURES

Figure 1. An overview of the role of fatty acids (green), gluconeogenic amino acids (red) and ketogenic amino acids (blue) in energy production. *Adapted from Garrett and Grisham (2012)*.

Figure 2. A schematic diagram of glycine biosynthesis, degradation and various physiological roles in mammals. *AGT, Alanine:glyoxylate aminotransferase; SHMT, Serine hydroxymethyltransferase; GCS, glycine cleavage system. Adapted from Meléndez-Hevia et al. (2009); Wang et al. (2013).*

Figure 3. Overview of the study methodology including study design, feeding intervention, data analysis and statistical analysis.

Figure 4. A two dimensional PCA (top) and PLS-DA (bottom) scores plot of the metabolites identified in the urine of 10 captive cheetahs at baseline, on the control and glycine supplemented diets. *Each point represents an individual cheetah at baseline (red), control (blue) and glycine supplemented (green) with a corresponding 95% confidence ellipse for each diet.*

Figure 5. The normalised concentration of 10 potentially important urinary metabolites identified in 10 captive cheetahs at baseline (red), on a control diet (blue) and on a glycine supplemented diet (green).

Figure 6. A two dimensional PCA (top) and PLS-DA (bottom) scores plot of the metabolites identified in the serum of 10 captive cheetahs at baseline, on the control and glycine supplemented diets. *Each point represents an individual cheetah at baseline (red), control (blue) and glycine supplemented (green) with a corresponding 95% confidence ellipse for each diet.*

Figure 7. The normalised concentration of 7 potentially important serum metabolites identified in 10 captive cheetahs at baseline (red), on a control diet (blue) and on a glycine supplemented diet (green).

Figure 8. Overview of the important metabolites and their associated pathways identified in the current study.

Supplementary Figure S1. PLS-DA classification for urine and serum of captive cheetahs using different numbers of components where the leave-one-out validation was used.

Supplementary Figure S2. A two dimensional PCA (left) and PLS-DA (right) scores plot of the metabolites identified in the urine of captive cheetahs at baseline (top panel), on the control (middle panel) and glycine supplemented (bottom panel) diets. The combined n=10 group is reflected in blue and the split n=6 group is reflected in red. *Each point represents an individual cheetah with a corresponding 95% confidence ellipse for each group.*

Supplementary Figure S3. A two dimensional PCA (left) and PLS-DA (right) scores plot of the metabolites identified in the serum of captive cheetahs at baseline (top panel), on the control (middle panel) and glycine supplemented (bottom panel) diets. The combined n=10 group is reflected in blue and the split n=6 group is reflected in red. *Each point represents an individual cheetah with a corresponding 95% confidence ellipse for each group.*

LIST OF TABLES

Table 1. Nutrient breakdown of horse muscle meat, vitamin and mineral supplement and glycine supplement.

Table 2. Average daily amount of meat, vitamin and mineral supplement and glycine supplement for male (n = 5) and female (n = 5) cheetahs during the habituation, control and glycine intervention periods.

Table 3. Body weight and measurements of male (n=5) and female (n=5) captive cheetahs at baseline, control and glycine diet.

Table 4. Haematology data of 10 captive cheetahs at baseline, with the control and glycine diet.

Table 5. Serum biochemistry data of 10 captive cheetahs at baseline, control diet and glycine diet.

Table 6. The 30 most abundant metabolites identified by NMR in the urine of 10 captive cheetahs at baseline, on a control diet and on a glycine supplemented diet.

Table 7. Significant metabolites identified by univariate analyses in the urine of captive cheetahs at baseline, on a control diet and glycine supplemented diet (n=10).

Table 8. The (A) PCA loadings, (B) PLS-DA VIP scores and (C) PLS-DA diet regression coefficients for the top 10 urinary metabolites in 10 captive cheetahs.

Table 9. The 20 most abundant metabolites identified by NMR in the serum of 10 captive cheetahs at baseline, on a control diet and on a glycine supplemented diet.

Table 10. The (A) PCA loadings, (B) PLS-DA VIP scores and (C) PLS-DA diet regression coefficients for the top 10 serum metabolites in 10 captive cheetahs.

Supplementary Table S1. ASICS reference library of 190 metabolites.

Supplementary Table S2. Relative concentration of the 112 metabolites identified in the urine of captive cheetahs at baseline, on a control and glycine supplemented diet (n=10).

Supplementary Table S3. The (A) PCA loadings, (B) PLS-DA VIP scores and (C) PLS-DA diet regression coefficients for the top 20 urinary metabolites in 10 captive cheetahs.

Supplementary Table S4. The list of 65 potentially important urinary metabolites according to their frequency in the top 20% in univariate and multivariate analyses.

Supplementary Table S5. Relative concentration of the 48 metabolites identified in the serum of captive cheetahs at baseline, on a control and glycine supplemented diet (n=10).

Supplementary Table S6. The list of 32 potentially important serum metabolites according to their frequency in the top 20% in univariate and multivariate analyses.

Supplementary Table S7. Description of cheetahs included in the study (n=10).

Supplementary Table S8. Significant metabolites identified by univariate analyses in the urine of captive cheetahs at baseline, on a control diet and glycine supplemented diet (n=6).

ABSTRACT

Cheetahs are a vulnerable species and their conservation is of global importance. In captivity, cheetahs are prone to a wide range of gastrointestinal diseases that are believed to be linked to the diets they receive in captivity. In the wild, cheetahs eat whole carcass diets rich in collagen. This is not always feasible in captive settings resulting in most facilities feeding their cheetahs only raw muscle meat, which has lower collagen concentrations. Glycine is a simple amino acid that has many physiological functions in mammals, most notably its role on collagen biosynthesis, where it is the most abundant amino acid present. Anecdotal evidence showed that dietary glycine supplementation may improve gastritis. Thus, this study aimed to determine the effect of glycine supplementation on the metabolism of the cheetah. Ten healthy male and female captive cheetahs housed at Cango Wildlife Ranch in Oudtshoorn, Western Cape were fed either a meat only or glycine supplemented meat diet for four weeks, followed by a 4 week cross-over. Body measurements, urine and blood samples were collected at baseline, four weeks (diet 1) and eight weeks (diet 2). Haematology, serum biochemistry and untargeted ¹H nuclear magnetic resonance analysis was performed followed by the appropriate spectra, data and statistical analyses. The glycine diet resulted in a decreased serum albumin, alkaline phosphatase and total calcium concentration and an increased eosinophils and basophils count compared to the baseline or control diet. Body weight also decreased on the glycine diet which may be due to shifts in energy metabolism when glycine was supplemented. A total of 151 and 60 metabolites were identified in the urine and serum, respectively, in the 10 cheetahs. After data reduction, 10 and 7 metabolites were highlighted as important contributors towards the metabolome differences between the diet groups. The important metabolites identified in urine were dimethyl sulphone, proline, fructose, dimethylamine, trimethylamine, pyroglutamic acid, 1,3-diaminopropane, dihydrothymine, methylmalonic acid and pimelic acid. The metabolites identified in serum were glutamic acid, threonine, α -aminobutyric acid, glucose-6-phosphate, ethanolamine, methionine and propionic acid. These metabolites play various metabolic roles including energy production, immune function, protein and collagen biosynthesis or as products of gut microbiome fermentation. The collagen rich diet fed prior to the study, which was reflected in the baseline samples, had a more profound effect on the identified metabolites compared to the glycine supplemented diet. However, glycine supplementation did influence these pathways and had a direct effect on threonine and methionine related pathways. Specifically that of threonine sparing, pyrimidine biosynthesis and decreasing the production of bacterial fermentation products which may prove to be extremely beneficial in improving gastrointestinal health. This indicates the importance of this simple amino acid in whole body metabolism of cheetahs, especially when dietary collagen is limited. Glycine supplementation may be a simple method to improve the gastrointestinal and overall health of captive cheetahs and future studies should use targeted approaches to further understand the roles of the identified metabolites.

CHAPTER ONE

1. LITERATURE REVIEW

1.1. The cheetah (Acinonyx jubatus)

1.1.1. Background and the need for conservation

The cheetah (Acinonyx jubatus) is a unique felid highly specialised for the rapid pursuit of prey, reaching speeds of 29 m/s (104 km/h), and is the last remaining species of the genus Acinonyx (Durant et al., 2015; Hayward et al., 2006; Sharp, 1997). Adult cheetahs typically weigh between 25 to 65 kg, with males being larger for many morphometric measures compared to females (Marker and Dickman, 2003). Historically, cheetahs occupied a wide variety of habitats ranging from dry forests to grasslands across Africa and southwest Asia (Durant et al., 2015). Sadly, they now only occur in 9% of their historic range with an estimated 7100 adult and adolescent cheetahs distributed across 33 global population groups – of which half occur in a single population spanning six southern African countries (Durant et al., 2017). Cheetahs are currently listed as a vulnerable species on the International Union of Conservation of Nature (IUCN) Red List (Durant et al., 2015). According to Durant et al. (2017), the IUCN threat assessments are based on relatively well monitored populations within protected areas. For cheetahs, 67% of their population occurs on unprotected land which leads to increased threats, such as habitat loss, fragmentation, prey loss, human-wildlife conflict and illegal trade, with population densities rarely exceeding 0.02 individuals/km² (Durant et al., 2017). Additionally, the wide spread and elusive nature of the cheetah makes it extremely challenging to monitor them in unprotected areas and, therefore, declines within populations will go largely undetected. This along with the poor projection of growth rates when using a model of cheetahs located in protected areas, suggests that cheetahs should be categorised as endangered under the IUCN Red List (Durant et al., 2017).

Along with the threats experienced by cheetahs in protected and unprotected areas (such as land loss, prey loss and predation by larger predators) the limited genetic diversity of the cheetah also contributes to their vulnerability (Durant et al., 2017; O'Brien et al., 1985). In a 1985 study, the semen samples from 18 cheetahs were analysed and compared to domestic cats (O'Brien et al., 1985). It was found that cheetahs had lower spermatozoal concentrations and that 71% of all examined spermatozoa were morphologically abnormal compared to 29% in domestic cats. An earlier study investigated 200 isozyme and protein loci in 55 cheetahs and found that they contain up to 100 times less genetic variation than other mammalian species (O'Brien et al., 1983). These studies point to a severe population bottleneck in the cheetah's recent evolutionary history which has led to such high genetic uniformity (O'Brien et al., 1985). This genetic uniformity allows the expression of deleterious recessive alleles and limits adaptive ability to changes in their ecological

niche. Additionally, cheetahs have a notoriously high juvenile mortality rate in captivity which has made breeding difficult (O'Brien et al., 1985). Due to the vulnerability of the cheetah, it is necessary to focus on their management and conservation.

1.1.2 Brief anatomy

The remarkable anatomy and morphology of the cheetah enables it to reach high speeds rapidly. Adaptations include the long, slender legs, high respiratory, cardiovascular capacity, specialised muscle for high acceleration, pointed foot pads, semi- retractile claws, aerodynamic frame and a small skull with large sinuses for airflow (Marker and Dickman, 2003; O'Brien et al., 1985; Williams et al., 1997). The fibre type composition and metabolic enzyme activity of the skeletal muscle in cheetahs also highlights their speed capabilities with a predominance of the fast twitch type IIX fibres, a reliance on the anaerobic pathway and rapid hydrolysis of creatine phosphate for ATP regeneration (Williams et al., 1997).

In a comprehensive study by Hudson et al. (2011a), the hindlimb musculoskeletal anatomy of the cheetah was examined to provide insight into their locomotor capabilities and compared to racing greyhounds. The cheetah had longer hindlimb bones which would lead to increased stride length, a smaller volume of hip extensor muscles potentially indicating that the power for acceleration is not produced by the hip but by the vast back musculature. Lastly, they found that the cheetah had a large psoas muscle (links the lower back to the femur through the pelvis) which the authors believe enable rapid protraction of the hindlimb and resists certain forces generated around the hip during acceleration. In a similar study looking at the forelimb musculature, they found that that the total muscle mass of the forelimbs represented $15.1 \pm 1.2\%$ of the total body mass, while the hindlimbs represented 19.8 ± 2.2% (Hudson et al., 2011b). This may indicate that the forelimbs play a greater role in deceleration, while the hindlimbs accelerate the centre of mass. The cheetah also had longer forelimb bones than the greyhound which has similar implications to the longer hindlimbs. Lastly, the digital flexor muscles were larger which they hypothesise is an adaptation to aid in the protraction of the dew claw to capture prey. These studies highlight the unique anatomy of the cheetah for speed, however, the trade-off is that their smaller stature make them vulnerable to either being killed or for their prey to be stolen by more powerful predators such as lions and hyenas (Hunter et al., 2007; Marker and Dickman, 2003).

1.2. Cheetahs in captivity

1.2.1. History and management

Of the big cats, cheetahs are the easiest to tame and adapt quickly to captive settings. As a result, they have a long history with humans dating back to Samaria 3000 BC (Marker-Kraus, 1997; Woc Colburn et al., 2018). The earliest record of a captive cheetah is in 1829 at the Zoological Society

of London with the first confirmed captive birth being recorded many years later in 1956 at the Philadelphia Zoo (Marker-Kraus, 1997). Breeding in captivity is notoriously challenging due to a combination of their genetic uniformity and high infant mortality rates, with only limited facilities having success with limited males and females (Marker-Kraus, 1997; O'Brien et al., 1985). However, when facilities incorporate correct breeding management programmes this may drastically improve the success and sustainability of breeding in captivity as was demonstrated by the de Wildt Cheetah and Wildlife Centre in South Africa in which 785 cubs were born between 1975 to 2005 with a survival rate of 71% (before 12 months) and 66% (after 12 months) (Bertschinger et al., 2008). According to the latest International Cheetah Studbook (2020), the captive population consists of 1851 individuals in 272 known facilities across 46 countries. The United States of America and South Africa are the largest contributors to the captive population with 486 and 381 individuals, respectively (Marker and Johnston, 2020). These reported numbers have likely increased since the publication in 2020.

The correct management of captive cheetahs is vital in maintaining a healthy population. There are various aspects that need to be considered, including the housing requirements, stress management, restraint and handling, health examinations and hand rearing of cubs (Woc Colburn et al., 2018). Additionally, enrichment programmes that aim to enhance the quality of life by providing stimuli that promote natural and active behaviours are common in these facilities, and include changing the structure or content of enclosures or changing husbandry practices such as feeding times (Quirke and O' Riordan, 2011).

1.2.2. Diseases of captive cheetahs

The understanding of disease is important in maintaining and managing the captive cheetah population. In 1993, a comprehensive pathology survey was conducted in the North American captive cheetah population to determine the prevalence of disease (Munson, 1993). Of the 31 deceased adult cheetahs included in the survey, veno-occlusive disease (fibrosis or occlusion of hepatic sinusoids or veins) directly accounted for the death of 9 individuals and was present in 82% of the cheetahs, while the renal disease, glomerulosclerosis, accounted for the death of 8 individuals and, similarly, had a prevalence of 84%. Chronic lymphoplasmacytic gastritis (inflammation of the gastric muscosa) was present in 91% of the cheetahs and believed to be caused by spiral bacteria (*Gastrospirillum sp.* and *Helicobacter pylori*). These bacteria were also present in a corresponding study investigating the role of bacteria in chronic gastritis (Eaton et al., 1993). The authors concluded that the presence of one or both of these bacteria provided evidence that the cause of gastritis is bacterial. However, this was later refuted as a high prevalence of *Helicobacter sp.* was also found in free-ranging cheetah populations who showed no signs of gastritis (Munson et al., 2005).

In a South African captive cheetah population, gastritis had a prevalence of 100% followed by adrenal cortical hyperplasia (83%), glomerulosclerosis (80%), splenic lymphocytic depletion (73%) and veno-occlusive disease (36%) (Munson et al., 1999). The presence of the same unusual diseases, namely chronic lymphoplasmacytic gastritis, glomerulosclerosis and veno-occlusive disease, in two separate populations indicates that the cause may be due to genetic or environmental factors. Munson et al. (2005) then investigated a population of free-ranging Namibian cheetahs for the same unusual diseases. Interestingly, the free-ranging population was largely disease free with a much lower prevalence of glomerulosclerosis (13%), gastritis (11%) and veno-occlusive disease (8%) compared to their captive counterparts. The absence of these significant health problems in the free-ranging populations indicates that genetics alone cannot be the main factor determining the health and fitness of cheetahs, as the Namibian free-ranging population would form part of the historical founding population for North American and South African captive populations (Marker-Kraus, 1997). The higher prevalence of adrenal cortical hyperplasia in the captive population may indicate a physiological stress response to captivity (Munson et al., 2005, 1999). Munson et al. (2005) postulated that the lack of genetic diversity of the cheetah has limited their capacity to adapt to the captive environment resulting in a chronic stress response that exacerbates disease development by changing normal physiological homeostasis.

The high morbidity and mortality due to renal disease, such as glomerulosclerosis and medullary fibrosis, has been positively associated with age in captivity affecting cheetahs over the age of seven years (Mitchell et al., 2018). Medullary fibrosis was notably more severe in the outer third of the renal medulla where metabolic activity is highest and relative hypoxia is greatest. This led the authors to conclude that chronic stress levels may lead to constant activation of the sympathetic nervous and renin angiotensin aldosterone systems (which may also be linked to a NaCl deficiency in the diet) which diverts blood to the brain, muscle and heart at the expense of the kidney and liver. This may lead to repetitive subclinical hypoxic injury to the renal medullary tubules, contributing to the development of chronic renal disease in cheetahs (Mitchell et al., 2018). Many of the other unusual disease as well as the general disease burden (count of the number of common lesions) has been positively associated with age (Gillis-Germitsch et al., 2017). Interestingly, disease burden and gastritis were only moderately associated with adrenal morphology indicating that it is partly stress related, but that the relationship is likely multifactorial (Gillis-Germitsch et al., 2017).

Various aspects of captivity, including stress, diet, age and lack of physical activity may be contributing to the array of unusual diseases experienced by captive cheetahs compared to their free-ranging counterparts (Gillis-Germitsch et al., 2017; Mitchell et al., 2018; Munson et al., 2005; Woc Colburn et al., 2018). The metabolic and physiological mechanistic response of the cheetah to these various aspects of captivity and disease development is largely unknown.

1.2.3. Stress in captivity

The high prevalence of unusual disease in captive, but not free ranging cheetahs, as well as the high prevalence of adrenal cortical hyperplasia has led researchers to believe that the stress experienced in captivity is a major contributor (Munson et al., 2005, 1999). Chronic or repeated stress exposure leads to persistent adrenocorticotropic hormone (ACTH) release from the pituitary gland which may lead to hypertrophy and hyperplasia of the zona fasciculata of the adrenal cortex, and therefore, an enlarged adrenal cortex has been used as a morphological indicator of chronic stress (Terio et al., 2004). Terio et al. (2004) compared the adrenal gland morphology with the faecal cortisol concentration of captive and free-ranging cheetahs and found that captive cheetahs had a higher cortisol concentration and larger adrenal cortices than free ranging cheetahs. The faecal cortisol concentration was then investigated over 30 days in captive cheetahs moved between "on-exhibit" and "off-exhibit" sites, in which the former allows for public viewing (Wells et al., 2004). Individuals transferred from "off-exhibit" to "on-exhibit" showed a trend towards having higher and more prolonged stress responses compared to those housed "off-exhibit". The same observation was recorded by Terio et al. (2004). However, there was notable individual and daily fluctuation in cortisol concentrations in both of these studies which may indicate that individual animals respond differently to environmental changes (Terio et al., 2004; Wells et al., 2004). This also highlights that blood and faecal glucocorticoid concentrations may be influenced by other factors like age, sex and diet, and does not necessarily show the full extent of physiological stress experienced by these animals (Kirberger and Tordiffe, 2016).

The use of trans-abdominal adrenal gland ultrasonography and blood markers of oxidative damage, has been suggested as alternative methods to determine the stress response (Costantini et al., 2017; Kirberger and Tordiffe, 2016). Kirberger and Tordiffe (2016) found that the adrenal glands were easily identified and measured, with the right adrenal gland slightly more difficult to locate. Therefore, the left adrenal gland should be used in future studies, as measurements of the various adrenal dimensions may accurately reflect long term stressors. An advantage of this method is that it could also provide the opportunity to visualise and detect pathology of other abdominal organs (such as the kidney and liver) during a routine health check. In the past 10 years, there has been an increasing interest in molecular oxidative damage and antioxidant mechanisms of free-ranging animals and its relationship with environmental stressors and important life-history functions (Costantini et al., 2019, 2017). Blood markers of oxidative damage and antioxidant capacity increased when cheetahs were physically restrained in box traps, indicating that these parameters are sensitive to acute stress responses (Costantini et al., 2017). Additionally, as captive cheetahs showed higher levels of oxidative damage markers but no corresponding clinical signs of disease, the authors suggested that the higher level of reactive oxygen metabolites and protein carbonyls may be linked to the high intake of polyunsaturated fatty acids in captivity, which are quickly oxidised to produce free radicals (Costantini et al., 2017; Tordiffe et al., 2016). A later study by the same authors, showed that free ranging female cheetahs had higher levels of oxidative stress than male cheetahs, and male and female leopards occupying the same areas (Costantini et al., 2019). They believed this was due to the cumulative lifetime costs of reproduction for female cheetahs who produce six cubs on average for each reproductive event versus the two cubs produced on average for female leopards. These studies emphasizes the need to use alternative metabolic and physiological tools to uncover unnoticed costs that might be negatively affecting the fitness of threatened species.

1.2.4. Physical activity in captivity

In their natural distribution ranges, cheetahs cover wide areas, with home ranges in excess of 3000 km², while in captivity enclosure sizes can vary depending on the facility with a recommended minimum surface area of 750 m² housing two cheetahs (Durant et al., 2017; Woc Colburn et al., 2018). Therefore, the potential physical activity of free-ranging cheetahs would be markedly higher than their captive counterparts. Lack of physical activity has been stated as a potential contributor to the prevalence of various diseases within the population (Munson et al., 2005). In captive settings, the practise of enrichment would provide cheetahs with the best opportunity to do their natural active behaviours such as running, chasing and exploring (Quirke and O' Riordan, 2011). A common practice at different facilities would be lure-coursing or the "cheetah run" which uses a pulley system to move "prey" (usually a white rag) around a track to promote hunting behaviour (Quirke et al., 2013). Aside from the behavioural and management benefits, the physiological benefit of physical activity in captivity is unknown and should be an area of future research.

1.2.5. Diets in captivity

Cheetahs are obligate carnivores and, in the wild, consume bones, tendons and cartilage on a regular basis (Depauw et al., 2013). Cheetahs generally pursue small to medium sized prey with a similar body mass to themselves, ranging between of 23 to 56 kg, in order to optimise their success rate, limit injury and consume their meal before kleptoparasites arrive (Hayward et al., 2006). This makes them very efficient hunters with over 50% of pursuits ending in kills (Bissett and Bernard, 2007; Hayward et al., 2006). Common prey species include impala, blesbok, Grant and Thomson gazelle, springbok and common duiker (Hayward et al., 2006). Male coalitions will generally take on larger prey such as kudus, while single females or those with cubs will pursue smaller antelope such as common duiker or springbok (Bissett and Bernard, 2007).

In a global survey conducted by Whitehouse-Tedd et al. (2015) encompassing 86 facilities representing 12% of the captive cheetah population, it was found that a raw meat diet (37%) was the most popular diet. This was followed by a commercially prepared diet (20%) and carcass diet (8%), with 35% of cheetahs receiving a mixture of the three diets. Interestingly, commercially prepared diets were only provided in North American facilities, while raw meat was available

globally. This was also observed in an earlier study where 13 North American facilities fed on average 1.3 ± 0.4 kg per day commercially prepared frozen canine or feline diets (Dierenfeld, 1993). Additionally, whole or partial carcasses of rabbits, venison and chickens were fed once or twice a week. Raw meat diets mostly consisted of muscle meat (beef, horse, chicken etc.) with a carcass component (such as hide, skin, bone, fur, feather, feet or wing) included at least once a month (Whitehouse-Tedd et al., 2015). Vitamin and mineral supplementation were provided to 53% of captive cheetahs. The raw meat diet had a greater association with firm and dry faeces (considered "normal") while the carcass only diet was linked to extremely dry faeces – this may provide an indicator of gastrointestinal health (Whitehouse-Tedd et al., 2015). The different diets in captivity may also play a role in psychological wellbeing and enrichment, with carcass diets keeping cheetahs active and interested for longer periods of time compared to commercial diets (Bond and Lindburg, 1990).

With high variability in diet seen across free-ranging and captive cheetahs, facilities and continents, and the potential link to gastrointestinal and other disease, it becomes increasingly important to further understand the role of each dietary component on cheetah health.

1.3. Nutrition and metabolism

There have been comprehensive studies done in the domestic cat, and therefore, they are often used as a model of nutrient requirements in exotic felids. Indeed, when domestic cats and large exotic felids (including the cheetah, jaguar, Malayan and Amur tigers) where fed the same beef or horse based diet, they had very similar digestibility responses indicating that the domestic cat is an appropriate model (Vester et al., 2010). An overview of the amino acid and fatty acid energy production pathways are shown in Figure 1.

1.3.1. <u>Protein</u>

As a strict carnivore, dietary protein makes up a high percentage of the captive cheetahs diet with a crude protein range of 55 to 86% depending on the meat source and additional supplementation (Bechert et al., 2002; Depauw et al., 2012b; Dierenfeld, 1993). An animal's protein requirement is a reflection of the minimal nitrogen and amino acid requirement, most of which are used for energy production (MacDonald et al., 1984). Regardless of whether cats are fed a high or low protein diet, there is very little adaptation of aminotransferase or urea cycle enzymes, therefore they continue to use protein for energy production or in other metabolic pathways even with limited availability (Zoran, 2002). Virtually the same amino acids are essential for all species – arginine, histidine, isoleucine, lysine, leucine, methionine, phenylalanine, threonine, tryptophan and valine (MacDonald et al., 1984). However, the cat requires specific amino acids (taurine, arginine, methionine and cysteine) in their diet which are not produced endogenously (Zoran, 2002). A single meal devoid of arginine can lead to hyperammonemia within a few hours. The increased sensitivity



Figure 1. An overview of the role of fatty acids (green), gluconeogenic amino acids (red) and ketogenic amino acids (blue) in energy production. Adapted from Garrett and Grisham (2012).

to arginine deficiency is likely due to the lower ornithine biosynthesis (from glutamate) and less citrulline provided to the kidney for conversion to arginine (MacDonald et al., 1984). A dietary supply of taurine is also required, as inadequate amounts are produced from methionine and cysteine precursors, and a deficiency will lead to central retinal degeneration (MacDonald et al., 1984; Zoran, 2002). In addition to its conversion to taurine, cats have a higher requirement of cysteine and methionine because these gluconeogenic amino acids are catabolised to pyruvate and succinyl-CoA, respectively, in the citric acid cycle (Zoran, 2002). Tryptophan has limited movement through the quinolinic acid pathway and therefore does not synthesise enough niacin (vitamin B_3 or nicotinic acid) in the cat (MacDonald et al., 1984). Tordiffe et al. (2019) used an untargeted exploratory gas chromatography mass spectrometry (GC-MS) and liquid chromatography mass spectrometry (LC-MS) approach to determine the serum and urine amino acid profiles of captive cheetahs. In serum, 38 amino acids were identified with glutamine (1624 µmol/L), alanine (772 µmol/L), arginine (670 µmol/L) and glycine (479 µmol/L) having the highest concentrations. Similarly, 38 amino acids were identified in urine with arginine (740 µmol/L), glutamine (437 µmol/L), alanine (397 µmol/L) and serine (331 µmol/L) having the highest concentrations.

1.3.2. Carbohydrates and fats

Carbohydrate ingestion in cheetahs is extremely low and would make up a negligible portion of any commercial, carcass or supplemented meat based diet (Bechert et al., 2002; Dierenfeld, 1993). However, when fed carcass diets, connective tissue may contribute to the dietary carbohydrate portion as collagen covalently binds glucose and galactose (Depauw et al., 2012a). Excess

carbohydrates that are not enzymatically digested in the small intestine will provide substrates for increased microbial fermentation in the large intestine (Depauw et al., 2012a; Zoran, 2002). Domestic cats lack salivary amylase to initially breakdown the ingested carbohydrates, have low hepatic glucokinase activity to phosphorylate glucose to glucose-6-phosphate and have low glycogen synthase activity to synthesise glycogen from glucose (Zoran, 2002).

After protein, fat makes up a substantial proportion of the cheetah's diet with a range of 9% to 26% crude fat depending whether the diet is meat or carcass based (Bechert et al., 2002; Depauw et al., 2012b). The three major functions of dietary fat is (1) as a concentrated energy source, (2) to provide essential fatty acids and (3) as a carrier for fat soluble vitamins (MacDonald et al., 1984). Fatty acids are either saturated or unsaturated and consists of a carbon chain between 2 to 36 carbons long which can be synthesised in the cytoplasm by NADPH and Acetyl-CoA-carboxylase (Tvrzicka et al., 2011). Their degradation in the mitochondria by β -oxidation leads to the production of ATP for energy. Saturated fats do not contain any double bonds and vary from short chain (including acetic acid, butyric and propionic acid) to long chain (including palmitic and stearic acids). Unsaturated fatty acids, which contain double or triple bonds, can be broken into monounsaturated and polyunsaturated fatty acids (Tvrzicka et al., 2011). The latter contains the essential fatty acids α-linolenic acid and linoleic acid, which are then further elongated and desaturated to form eicosapentaenoic acid, docosahexaenoic acid and arachidonic acid. Cats, and likely cheetahs, have limited Δ -6-desaturase activity and other hepatic desaturase activity and are unable to produce arachidonic acid which is a dietary essential fatty acid in cats (MacDonald et al., 1984). Tordiffe et al. (2016), investigated the serum fatty acid profile of free-ranging and captive cheetahs using GC-MS and found that there was a decreased concentration of monounsaturated, polyunsaturated and total fatty acids in free-ranging cheetahs compared to their captive counterparts. This noticeable profile difference is likely due to differences in dietary fatty acid composition and fatty acid metabolism.

1.3.3. Vitamins and minerals

The fat soluble vitamins A and D are found in animal tissue (such as the liver) and their content should be met in the natural carnivorous or meat based diets of the cheetah (Zoran, 2002). The water soluble vitamin, niacin or nicotinic acid, cannot be synthesised in the cat as they cannot convert tryptophan via quinolinic acid to niacin and it is therefore required in the diet (MacDonald et al., 1984). They also require thiamine, riboflavin, pantothenic acid, pyridoxine, biotin, folacin and vitamin B₁₂ from the animal tissue in their diet. The mineral requirement (e.g. magnesium, potassium, sodium, zinc etc) would be fairly similar across species and should be obtained from the diet (MacDonald et al., 1984). Lack of bones in the captive diet may lead to calcium and phosphorous deficiencies which would require additional supplementation.

1.4. Dietary studies in cheetahs

An early study into the dietary composition of commercial feline diets fed to captive cheetahs in North America, found that these diets had two major isoflavones present which were absent in chicken and horse meat diets (Setchell et al., 1987). Daidzein and genistein, which have structural similarities to steroidal oestrogens and synthetic steroids, were found in the soybean component of the commercial diet. The authors hypothesised that cheetahs may be sensitive to these dietary oestrogens due to the poor hepatic conjugation of xenobiotics and phenolic compounds which is crucial for inactivation and excretion, and this may account for the poor reproductive success of captive cheetahs (Setchell et al., 1987).

Since 1987, there have been a number of studies investigating the nutritional composition of the various diets that captive cheetahs are fed in order to elucidate the potential cause of the high disease rate in captivity (Bechert et al., 2002; Depauw et al., 2013, 2012b, 2012a; Gosselin et al., 1989; Lane et al., 2012; Vester et al., 2008). Gosselin et al. (1989) compared the nutritional composition of eight different commercially prepared diets, ranging from horse meat, chicken, beef and different combinations of vitamins and minerals, using high performance liquid chromatography (HPLC) and GC-MS. The authors noted that the vitamin A levels fluctuated significantly between diets where they were either excessive or deficient based on the recommended dosage and may be a contributor to the veno-occlusive disease seen in captive cheetahs. Similarly, Bechert et al. (2002) compared a commercial diet with five supplemented vitamin and mineral muscle meat diets (cattle, horse, donkey, turkey and deer). The commercial diet had one to five times the level of recommended nutrients, notably vitamin A, while overall the supplemented muscle diets were deficient in vitamin E and copper. The protein to fat ratio was 6:1 for the supplemented meat diets (68 - 84% crude protein), while the ratio for a whole carcass meal is reportedly 3:1 - this high protein content may account for the elevated blood urea nitrogen (BUN) and creatinine concentrations (Bechert et al., 2002). The supplemented meat diets were also lower in fat, and authors suggested that they may require additional essential fatty acid supplementation. Plasma zinc was the only significantly different nutrient between cheetahs, with the supplemented meat diet yielding higher concentrations. These results were also echoed in a later study where cheetahs fed a supplemented beef diet had a higher zinc and lower vitamin A concentration than individuals fed a whole rabbit diet (Depauw et al., 2012b). Although the whole rabbit diet had a higher fat content, both diets led to a deficiency in essential fatty acids, such as linoleic acid and arachidonic acid. Another concern was the high serum levels of vitamin A from the whole rabbit diet, likely due to consumption of the liver, after just four weeks of the diet. The authors concluded that the exclusive use of a single long term food source is not recommended for captive cheetahs.

The diet of captive cheetahs generally consists of high amounts of protein. Excess protein in the diet may result in increased amounts of undigested protein which can lead to bacterial fermentation

in the intestine producing more detrimental by-products such as ammonia, indolic and phenolic compounds (Depauw et al., 2013; Vester et al., 2008). Vester et al. (2008) investigated the nutrient digestibility, faecal characteristics and faecal fermentation end product concentrations of five exotic captive felids (bobcat, jaguar, cheetah, Siberian tiger and Indochinese tiger) fed the same commercially prepared beef diet. The cheetah had a lower fat digestibility and higher concentrations of faecal ammonia and indole compared to the other species. This highlights that different diets should be considered for different exotic felids in captivity, but also that captive cheetahs produce greater concentrations of putrefactive or detrimental by-products compared to their felid counterparts. Similarly, in a cross-over study, cheetahs fed a supplemented beef diet had a higher faecal concentration of short chain fatty acids (SCFA; propionic acid and butyric acid), branched chain fatty acid (BCFA; isobutyrate and isovalerate) and putrefactive compounds (indole, phenol and p-cresol) compared to a whole rabbit diet (Depauw et al., 2013). This was somewhat surprising as the connective tissue found in the rabbit diet is considered to be poorly digestible thus leading to a higher fermentation rate and more putrefactive compounds. The authors concluded that the indigestible parts of the rabbit (bone, cartilage and hair) may act as a bulking agent in the intestine creating a barrier between the bacteria and the substrate which they termed "animal fibre". When investigating the in vitro kinetic fermentation rates of different animal substrates, glucosamine and glucosamine-chondroitin (saccharides that make up cartilage) were highly fermentable (Depauw et al., 2012a). This was followed by the moderately fermentable chicken cartilage and collagen, while rabbit hair, skin and bone were poorly fermentable. Therefore, the presence or absence of certain animal tissues may cause changes in digestion and intestinal fermentation which could play a role in the gastrointestinal health of the cheetah (Depauw et al., 2013, 2012a). Indeed, cheetahs fed a commercial diet over a two year period took longer to develop gastritis or renal disease than cheetahs fed a supplemented horse meat diet (Lane et al., 2012). Although the diet and health status of the cheetahs were variable and no serum or faecal analysis was performed, this study still highlights the need to further understand the role of diet in disease progression and overall health of the cheetah.

1.5. The role of glycine – a simple amino acid

Glycine is the simplest amino acid and is traditionally classified as "nutritionally non-essential" as it is synthesised endogenously in the body (Wang et al., 2013). It was first isolated in 1820 by French Chemist Braconnot and named after its sweet taste. Glycine plays a key role in the metabolism, cytoprotection, immune response, growth and development of many mammals and humans, and represents 11.5% of the total amino acid content in humans (Razak et al., 2017). An overview of glycine biosynthesis, degradation and its various physiological roles are provided in Figure 2.



Figure 2. A schematic diagram of glycine biosynthesis, degradation and various physiological roles in mammals. AGT, Alanine:glyoxylate aminotransferase; SHMT, Serine hydroxymethyltransferase; GCS, glycine cleavage system. Adapted from Meléndez-Hevia et al. (2009); Wang et al. (2013).

1.5.1. Glycine biosynthesis

Glycine is synthesised endogenously via four pathways. Firstly, glycine can be synthesised from serine obtained from the diet or produced endogenously via glucose and glutamate (Wang et al., 2013). Serine hydroxymethyltransferase (SHMT), which catalyses the reaction, is found in the cytosol and mitochondria of cells. However, the mitochondrial SHMT is ubiquitous in the body and responsible for most of the conversion of serine to tetrahydrofolate which yields glycine and N5-N10-methylene tetrahydrofolate. Secondly, glycine can be synthesised from threonine via threonine dehydrogenase (Wang et al., 2013), but this is a minor source of glycine and is dependent on the dietary intake of threonine.

During the oxidative degradation of choline, an essential nutrient, to betaine, the three methyl groups of choline are released and involved in other reactions leading to the production of glycine (Wang et al., 2013). Firstly, betaine is used as the methyl donor converting homocysteine into methionine which produces dimethylglycine. Dimethylglycine is then converted to sarcosine (N-methylglycine) by dimethylglycine dehydrogenase, before being converted to glycine by sarcosine dehydrogenase. Sarcosine and glycine are interconvertible and this is regulated by sarcosine dehydrogenase. Although 40 to 70% of ingested choline may be converted to glycine (which is the case in adult rats), generally, the choline content of diets are quite low in animals and are particularly important when the dietary methionine is minimal (MacDonald et al., 1984; Wang et al., 2013).

The final pathway for glycine biosynthesis is via hydroxyproline which is formed from the post translation hydroxylation of proline in collagen biosynthesis (Li and Wu, 2018). In the mitochondria, hydroxyproline is broken down to glyoxylate which undergoes transamination via Alanine:glyoxylate aminotransferase (AGT) to form glycine and pyruvate (Wang et al., 2013). It is evident that glycine is an important amino acid as there are multiple pathways to synthesise it. However, evidence has shown that the amount of glycine synthesized *in vivo* is not enough to meet metabolic demands in humans, pigs and rodents (Wang et al., 2014, 2013).

1.5.2. Glycine degradation

The catabolism of glycine mainly takes place in the small intestine and is a preferred substrate for many intestinal bacteria such as *Escherichia coli* (Dai et al., 2011). The glycine cleavage system (GCS) is a complex enzyme system consisting of three proteins and one carrier protein, found in the mitochondria of most species. This complex is responsible for the deamination and decarboxylation of glycine producing CO₂, NH₃ and tetrahydrofolate. This reaction couples with SHMT, as tetrahydrofolate is an essential cofactor, to convert glycine into serine which is a reversible reaction (Wang et al., 2013). Therefore, glycine can either be converted to serine or degraded to CO₂ and NH₃ via GCS. GCS favours the oxidation of glycine and not its synthesis, and is crucial when metabolic acidosis, high protein diets, and glucagon lead to an increased glycine degradation (Razak et al., 2017). Lastly, glycine can be converted to glyoxylate through D-amino acid oxidase, although this degradation process would be highly insignificant (Wang et al., 2013).

1.5.3. Physiological benefits of glycine

Glycine has many physiological functions. The most notable is its role, along with proline and hydroxyproline, in the biosynthesis of collagen (Li and Wu, 2018). Collagen is the largest and most abundant protein in the body, accounting for one third of the total proteins in animals, with glycine contributing to every third position of the collagen triple helix. Collagen provides the strength, rigidity and flexibility of connective tissue found in bones, skin, cartilage and blood vessels, and is therefore crucial in the growth, development and health of all animals (Li and Wu, 2018).

Glycine also provides flexibility at the active sites of many enzymes, generates glutathione (major antioxidant), creatine (involved in muscle and nerve metabolism), purines (DNA synthesis), haem (oxygen transport) and serine (Wang et al., 2013). Creatine or phosphocreatine is an important rapid energy provider in tissue such as skeletal muscle, cardiac muscle, brain, the retina and spermatozoa (Wyss and Kaddurah-Daouk, 2000). The high speed lifestyle of cheetahs undoubtedly adds to the importance of creatine as a fuel source (Williams et al., 1997). The transfer of the amidino group of arginine to glycine yielding L-ornithine and guanidinoacetic acid is the first step in the biosynthesis of creatine and is considered the rate-limiting step of creatine and phosphocreatine

are the main energy providers for spermatozoa. Since creatine biosynthesis is dependent on glycine, the latter may have an impact on semen motility and, therefore, reproductive ability.

Glycine also plays a role in the digestion and absorption of lipids and lipid-soluble vitamins through the conjugation of bile acids. Also, glycine regulates the production of cytokines and acts as a neurotransmitter in the central nervous system (Wang et al., 2013). Lastly, several xenobiotic organic acids (such as benzoic acid and phenylacetic acid which are associated with neurological symptoms) are metabolised by conjugation to glycine in the mitochondria which allows them to be excreted from the body (Badenhorst et al., 2014).

1.5.4. Dietary glycine supplementation

As noted above glycine has many physiological and metabolic benefits, and chronic glycine shortage could lead to suboptimal growth, impaired immune response and other adverse metabolic effects (Wang et al. 2013). Meléndez-Hevia et al. (2009) used available literature to calculate various rates of glycine biosynthesis and consumption in humans, pigs, rats and rabbits and extrapolated it to an average 30 to 50 year old 70 kg adult human. They showed that the endogenous biosynthesis of glycine accounts for ± 3 g/day, and dietary glycine accounts for 1.5 to 3 g/day. This leaves a 10 g/day deficit for all the metabolic uses of glycine, including collagen biosynthesis. Therefore, glycine should be considered a "conditionally essential" amino acid with the appropriate dose of supplemented glycine generally considered safe and beneficial for humans and animals (Wang et al., 2013). Animal studies have used up to 3% glycine supplementation while clinical studies in humans have ranged between 3 g per day to 0.8 g/kg body weight per day with no adverse side effects which is likely due to the efficient absorption of glycine (McCarty et al., 2018; Petzke et al., 1986). Additionally, glycine is inexpensive, soluble and highly palatable due to its sweet taste, so it can easily be added to beverages or food (McCarty et al., 2018).

The effects of dietary glycine supplementation has been studied *in vivo* in rats, pigs, chickens, fish, rabbits and humans (Imenshahidi and Hossenzadeh, 2022; Li and Wu, 2018; McCarty et al., 2018; Wang et al., 2014). Briefly, glycine supplementation has been shown to increase plasma glycine and serine concentrations, decrease plasma ammonia, urea and glutamine concentrations, and increase oxidative defence in piglets (Wang et al., 2014). A high glycine diet in rats led to higher SHMT activity and CO₂ production from glycine catabolism, which indicates that glycine catabolism is dependent on and adapted to the glycine diet content (Petzke et al., 1986). In elderly humans, cysteine and glycine supplementation (which are the pre-cursors for glutathione synthesis) led to a 95% increase in glutathione concentration and a decrease in oxidative stress markers, which indicates that supplementation is a practical way to reduce oxidative stress caused by ageing (Sekhar et al., 2011). Plasma glycine concentrations are lower in individuals with metabolic syndrome (hypertension, diabetes etc.) while lifestyle and exercise interventions lead to an increase

in circulating glycine concentrations, indicating that glycine supplementation may improve various components of metabolic syndrome (Imenshahidi and Hossenzadeh, 2022). Supplemental glycine also effectively protects skeletal muscle in a variety of wasting models potentially due to receptor mediated response and modulation of intracellular metabolism (Koopman et al., 2017). There are a wide array of health benefits linked to glycine supplementation – controlling metabolic syndrome and diabetes, inhibiting atherosclerosis, reducing oxidative damage, improving sleep quality, collagen biosynthesis and promoting a healthy and optimum metabolism (Imenshahidi and Hossenzadeh, 2022; McCarty et al., 2018; Meléndez-Hevia et al., 2009).

While glycine supplementation has been investigated in a variety of smaller animals and humans, the effects have not been investigated in felid species such as the cheetah. Tordiffe et al. (2017) showed that a third of the 30 most abundant organic acids excreted in the urine of cheetahs is associated with phenylalanine, tyrosine and glycine conjugates. Additionally, cheetahs are prone to a variety of unusual gastrointestinal diseases in captivity which may be associated with their high protein diet. Excess protein in the diet or incomplete protein digestion can lead to bacterial fermentation in the intestine producing detrimental by-products such as ammonia, indolic and phenolic compounds (Depauw et al., 2013; Vester et al., 2008). Glycine plays a vital role in the conjugation of these compounds to allow urinary excretion (Badenhorst et al., 2014). This, together with their apparent need for large quantities of creatine, suggests that cheetahs have a substantial metabolic need for glycine which may not be met through endogenous biosynthesis or conventional captive diets as has been found in other species (Meléndez-Hevia et al., 2009). However, the metabolic implications of glycine supplementation in captive cheetahs are still unknown and warrant further investigation.

1.6. The metabolomics approach

Metabolomics is a rapidly evolving field of research that uses analytical chemistry techniques to comprehensively characterize the metabolome (Wishart, 2019). The metabolome is generally defined as the complete collection of metabolites found in a specific organelle, cell, organ, biofluid or organism. Metabolites are endogenous compounds such as lipids or amino acids that are produced by endogenous catabolism or anabolism. Although the metabolome is highly conserved across populations and species, an individual's metabolome is sensitive to internal and/or external variables – such as age, gender, diet, geographical location, time of day, environment and genetics (Wishart, 2019). Therefore, this metabolic phenotype is the product of an individual's genotype *and* environmental factors under certain conditions, and provides valuable information about what is happening within an individual as opposed to what might happen. This "top down" approach has many potential applications such as in personalised medicine, public healthcare, drug testing, nutrition analysis and veterinary science (Kim et al., 2016).

1.6.1. Overview of the metabolomics workflow

There are four key experimental designs to consider with metabolomics studies (Wishart, 2019). Firstly, the targeted approach which allows the identification of a small number of metabolites within a focused area of the metabolome and is generally for hypothesis testing. The untargeted approach is considered hypothesis generating as it allows the identification of as many metabolites as possible within the complex sample mixture (such as urine, serum, plasma etc.). Fluxomics, which measures the rate of metabolic reactions, and metabolite imaging, which allows for the in vivo or in vitro visualisation of metabolites, are the third and fourth approaches, respectively. These study designs can incorporate various methods to measure the metabolome, such as nuclear magnetic resonance (NMR), MS, LC-MS, GC-MS (Lenz and Wilson, 2007). Each method has its own advantages and disadvantages and the selected method would depend on the tools, time and funding available (Beckonert et al., 2007; Lenz and Wilson, 2007; Wishart, 2019). NMR, which was first utilised in the 1980s, provides structural information of small organic molecules based on atomcentred nuclear interactions and, although it is less sensitive than GC-MS and other MS methods, it is considered a robust, reliable and reproducible technique (Lenz and Wilson, 2007). NMR is inherently quantitative as the signal intensity is directly proportional to the number of nuclei present in the sample, this can either provide a relative or absolute quantification (Crook and Powers, 2020). Relative quantification is the measurement of molar concentrations relative to control samples or groups within a study which is determined by multivariate and univariate analyses of NMR spectral data. On the other hand, absolute quantification is the direct measurement of individual compound concentrations using internal standards or metabolite deconvolution (i.e. separation of molecules achieved computationally or experimentally) (Crook and Powers, 2020).

Once raw spectral data has been obtained via the selected method, the next challenge would be processing of this data by various software packages to allow for the identification and quantification of the metabolites within the complex sample mixture (Hendriks et al., 2011). Due to the rapidly growing field, the available software packages or tools are constantly improved and updated – roughly 85 new metabolomics software resources, tools, packages and databases were introduced in 2020 (Misra, 2021). Many of these packages are incorporated in developing software, such as R or Python, or are web based, and provide varying degrees of pre-processing, processing, annotation, metabolite identification and metabolite quantification from the raw data obtained from different platforms (such as GC-MS, NMR). Both univariate and multivariate statistical analyses can then be performed on the quantitative metabolite data (Saccenti et al., 2014). Univariate analysis (e.g. t-test, ANOVA) allows the comparison of one variable and focuses on the independent changes in metabolites, whereas multivariate analysis (e.g. correlations, linear regression) allows the comparison of two or more variables simultaneously, focusing on the relationship between metabolites and their complimentary behaviour in relation to biological processes (Saccenti et al., 2014). Both analytical methods should be used and interpreted within their statistical framework in

order to gain a holistic picture of the metabolomics data (Hendriks et al., 2011; Saccenti et al., 2014).

1.6.2. <u>Metabolomics in veterinary science</u>

The word "metabolome" first appeared in the literature in 1998 (Oliver et al., 1998) and since then the field has rapidly evolved with a large diversity of research including nutrition, toxicology, gene function, physiology, disease mechanisms and environmental science (Jones and Cheung, 2007; Wishart, 2019). Despite this diversity of research applications, there is limited metabolomics research within the field of veterinary and animal science (Jones and Cheung, 2007). In a systemic review by Tran et al. (2020) investigating the number of metabolomics studies in the development of spontaneous animal disease, the authors found a total of 38 in which most of them utilised MS or NMR techniques. Interestingly, the majority of the studies were in dogs (13 studies), cows (12 studies) and horses (5 studies) and focused on the digestive and reproductive systems. In addition to this data, the authors also noted that there were 45 studies related to food or nutrition, 23 related to reproduction, 29 related to toxicology or doping with the bulk of the studies linked to laboratory disease models, cell culture and induced experimental diseases (88 studies). There is an increasing interest in the metabolomics of skeletal muscle and meat in domestic animals, termed "MEATabolomics", which is particularly useful in identifying the metabolome profiles of meat quality traits such as colour, water holding capacity, pH, flavour and palatability (Muroya et al., 2020). This is also important in highlighting how factors such as species, genetics, breed, feeding, muscle type, post-mortem ageing, meat storage, processing and spoilage, affect meat quality through the distinguishable metabolite differences (Muroya et al., 2020).

1.6.3. <u>Metabolomics studies in the cheetah</u>

The majority of metabolomics studies have been conducted in domestic or companion animals, with limited studies in wild or captive animals. Only a handful of studies have been conducted using metabolomic methods in captive or free-roaming cheetahs to investigate metabolic properties (Koester et al., 2017; Tordiffe et al., 2017, 2016; Tordiffe and Mienie, 2019; Weiner et al., 2019). Weiner et al. (2019) used GC-MS to investigate hexose and pyruvate utilisation and fatty acid oxidation in the spermatozoa in three felid species (cheetah, clouded leopard and domestic cat). While Koester et al. (2017) identified a novel faecal biomarker, Immunoglobulin J, of early pregnancy in captive cheetahs using mass spectrometry – this protein marker was elevated in pregnant females within 4 weeks of mating. Both these studies were focused on improving the reproductive knowledge of cheetahs.

Tordiffe et al. (2016), investigated the serum fatty acid profile of free-ranging and captive cheetahs using GC-MS and found that there was a decreased concentration of monosaturated, polyunsaturated and total fatty acids in free-ranging cheetahs compared to their captive

counterparts. A year later, Tordiffe et al. (2017) used a similar approach to study the urinary organic acid metabolites in apparently healthy captive cheetahs whose diet mostly consisted of donkey meat supplemented with minerals and vitamins. A third of the most abundant organic acids detected were phenolic compounds associated with microbial fermentation of phenylalanine, tyrosine and glycine conjugates. Additionally, an unknown compound was identified that is believed to be metabolised from cadeverine, which is produced by bacterial decarboxylation of lysine. The increased lysine and the high number of phenolic compounds is potentially a result of the increased muscle meat intake in captivity. Lastly, Tordiffe and Mienie (2019) used an untargeted exploratory GC-MS and LC-MS approach to determine the serum and urine amino acid profiles of captive cheetahs. In serum, 38 amino acids were identified with glutamine, alanine, arginine and glycine having the highest concentrations. Similarly, 38 amino acids were identified in urine with arginine, glutamine, alanine and serine having the highest concentrations. Arginine, an essential amino acid in carnivores, is required for ammonia detoxification in the urea cycle as well as creatine and nitric oxide synthesis. Under normal dietary conditions, arginine deficiencies are unlikely with large felids generally maintaining large arginine or ornithine reserves to deal with the acute ammonia loads resulting from protein rich diets. The latter three studies provide baseline data on serum and urine metabolites, which emphasises the benefit of taking a systems biology approach to identify a broad range of metabolites in a number of different metabolic pathways. Additionally, it highlights the role of diet in the whole body metabolism of both captive and free-ranging cheetahs. This is particularly important in captive cheetahs that are fed primarily muscle meat diets high in protein, with the metabolic effect of these diets requiring further investigation.

It is clear that the effect of the captive diet has been a growing area of research in cheetahs over the past 20 years and has been suggested as a contributor to the high prevalence of gastrointestinal diseases seen in this population. Glycine, as the most abundant amino acid in collagen, is required in substantial amounts in the cheetah, yet it is likely deficient in the captive diet. Therefore, understanding the metabolic role of collagen and glycine in cheetahs may provide crucial information on the effects of the captive diet. The use of metabolomics techniques, which are becoming increasingly popular in veterinary and nutritional studies, may be a useful tool to investigate these effects. This approach has not previously been used to investigate the effect of diet in captive cheetahs or any other exotic felid.

1.7. Aims of the study

The aim of the study was to investigate the metabolic differences between glycine supplemented and non-supplemented captive cheetahs in a randomised cross-over study design. This was achieved through haematology and serum biochemistry analyses as well as an untargeted metabolomics approach using NMR analysis on the urine and serum samples in 10 captive cheetahs.

CHAPTER TWO

2. STUDY DESIGN AND METHODOLOGY

2.1. Introduction

Randomised cross-over controlled trials are designed to allow the individuals within the study to experience one or more treatments in a randomised order, and are considered the strongest study design in biomedicine and pharmacology (Elbourne et al., 2002). There are a number of strengths for this design, namely the ability of individuals to act as their own controls which limits the amount of inter-group variation and allows for a smaller sample size to be used (Elbourne et al., 2002). Randomised controlled trials are frequently used with human participants – a systematic review found that of the 519 randomised trials published in December 2000, only 22% were cross-over designs, of which 70% included a washout period (Mills et al., 2009). Washout periods reduce the risk of a potential carryover effect between treatments, whereby an effect of one intervention is carried over to the next intervention (Elbourne et al., 2002). In terms of dietary intervention controlled trials, the limitations are largely linked to the behaviour of humans – high dropout rates, poor adherence to interventions, incorrect reporting, and limited generalisability to other population groups (Hébert et al., 2016). These limitations would not necessarily apply to diet interventions in captive or domestic animals. However, intervention study designs have rarely been used in captive animals, including cheetahs.

The use of metabolomics techniques (such as NMR, GC-MS) provides valuable information about what is happening within an individual, in response to external or environmental stimuli, and is becoming a popular tool in a variety of scientific fields (Wishart, 2019). Nutritional metabolomics is the integration of metabolic profiling and nutrition in complex biosystems with the goal of characterising the endogenous metabolome and the food metabolome, the latter including the metabolites derived from food consumption and their subsequent metabolism in the body (Guasch-Ferre et al., 2018). Compared to plasma and serum, urine contains a higher concentration of non-metabolite and non-nutrient compounds that result from food phytochemicals and other chemicals making it an acute marker of frequently consumed foods (Guasch-Ferre et al., 2018). Additionally, analysis of faecal and urine metabolic composition reflects the status of the gut microbiome, which is largely affected by diet, and bridges the gap between the microbiome and the health of the host (Jain et al., 2019). These nutritional metabolomics studies are becoming increasingly popular in human populations through observational or intervention studies, but are less frequently used in exotic or domestic animal populations.

Therefore, the use of a randomized cross-over design in combination with metabolomics techniques and conventional serum and haematology analysis, may be a useful tool to investigate metabolic changes in response to diet in captive cheetahs. The aims of this chapter are to:

- describe the study design and glycine diet intervention of 10 captive cheetahs;
- describe the sample collection, body measurements, serum biochemistry and haematology methods;
- describe NMR analysis, data analysis and statistical analysis workflow for the urine and serum samples.

2.2. Ethical approval

Ethical approval for this study was obtained from the Animal Research Ethics Committee of the Faculty of Veterinary Science, University of Pretoria (reference number: REC231-19) and the Animal Research Ethics Committee of the University of the Western Cape (reference number: AR20/3/4). This study was conducted under the standing Threatened or Protected Species (TOPS) permit for the Faculty of the Veterinary Science, University of Pretoria (reference number: S02559). A Department of Agriculture, Forestry's and Fisheries Section 20 permit was also granted for the undertaking of this research (reference number: 12/11/1/7).

2.3. Study design and feeding intervention

2.3.1. Overview of the study design

The study was designed as a randomised cross over dietary intervention on 10 captive cheetahs (Acinonyx jubatus). The research venue was located at Cango Wildlife Ranch, Oudtshoorn, Western Cape, South Africa and took place during the months of August to November 2020. Ten apparently healthy adult male and female cheetahs between the ages of 2 and 5 years of age were included in the study group (Supplementary Table S7). To ensure the cheetahs had the same baseline diets, all animals were habituated to the control horse muscle minced meat (referred to as "horse meat" hereafter) and vitamin-supplemented diet for three weeks before randomly being assigned to a control or intervention group. The detail of the diets will be discussed later, but briefly: the control group was fed horse meat with a basic vitamin and mineral supplement (Panthera supplement, WildCat Nutrition, Pretoria, South Africa) and the intervention group was fed the same diet, but with the addition of 30g glycine powder per 1 kg meat fed (Glycine supplement, WildCat Nutrition, Pretoria, South Africa). The assigned cheetahs were fed the control or glycine supplemented diet for four weeks. This was followed by a two-week washout period where all animals received the control diet, thereafter the groups were crossed over to receive the opposite diet for a further four weeks. Samples were collected on three occasions - at baseline (after the habituation), after the first intervention and after the second intervention. An overview of the study design is shown in Figure 3.
2.3.2. Study animals and setting

The study included five male and five female cheetahs housed at Cango Wildlife Ranch in Oudtshoorn, Western Cape, South Africa. The males and females had an average age of 4.2 and 3.2 years, respectively, with an age range of 2 to 5 years old (Supplementary Table S7). At the time of the study, all animals were healthy with no clinical signs of gastritis or renal failure. The animals were housed in medium to large enclosures (400 m² to 1350 m²) away from the public in a rural setting. All the females were housed individually. Two males were housed individually, two males were housed together and a third male was housed with his brother who was not included in this study. It is common for male siblings to remain housed together in captivity as they would form coalitions in the wild (Hayward et al., 2006; Woc Colburn et al., 2018). None of the female cheetahs had cubs prior to the study. The study took place towards the end of winter and progressed into spring, with the average maximum temperature ranging from 19°C (during the habituation phase), to 25°C (during the first intervention phase) and 32°C (during the second intervention phase).

2.3.3. Diet before the study

Prior to the start of the study, all animals received a varied diet containing 1.3 to 1.8 kg horse or donkey muscle meat with 100 g organ mince and 100 g shredded skin six days a week, with one fasting day in the week. Additionally, they received bones with meat once a week and a vitamin and mineral predator supplement, calcium mixture and iodated salt on alternating days. This diet combination was implemented for at least 6 months prior to the start of the study and likely in excess of 2 years for the older cheetahs in the study. Therefore, the cheetahs received a diet with moderate to high collagen content (and therefore glycine content) for an extended period of time which may have long lasting effects. In order to standardise the diet across the 10 cheetahs, a three week habituation period was implemented prior to the first baseline sample collection (T1).

2.3.4. Habituation period

All animals were fed the control diet during the three week habituation period. The standardisation was implemented to control the glycine intake of the cheetahs because glycine is found in high concentrations in collagen (in skin and bones), but is relatively low in meat. Therefore, a meat only diet was fed during the habituation and control phase.

During the habituation period, cheetahs received between 1.5 and 2.2 kg horse meat with 10 g of vitamin and mineral supplement (Panthera supplement, WildCat Nutrition, Pretoria, South Africa) for every 1 kg meat fed. The supplement was sprinkled over and rubbed into the meat. The weekly feeding frequency remained the same, which included one fasting day, resulting in a total of 18 feeding days. Food consumption was monitored and all food was consumed. Baseline data and samples were collected after the three week habituation period.

2.3.5. Trial period

Following the first sample collection (T1), cheetahs were randomly assigned to a Control (no glycine) or Intervention group (glycine supplemented). Due to a logistical matter, the groups were not evenly split. During the first phase, six cheetahs (two males, four females) were assigned to the control diet, whereas only four cheetahs (three males, one female) where assigned to the glycine supplemented diet. Those on the control diet simply continued with the same diet that was implemented during the habituation period. The cheetahs in the intervention group were fed the same horse meat with the vitamin and mineral supplement and an additional 30 g of glycine powder per 1 kg meat which was sprinkled and mixed into the meat. Cheetahs remained on these diets for four weeks, amounting to 25 feeding days, thereafter the second sample collection occurred (T2).

All animals then underwent a two-week washout period where they were fed the control diet to negate any effects caused by the glycine supplementation and return to baseline. The groups were then crossed-over and fed the corresponding diet for another four weeks with a total of 25 feeding days. Therefore, the six cheetahs previously on the control diet received the glycine supplemented diet, and the four cheetahs previously on the glycine supplemented diet received the control diet (without glycine). The third and final sample collection (T3) took place after the four weeks.

Throughout the study, the standard feeding procedures of Cango Wildlife Ranch were followed, particularly the time of day when food was provided, number of days and the amount of meat provided. The standard feeding procedures were determined by the appropriate staff based on the cheetah's feeding history, appetite and behavioural response to the food. The nutritional details of the horse meat (from literature), vitamin and mineral supplement and the glycine supplement are reported in Table 1.

2.4. Sample collection procedure

2.4.1. Immobilisation

Samples were collected over two days on three occasions – at baseline, after the first intervention and after the second intervention. The cheetahs were immobilised at each sampling event via intramuscular hand injection with 30 µg/kg medetomidine hydrochloride (10 mg/ml, Medetomidine, Kyron Laboratories Pty LTD, South Africa) in combination with 1 mg/kg zolazepam/tiletamine (100 mg/ml, Zoletil ®, Virbac, South Africa). Once sedated, the cheetahs were transported to the medical centre and maintained under anaesthesia with 1-2% isoflurane in oxygen for the duration of the collection period which lasted approximately one hour. Throughout the sedation period the standard anaesthetic and cardiovascular parameters of the cheetahs were monitored. Once all the samples were collected, the sedation was reversed with 5 mg intramuscular atipamezole (5 mg/ml, Antisedan®, Pfizer, South Africa). Recoveries took between 15 to 20 min. Cheetahs were monitored regularly throughout the day and the following day.

Nutrient	Horse muscle meat (per 100g) ^a	Vitamin and Mineral (per 10g) ^b	Glycine (per 30g) ^b
Moisture (g)	70.9		
Protein (g)	19.8		
Lipids (g)	6.63		
Ash (g)	0.98		
Cholesterol (mg)	61		
Energy (kcal)	140		
SCFA (g)	2.0		
Monounsaturated FA (g)	2.7		
Polyunsaturated FA (g)	1.1		
Linoleic acid (mg)	678		
α-linolenic acid (mg)	318		
20:2 n-6 (mg)	19		
20:3 n-6 (mg)	14		
Arachidonic acid (mg)	58		
Ammonia (g)	0.26		
Essential amino acids (g)			
Histidine	0.90		
Isoleucine	0.91		
Leucine	1.52		
Lysine	1.57		
Methionine	0.48		
Phenylalanine	0.82		
Threonine	0.84		
Tryptophan	0.15		
Tyrosine	0.67		
Valine	0.96		
Non-essential/ Conditionally essential amino acids (g)			
Alanine	1.18		
Arginine	1.16		
Aspartic acid	1.77		
Cystine	0.20		
Glutamic acid	2.83		
Glycine	1.04		30
Proline	0.89		
Serine	0.69		
Hydroxyproling (g)	0.15		
Total collagen (g)	0.15		
Soluble collagon (%)	1.17		
Burine bases (mg)	10.2		
runne vases (my) Adonino	10.00		
Guanina	10.23		
Vanthina	0.20		
	9.01		
пуроханише	74.03		
		l	

Table 1. Nutrient breakdown of horse muscle meat, vitamin and mineral supplement and glycine supplement

Mineral (mg)			
Sodium	74.2		
Potassium	331.0		
Magnesium	28.9		
Calcium	3.8	3000	
Phosphorous	231.0	670	
Iron	3.9		
Zinc	3.7		
Copper	0.2	4.0	
Manganese		1.5	
Sodium chloride		340	
Vitamins			
Vitamin A (IU)		5000	
Thiamin (B1) (mg)	0.043	5.0	
Riboflavin (B ₂) (mg)	0.18	2.5	
Niacin (B₃) (mg)	5.54		
Pantothenic acid (B₅) (mg)		10	
Pyridoxine (B ₆) (mg)	0.64		
Biotin (B ₇) (µg)		100	
Folic acid (B ₉) (µg)		800	
Vitamin B ₁₂ (µg)	2.08		
Vitamin C (mg)		125	
Vitamin D ₃ (IU)		100	
Vitamin E (mg)		50	

^a Values obtained from Badiani et al. (1997); ^b values obtained from WildCat Nutrition, Pretoria, South Africa. SCFA, short chain fatty acids

2.4.2. Sample collection and storage

At the medical centre, the cheetahs were weighed, and body length and height were measured while the cheetah was in lateral recumbency. Length was measured from the occiput at the caudal edge of the skull to the base of the tail and height was measured from the dorsal rim of the scapula to the metacarpal pad. Length and height were only measured during the first sampling (T1). The size index of each cheetah was calculated by multiplying the body length with the shoulder height. The size index was divided by weight to determine the body mass index (BMI) of the cheetahs (Kirberger and Tordiffe, 2016).

Blood was collected from the jugular vein, using an 18G needle and 20 ml syringe, and immediately transferred to an EDTA and serum tube. The fresh blood was handled with care to avoid haemolysis of the red blood cells. The whole blood in the EDTA tube was used for haematological analysis on site (Abaxis Vetscan HM5 haematology system). After analysis, whole blood was stored at -20°C. The blood in the serum tube was allowed to clot for 30 min at room temperature and subsequently centrifuged for 10 min at 6000 rpm. The supernatant was transferred to a new microtube and stored at -20°C.

Urine was collected via urethral catheterisation using a sterile 6FG 120 mm feeding tube, after which it was centrifuged for 10 min at 6000 rpm to remove any cellular debris. The urine was then transferred to a microtube and stored at -20°C. The specific gravity of the urine was determined using a handheld refractometer: a drop of urine was placed on the glass cover and the specific gravity visually determined using an internal calibrated grid. The urine of the cheetah in general is more concentrated compared to humans and, therefore, values of between 1065 to 1080 au would indicate normal functioning kidneys. Specific gravity was evaluated at each sample collection time point (T1, T2 and T3) to ensure that all animals had normal kidney function to continue with the study. The urine specific gravity was appropriate for each cheetah with an average of 1069 ± 6 au.

Serum, whole blood and urine were transported on dry ice (-60°C) to the Department of Paraclinical Sciences, Faculty of Veterinary Science, University of Pretoria and stored at -80°C. Serum biochemistry analysis was conducted in the analytical laboratory at the University of Pretoria, while the remaining serum, urine and skeletal muscle samples were transported on dry ice to the Centre for Human Metabolomics, North West University, where they were stored at -80°C until NMR analysis.

2.5. Blood analyses

2.5.1. Serum biochemistry and haematology

Whole blood was analysed on site at Cango Wildlife Ranch using the Abaxis Vetscan HM5 haematology system set to the cat species setting. The white blood cell (WBC) and red blood cell (RBC) count, haemoglobin, haematocrit, mean corpuscular volume (MCV), mean corpuscular haemoglobin (MCH), mean corpuscular haemoglobin concentration (MCHC), red blood cell distribution width (RDW), platelet count and percentage, mean platelet volume (MPV) and platelet distribution width (PDW) were determined.

Routine serum biochemistry analyses were performed in the Clinical Pathology laboratory, Department of Companion Animal Clinical Studies, Faculty of Veterinary Science, University of Pretoria. The concentration of the following were determined using the Cobas Integra® 400 plus analyser (Roche, Randburg, South Africa): Total serum protein (TSP), albumin (Alb), alanine aminotransferase (ALT), alkaline phosphatase (ALP), urea, creatinine, phosphate, triglycerides and total calcium. This is a high throughput, fully automated, software controlled system that uses absorbance spectrophotometry, fluorescence polarisation and electrolyte analysis. Sodium, potassium and chloride were measured using the Rapidpoint® 500 System (Siemens Ltd, Isando, South Africa) specific electrolyte assay cartridges.

2.5.2. Statistical analysis

Statistical calculations for blood analyses were performed using GraphPad Prism 5 version 5.03 (GraphPad Software, CA, USA). Values are expressed as mean ± standard deviation (SD) and

normality was checked within diet groups using the Shapiro-Wilks normality test. As the study design split the animals into Group 1 with six animals and Group 2 with four animals (Figure 3), a non-parametric Mann-Whitney U test was performed to identify if there was any difference between the split groups for all measured variables. There was no difference between groups, therefore, groups were combined under each diet, i.e. a combined baseline, control (no glycine diet) and glycine supplemented diet. The groups were combined in all further analysis. A repeated measures one-way ANOVA with a Tukey post hoc test was used to compare the three groups. Significance was set at p < 0.05.

2.6. NMR analysis

At the conclusion of the feeding intervention and sample collection period, all samples were transported on dry ice to the Centre for Human Metabolomics, North West University, for NMR analysis. An untargeted whole metabolome ¹H-NMR approach was utilised to identify and quantify the metabolites present in the urine and serum samples of 10 captive cheetahs at baseline, on a control (no glycine) diet and on a glycine supplemented diet.

2.6.1. Urine sample preparation

A 1 mL urine aliquot was added to a micro tube and centrifuged for 5 min at 12 000 x g. Thereafter, 540 μ L was transferred to another micro tube and mixed with 60 μ L of NMR buffer solution (9:1). The NMR buffer solution consisted of 1.5 M potassium phosphate buffer (pH 7.4) in deuterium oxide and 5.8 mM trimethylsilyl-2,2,3,3-tetradeuteropropionic acid (TSP; Sigma-Aldrich, St. Louis, Missouri, USA) with a trace amount of sodium azide included to prevent bacterial growth in the sample. This was followed by centrifugation for 5 min at 12 000 x g. A final volume of 540 μ L was transferred to a 5 mm NMR tube for metabolomics analysis.

2.6.2. Serum sample preparation

Samples were filtered using Amicon Ultra 2 mL centrifugal units with 10 kDa membrane filters (Merck; UFC201024). Prior to use, each centrifugal filter unit was pre-rinsed with dH₂0 and centrifuged at 4500 x g for 10 min to remove trace amounts of glycerol and glycerine from membrane filters, which can interfere with NMR signals. 800 μ L of serum was then filtered and centrifuged at 4500 x g for 30 min. 540 μ L of each filtered sample and 60uL buffer solution (same as urine) was then added to a micro centrifuge tube and mixed. The sample mixture was centrifuged at 12 000 x g for 5 min. A final volume of 540 μ L was transferred to a 5 mm NMR tube for metabolomics analyses.

2.6.3. System and experimental settings

The same settings were used for urine and serum. A 500 MHz Bruker Avance III HD NMR spectrometer equipped with a triple-resonance inverse (TXI) ^{1}H { ^{15}N , ^{13}C } probe head and x, y, z

gradient coils was used to analyse the samples. ¹H spectra were acquired as 128 transients in 32K data points with a spectral width of 10504 Hz and acquisition time of 3.12 s. The receiver gain was set to 90.5. The sample temperature was maintained at 300K and the H₂O resonance was pre-saturated by single-frequency irradiation during a relaxation delay of 4 s, with a 90° excitation pulse of 8 μ s. Shimming of the sample was performed automatically on the deuterium signal. Fourier transformation and phase and baseline correction were done automatically. Software used for NMR processing was Bruker Topspin (version 3.5).

2.7. Data analysis

The raw Bruker Topspin spectral data files were obtained and spectra analysis was performed using R programming software (R, version 4.0.4) and RStudio integrative environment (RStudio, version 1.4.1106, MA, USA). The Automatic Statistical Identification in Complex Spectra (ASICS; version 2.6.1, Bioconductor, NY, USA) R package was used to identify and quantify metabolites in the complex spectra. This is described in detail by Lefort et al. (2019) and Tardivel et al. (2017). The following sections are a summary of the various data analysis steps.

2.7.1. Data processing

Briefly, the raw Bruker spectral files were imported into RStudio and the complex spectrum underwent several pre-processing steps. Firstly, baseline distortions caused by instrument instability were corrected by constructing a baseline and removing it from the spectrum. Secondly, a speag algorithm was used to align the spectra against an automatically detected reference spectrum in order to remove any metabolite peak position change due to temperature or pH variations. Thirdly, unwanted regions in the spectrum corresponding to water (4.5 to 5.1 ppm) and urea (5.5 to 6.5 ppm) were removed as their wide range can interfere with the detection of less concentrated molecules. The final pre-processing step was normalisation which minimises the systematic variations due to differences in sample dilutions, therefore making samples comparable - the Probabilistic Quotient Normalisation (PQN) method was used for this dataset (Dieterle et al., 2006). The PQN method assumes that changes in concentrations of a single analyte only influence parts of the spectra and, therefore, the intensity of a majority (but not all) of the signals is a function of dilution. PQN appears to be a more robust and consistent method than normalising to a constant sum. This method of normalisation was selected over correcting to creatinine concentration because creatinine has been shown to be highly variable in cheetahs, and this more holistic method of normalising may provide a better representation of the NMR spectrum (Ryan et al., 2011; Tordiffe et al., 2017).

The ASICS package provides a pre-processed library of the spectra of pure compounds which was used as a reference to identify and quantify metabolites found in the complex spectra of interest. The list of library compounds is included in Supplementary Table S1. Several steps were required

to adapt the library to the spectra of interest, such as noise thresholding, removal of library metabolites that were not found in the complex mixture spectra and the alignment of the library spectra with the complex mixture spectra.

2.7.2. Metabolite identification and quantification

The identification of metabolites in the complex sample mixture relies on various linear and warping functions. This provides reliable identification by controlling two possible sources of error: identifying a metabolite that is not present in the complex mixture (false positive) or not identifying a metabolite that is present (false negative). A least squares method was used to quantify the identified metabolites. This method did not allow identification of metabolites at an absolute concentration below 1 μ M, as the signal to noise ratio was too low, as well as metabolites at an absolute concentration above 1 M which resulted in broadening and overlapping of the NMR signals.

This process identified metabolites and provided an estimated relative concentration which is a function of the complex mixture, reference library and number of protons for each selected metabolite. This relative concentration or quantification allows comparison between samples, but it is not an absolute concentration. The list of identified metabolites and their relative quantifications for each cheetah was exported from RStudio as an Excel workbook.

2.8. Statistical analysis

All statistical analysis for NMR data was conducted using MetaboAnalyst version 5.0 (https://www.metaboanalyst.ca) (Chong et al., 2019; Pang et al., 2021). JMP Pro version 16.1.0 (SAS Institute, North Carolina, USA) was used to create the figures. Data are expressed as mean ± standard deviation where appropriate and significance was set at p<0.05.

MetaboAnalyst is a comprehensive tool that offers a variety of univariate and multivariate tests specifically designed for large datasets, such as in metabolomics studies, but limitations within this application do exist. In particular, MetaboAnalyst has very limited analyses for paired data of multiple groups, with its strength being the analyses of unpaired groups or two paired groups. The current study has a cross-over design which creates three paired groups at baseline, on the control diet and on the glycine supplemented diet. In order to obtain a more holistic statistical overview, multiple univariate and multivariate tests were utilised after the data normalisation steps. This more "thoughtful" approach, as described by Wasserstein et al. (2019) prioritises the scientific context and makes use of multiple approaches to solve a problem, which is especially relevant considering the exploratory nature of this study. The various approaches are described below, which combined, give rise to the notable potentially important metabolites of interest. This process is summarised in Figure 3.

2.8.1. Pre-analysis check of groups

The current study design included a total sample of 10 cheetahs which were divided into 2 uneven groups of n=6 (Group 1) and n=4 (Group 2) which is displayed in Figure 3. These groups experienced different lengths of time on the control (or glycine deficient diet) as well as either one (Group 1) or two (Group 2) dietary changes during the study. Group 1 experienced a total of 9 consecutive weeks on the control diet, while Group 2 experienced 6 consecutive weeks on the control diet as the glycine intervention occurred immediately after the habituation period in the latter group. Therefore, these groups may have different levels of glycine deficiency with Group 1 potentially being a better representation of glycine deficiency before experiencing the glycine intervention. As a result, it was necessary to first check the urine and serum metabolite analysis for each group before examining the combined effect. The steps described below were conducted for both the n=6 group and the combined n=10 group. The details and results of the n=6 analysis are described in Appendix 2. After the analysis it was deemed suitable to use the combined n=10 groups as it produced more consistent data and this approach was utilised in Chapters 4 to 6.

2.8.2. Data cleaning and scaling

Metabolites that were present in less than 50% of the samples in all three groups (i.e. baseline, control and glycine supplemented) were removed in order to reduce the noise or effect of infrequently observed metabolites, across all samples, in the subsequent analyses. After cleaning, the metabolite concentration list was imported into MetaboAnalyst 5.0. The data underwent various steps prior to analysis, including an integrity check, replacing zero values by 20% of the lowest value, data filtering based on interquartile range, which removed approximately 10% of outlying values and auto scaling normalisation which makes the metabolites more comparable by scaling to the sample mean and standard deviation. As there was a wide range of concentrations between metabolites, these steps were necessary to ensure that all the values were equally weighted especially for multivariate analysis. It makes use of the following equation to calculate the scaled value for a single metabolite of a sample: scaled value = $(X_2 - X_1)/SD_1$ where X_2 is an individual sample metabolite concentration, X_1 is the metabolite mean for all samples, SD_1 is the metabolite standard deviation for all samples.

2.8.3. Univariate analysis

A one-way repeated measures ANOVA was performed to identify differences in metabolites between the baseline, control and glycine supplemented groups. False Discovery Rate (FDR) was used to correct for the multiple univariate comparisons and protect against false positive results. A post hoc analysis was not possible using MetaboAnalyst. Thus, the normalised and scaled data of the significantly identified metabolites from MetaboAnalyst was imported into JMP to identify where the difference was between groups using a Tukey-Kramer HSD post hoc analysis adjusting for multiple comparisons.

2.8.4. Multivariate analysis

Principle component analysis (PCA) and partial least squares discriminant analysis (PLS-DA) was performed on the normalised data in MetaboAnalyst. PCA is an unsupervised data reduction method which does not take class labels (diet) into account when explaining the variance within the data, while PLS-DA is a supervised data reduction method which takes the class labels into account when creating predictive models for the different classes. Importantly, both these methods allow the reduction of large or multi-dimensional datasets with many variables into simple onedimensional datasets that retains the most important components contributing to the variation. This allows you to work with compressed datasets without losing the crux of the original data (Saccenti et al., 2014) For PLS-DA, the leave-one-out cross validation method was used for five components in which the cross validated sum of the squares (Q2) was an estimate of the predictive ability of the model. Loading scores, which indicate the influence of each metabolite on the variation seen in the data, were recorded for PCA. The important measures for PLS-DA, namely the variable importance in projection (VIP) and the weighted sum of absolute coefficient of regression for each diet, were recorded for each metabolite. All three recorded measures for PCA and PLS-DA provide an indication of the metabolites that are contributing the most to the variation or separation seen in the analyses and can be used as a tool to highlight metabolites of interest. All metabolites were ranked from highest to lowest influence for PC 1 and PC 2 for the PCA loading scores and PLS-DA VIP sores and for each diet of the PLS-DA coefficient of regression.

2.8.5. Identifying metabolites of interest

The metabolites identified with the one-way ANOVA and the top 20% of metabolites with the greatest influence under each parameter of the PCA (loading scores) and PLS-DA (VIP and coefficient of regression) analysis were highlighted. These metabolites were further reduced to those that were present in four or more analyses out of a potential of 8. This ultimately created a list of potentially important urinary and serum metabolites in captive cheetahs on a baseline, control and glycine supplemented diet.

An overview of the study methodology is shown in Figure 3. The results obtained will be discussed and structured as follows: Chapter 3 will focus on the body measurements, serum biochemistry and haematology; Chapter 4 will focus on the urine metabolite; Chapter 5 will focus on the serum metabolite; and Chapter 6 will provide a general discussion and conclusion.



Figure 3. Overview of the study methodology including study design, feeding intervention, data analysis and statistical analysis.

CHAPTER THREE

3. BODY MEASUREMENTS AND BLOOD DATA

3.1. Introduction

The morphology and physical condition were extensively investigated in wild Namibian cheetahs, in which there was notable sexual dimorphism. Adult males weighed 45.6 ± 6.0 kg and females 37.2 ± 5.1 kg, with the adult body mass only being reached between four to eight years of age (Marker and Dickman, 2003). The average body length and shoulder height of males and females were 1.02 m and 0.77 m, and 0.98 m and 0.74 m, respectively. In a captive Namibian population, the average body weight of males and females were 40.1 kg and 36.4 kg, respectively, while the body length and shoulder height were 0.99 m and 0.73 m (for males), and 0.96 m and 0.69 m (for females) (Kirberger and Tordiffe, 2016). Kirberger and Tordiffe (2016) used those three measurements to establish a size index and body mass index (BMI) for cheetahs. There is variation in the size and dimensions of cheetahs depending on the population group, with captive cheetahs generally being smaller and lighter, as well as the methods of measurements (Marker and Dickman, 2003).

Haematology and serum biochemistry have been analysed in both captive and free-ranging cheetahs in several studies (Bechert et al., 2002; Caro et al., 1987; Depauw et al., 2012b; Hawkey and Hart, 1986; Hudson-Lamb et al., 2016). Of all these studies, two stand out, where Bechert et al. (2002) investigated the blood values of captive cheetahs fed a commercial or supplemented meat diet and Depauw et al. (2012) investigated the serum data of captive cheetahs fed a whole rabbit or supplemented beef diet.

Although diet has been investigated in a limited number of studies in captive cheetahs, the effect of glycine supplementation has not been investigated in this population specifically in regards to morphometric and blood parameters. Therefore, the aims of this chapter are to:

- Compare the feeding and body measurements between baseline, control and glycine supplemented groups.
- Compare the haematology data between baseline, control and glycine supplemented groups.
- Compare the serum biochemistry data between baseline, control and glycine supplemented groups.

3.2. Results

3.2.1. Diet and feeding

No adverse response to the control or glycine supplemented diet was reported. Glycine is sweet and highly palatable (McCarty et al., 2018) and all the food offered to the cheetahs was consumed. The amount of food provided to each animal remained fairly consistent throughout the study with three males, from Group 2, having a 200g meat increase as the study progressed (12.5% average increase). One male increased from the habituation period to the glycine supplemented diet, and the other two increased from the glycine supplemented diet to the control diet. The average consumption for males and females per dietary phase is reported in Table 2.

(ii = 5) and remain (ii = 5) cheetans during the nabituation, control and grycine intervention periods.								
	Horse muscle meat (kg)	Vitamin and mineral supplement (g)	Glycine supplement (g)					
Habituation (3 weeks)								
Male	1.7 ± 0.3	17.4 ± 2.5	0					
Female	1.8 ± 0.3	18.0 ± 2.6	0					
Control diet (4 weeks)								
Male	1.9 ± 0.2	18.6 ± 1.5	0					
Female	1.8 ± 0.3	18.0 ± 2.6	0					
Glycine diet (4 weeks)								
Male	1.8 ± 0.3	17.8 ± 2.6	53.4 ± 7.8					
Female	1.8 ± 0.3	18.0 ± 2.6	54.0 ± 7.6					

Table 2. Average daily amount of meat, vitamin and mineral supplement and glycine supplement for mal
(n = 5) and female $(n = 5)$ cheetahs during the habituation, control and glycine intervention periods.

Data are expressed as mean \pm SD.

3.2.2. Body measurements

The weight and body measurements of the study cheetahs are reported in Table 3. As the body parameters of cheetahs have been shown to be sexually dimorphic (Marker and Dickman, 2003), male and female data have been shown separately. However, statistical difference was only investigated between the different diets, and not sex, as groups were matched. The average weight at baseline and on the control diet were 40.2 ± 4.5 kg and 39.5 ± 4.0 kg, respectively. Weight significantly decreased to 38.8 ± 4.4 kg from baseline on the glycine supplemented diet. At baseline, the BMI of the cheetahs was 52.2 ± 3.6 kg/m² which was similar to the control diet at 51.2 ± 2.9 kg/m². The glycine diet had a significantly lower BMI of 50.3 ± 3.5 kg/m² compared to the baseline. All the parameters fall within the ranges reported by Kirberger and Tordiffe (2016).

	Baseline		Contro	ontrol Diet G		Diet	Reference range
	Mean	SD	Mean	SD	Mean	SD	
Weight (kg)	40.2*	4.5	39.5	4.0	38.8*	4.4	
Female	37.5	2.6	36.8	2.3	36.5	2.3	30.9-44.1
Male	43.0	4.6	42.2	3.6	41.1	4.9	33.6-50.6
Body Length (m)							
Female	1.00	0.04	—	—	—	—	0.92-1.01
Male	1.03	0.06	_	—	—	—	0.93-1.10
Shoulder height (m)							
Female	0.74	0.02	—	—	—	—	0.64-0.73
Male	0.77	0.02	_	—	—	—	0.68-0.77
Size index (m ²)							
Female	0.74	0.04	—	—	—	—	0.61-0.72
Male	0.80	0.06	—	—	—	—	0.63-0.84
BMI (kg/m ²)	52.2*	3.6	51.2	2.9	50.3*	3.5	
Female	50.6	1.9	49.7	1.1	49.3	0.8	45.9-66.4
Male	53.7	4.4	52.7	3.5	51.3	5.0	42.9-63.1

Table 3. Body weight and measurements of male (n=5) and female (n=5) captive cheetahs at baseline, control and glycine diet.

* indicates p<0.05 between combined male and female Baseline and No Glycine groups; — indicates data was not measured; BMI, body mass index. Size index calculated as body length x shoulder height. BMI calculated as weight/size index. Reference range from Kirberger and Tordiffe (2016).

3.2.3. <u>Haematology</u>

The whole blood haematology data are shown in Table 4. Neutrophils contributed the most to the count with 7.97 \pm 2.45 x10⁹/L (baseline), 8.49 \pm 4.09 x10⁹/L (no glycine control) and 8.62 \pm 2.74 x10⁹/L (with glycine). Eosinophils increased significantly from baseline (0.15 \pm 0.07 x10⁹/L) to the control diet (0.26 \pm 0.08 x10⁹/L) and the glycine diet (0.25 \pm 0.09 x10⁹/L). Similarly, the basophil count also increased from the baseline diet (0.01 \pm 0.01 x10⁹/L) to the control and glycine supplemented diets with 0.02 x10⁹/L, albeit remaining the lowest count of the white blood cells.

The RBC count remained similar across the diets with $8.72 \pm 0.45 \times 10^{12}$ /L at baseline, $8.54 \pm 0.44 \times 10^{12}$ /L with the control diet and $8.33 \pm 0.59 \times 10^{12}$ /L with the glycine diet. The various RBC parameters remained similar across the three diets – haemoglobin, haematocrit, MCV, MCH and RDW. The platelet count was 406 ± 124 ×10⁹/L (baseline), 396 ± 78 ×10⁹/L (no glycine diet) and 429 ± 89 ×10⁹/L (glycine diet). The platelet parameters, including mean platelet volume (MPV) and volume in the blood (PCT), remained similar across the diets. The platelet distribution width significantly increased from baseline (42.5 ± 0.3%) to the glycine supplemented diet (42.8 ± 0.5%), with the control diet at 42.6 ± 0.5%.

The reference haematology data are from Bechert et al. (2002) and Hawkey and Hart (1986) to provide an indicator of the range found in captive cheetahs. All the haematological data across the diets fall within the reference ranges provided in Table 4.

	Base	eline	Contro	l Diet	Glycin	e Diet	Previous	vious literature	
Parameters	Mean	SD	Mean	SD	Mean	SD	Bechert ^a	Hawkey ^b	
WBC (x10º/L)	11.12	3.24	11.30	4.19	11.31	2.81	7.6-16.2	5.9-12.0	
Lymphocytes	2.44	1.51	1.90	0.47	1.88	0.31	0.9-3.1	0.9-2.7	
Monocytes	0.55	0.28	0.63	0.37	0.54	0.30	0.07-0.64	0-0.3	
Neutrophils	7.97	2.45	8.49	4.09	8.62	2.74	5.4-12.9	5-9.4	
Eosinophils	0.15*#	0.07	0.26*	0.08	0.25#	0.09	0.09-0.81	0.0-1.4	
Basophils	0.01*#	0.01	0.02*	0.00	0.02#	0.01	_	0	
RBC (x10 ¹² /L)	8.72	0.45	8.54	0.44	8.33	0.59	5.7-10.3	6-8.3	
Haemoglobin (g/L)	160.00	8.46	155.80	8.72	153.60	8.81	102-195	116-166	
Haematocrit (%)	48.43	2.06	47.47	1.74	46.51	2.57	29.3-58.7		
MCV (fL)	55.60	1.71	55.60	1.71	55.90	1.73	46.1-58.1	50-62	
МСН (рд)	18.38	0.87	18.23	0.69	18.46	0.50	15.9-20.0	18.4-21.8	
MCHC (g/L)	330.60	13.90	327.70	7.39	330.30	6.07	328-348	315-369	
RDWc (%)	19.59	0.50	19.80	0.49	19.48	0.35	—		
RDWs (fL)	40.55	1.83	40.92	1.32	40.54	1.40	_	—	
Platelets (x10 ^s /L)	406.00	123.58	395.60	78.47	429.00	89.09	243-475	_	
MPV (fL)	12.60	1.30	12.61	0.71	12.54	0.81	_		
PCT (%)	0.50	0.14	0.50	0.09	0.53	0.08	_	_	
PDWc (%)	42.53#	0.31	42.57	0.49	42.84#	0.50	—		
PDWs (fL)	26.21	1.32	26.27	1.48	26.81	1.19	_	_	

Table 4. Haematology data of 10 captive cheetahs at baseline, with the control and glycine diet.

* indicates p<0.05 between Baseline and Control groups; [#] indicates p<0.05 between Baseline and Glycine groups; indicates data not available. WBC, white blood cells; RBC, red blood cells; MCV, mean corpuscular volume; MCH, mean corpuscular haemoglobin; MCHC, mean corpuscular haemoglobin concentration; RDW, red blood cell distribution width; MPV, mean platelet volume; PCT, platelet volume in blood; PDW, platelet distribution width. Reference range from Bechert et al. (2002) and Hawkey and Hart (1986).

3.2.4. Serum biochemistry

The serum biochemistry data of the 10 captive cheetahs are shown in Table 5. The TSP and, therefore, the albumin concentration was significantly different from the baseline (TSP: 63.5 ± 2.8 g/L; albumin: 42.7 ± 1.8 g/L) to the control diet (TSP: 61.9 ± 3.4 g/L; albumin: 40.6 ± 2.1 g/L) and glycine diet (TSP: 60.9 ± 3.1 g/L; albumin: 40.5 ± 1.6 g/L). ALP, an enzyme predominantly synthesised in the liver, was significantly higher at baseline (14.5 ± 9.4 U/L) compared to the control diet (11.2 ± 6.7 U/L) and glycine supplemented diet (10.2 ± 5.7 U/L). Potassium, on the other hand, was higher in the control diet (4.1 ± 0.1 mmol/L) compared to the baseline (3.9 ± 0.3 mmol/L), whereas the glycine supplemented diet (4.1 ± 0.2 mmol/L) was similar to the baseline diet. The total calcium concentration (bound and free) was higher at baseline (2.66 ± 0.06 mmol/L) compared to the glycine supplemented diet (2.59 ± 0.05 mmol/L), with the control diet at 2.61 ± 0.09 mmol/L. ALT, urea, creatinine, sodium, phosphate, chloride and triglyceride concentrations remained constant across the three groups.

The serum reference data included are based on the research conducted by Bechert et al. (2002), Depauw et al.(2012) and Hudson-Lamb et al. (2016). Most of the parameters in Table 5, including those that were significantly different, fall within the indicated reference ranges, but ALP activity is noticeably lower in the current study compared to what was reported by Bechert et al. (2002).

	Base	line	Contro	l Diet	Glycin	e Diet	Previous literature		
	Mean	SD	Mean	SD	Mean	SD	Bechert ^a	Hudson- Lamb ^b	Depauw ^c
TSP (g/L)	63.54*#	2.84	61.87*	3.37	60.93#	3.14	60-76		69 ±1
Albumin (g/L)	42.65*#	1.81	40.56*	2.08	40.51#	1.59	35-43		40 ±1
Globulins (g/L)	20.91	2.13	21.31	2.76	20.44	2.56	24-33		29 ±1
ALT (U/L)	79.19	23.78	74.37	17.80	71.96	18.06	66-198		69 ±13
ALP (U/L)	14.50*#	9.44	11.22*	6.68	10.23#	5.65	22-30		
Urea (mmol/L)	15.12	3.15	14.29	2.23	14.73	2.38		7.4-22.9	16.1 ±2.3
Creatinine (µmol/L)	190.50	29.30	182.90	20.19	184.80	26.93	176-389	114-276	180 ±23
Na (mmol/L)	155.58	0.68	155.27	1.01	155.09	0.47		148-160	157 ±1
K (mmol/L)	3.97*	0.25	4.22*	0.11	4.07	0.17		3.8-5.2	4.2 ±0.1
CI (mmol/L)	121.14	1.33	121.93	1.78	121.69	1.45		116-129	122 ±2
Phos (mmol/L)	1.65	0.23	1.66	0.24	1.72	0.28			1.9 ±0.3
Triglycerides (mmol/L)	0.53	0.12	0.54	0.12	0.47	0.12			0.59 ±0.16
Ca(T) (mmol/L)	2.66#	0.06	2.61	0.09	2.59#	0.05			2.7 ±0.1

Table 5. Serum data of 10 captive cheetahs at baseline, control diet and glycine diet.

* indicates p<0.05 between Baseline and No Glycine groups; [#] indicates p<0.05 between Baseline and Glycine groups. TSP, total serum protein; ALT, alanine aminotransferase; ALP, alkaline phosphatase; Na, sodium; K, potassium; Cl, chloride; Phos, phosphate; Ca(T), total calcium. Reference range from Bechert et al. (2002), Hudson-Lamb et al. (2016) and Depauw et al. (2012).

3.3. Discussion

3.3.1. Body measurements and weight loss

There was a decrease in the average body weight of the cheetahs from baseline to the glycine supplemented diet, and this was likely due to a reduction in the body weight of the males as the females remained fairly consistent in weight. This change in weight also led to a corresponding decrease in BMI from the baseline to the glycine supplemented diet, which was calculated using the weight, length and shoulder height of each animal. Although the weight of the animals decreased, they were still well within a healthy range, such as what was provided by Kirberger and Tordiffe (2016). As reported in Table 2, the amount of meat consumed by each animal remained constant throughout the study, with a marginal increase during the glycine phase with \pm 100 g meat – this was due to an increase in the food consumption of one male in Group 2 after the habituation period. The other change in the amount of meat consumed was for two males, also in Group 2, who

increased their intake by 200 g each as they moved from the glycine supplemented diet to the control diet. This is shown by the slightly higher amount of muscle meat consumed during the control diet phase. This increase was determined by the designated carers who deemed it necessary due to apparent hunger based on the cheetah's behaviour. Therefore, the weight loss is not due to a decrease in food intake, as the amount remained constant or increased slightly for three individuals.

The study started in August, which is the last month of winter in the Southern hemisphere, and progressed through to early November, marking the end of spring. As expected, there was a large average daily temperature increase between the habituation phase $(12 \pm 4 \text{ °C})$, the first diet phase $(16 \pm 4 \text{ °C})$ and the second diet phase $(22 \pm 5 \text{ °C})$. Colder ambient temperatures promote thermogenesis in which organisms activate adaptive metabolic processes to maintain core body temperature, therefore increasing daily energy expenditure (Hansen et al., 2010). Therefore, the increase in environmental temperature is unlikely to account for the weight loss of the cheetahs. Historically cheetahs have occupied a wide variety of habitats including dry forests to arid deserts and would undoubtedly be adapted to the hot and humid conditions of the area (Durant et al., 2015). It is likely that any of these acute changes (in particular in response to the changed diet) would have occurred during the habituation phase before the start of the study or would be controlled for by the split grouping of the animals.

A potential contributor to the weight loss is the study diet itself, which would be sub-optimal. Prior to the study, the animals received a combination of organ mince, skin and bones which would provide them with a greater array of nutrients more aligned to their carnivorous diet in the wild. However, in order to investigate the effects of the glycine alone, it was necessary to remove all other potential sources of glycine, such as the collagen of bones and skin (Li and Wu, 2018). While body weight appeared to decrease from the baseline to the control diet, it was not significantly different, indicating that the glycine supplemented diet potentially played a larger role in weight loss. Interestingly, when the group of four and six cheetahs were looked at separately there was still a significant reduction in weight from the baseline to glycine supplemented diet. In healthy humans supplemented with 1 mmol glycine/kg body weight, the concentration of plasma glycine increased 4 fold within 40 min and remained elevated for 2 hours, indicating rapid absorption of glycine (Gannon et al., 2002). Glycine also constantly stimulated the release of glucagon (which increases blood glucose), but in contrast, when glycine was ingested with glucose, this reduced the plasma glucose concentration by half. The authors speculated that the oral ingestion of glycine stimulates the secretion of a gut hormone that works with insulin to reduce glucose circulation and inhibits the effect of glucagon despite the increased concentration (Gannon et al., 2002). In sucrose fed rats (which promotes intra-abdominal fat accumulation), glycine supplementation reduced the amount of intra-abdominal fat, blood pressure, adipocyte size and the concentration of various nonesterified fatty acids (e.g. palmitic acid, arachidonic acid) returned to control levels (El Hafidi et al.,

53

2004). This forms part of the increasing evidence base of the beneficial glycine supplementation effects in the potential treatment of obesity and type 2 diabetes (Wang et al., 2013). Glucose and insulin concentrations were not measured in the serum of the captive cheetahs using this particular method, but glucose was identified with the NMR analysis. The effect of glycine on this aspect of metabolism may have contributed to their weight loss through the loss of fat and increased β -oxidation or other changes in carbohydrate and fat metabolism (EI Hafidi et al., 2004).

3.3.2. Haematology

Although not significant, there was an increase in the total WBC count from the baseline diet to the control and glycine supplemented diet. However, the increase in eosinophils and basophils were significantly higher between diets. The WBC measured in Table 4 forms part of the innate immune system and provides the front line, rapid response to foreign microbes - the innate and acquired immune system are dependent on the availability of certain amino acids for the biosynthesis of proteins and polypeptides (Li et al., 2007). Therefore, dietary protein deficiencies compromise the immune system and its ability to function effectively (Li et al., 2007). Glycine is an effective antiinflammatory and immunomodulatory agent, although the mechanisms of its action is not completely understood (Zhong et al., 2003). A potential mechanism is the activation of the glycine receptor (GlyR), which is a glycine gated chloride channel found in a variety of neuronal, inflammatory and immune response cells such as macrophages, monocytes and neutrophils (Zhong et al., 2003). The binding of glycine acts as an inhibitory neurotransmitter leading to the influx of chloride into the cells, which counters the effect of any excitatory neurotransmitter and also blocks the influx of intracellular calcium. This prevention of intracellular calcium suppresses the inflammatory response of these cells, and therefore, glycine plays a regulatory role in the production of cytokines and inflammatory mediators, enzyme activation, cell proliferation and cell death (Zhong et al., 2003).

However, the seemingly slight increase in WBC in the control and glycine supplemented diet, largely driven by the increase in the granulocytes (neutrophils, basophils and eosinophils), is likely due to the sub-optimum nature of the diet as highlighted under the weight loss section. Additionally, the similarity between the control and glycine groups may indicate that an aspect other than glycine, such as a different component of collagen, is driving this WBC increase. The WBC increase may lead to an increased immune response to counter the effects of potential infections caused by malnutrition (Li et al., 2007). Similarly, the significant increase in platelet distribution width between the baseline and the glycine supplemented diet, which is an indicator of the size variation in the platelets, may be linked to the slight increase in platelet count in the blood. The immune cell range in the current study is similar to what was found in Bechert et al. (2002) in which the cheetahs were also fed a variety of supplemented meat over a 12 month period, confirming that the cheetahs from the present study were within a normal immune response range.

Glycine is crucial in the biosynthesis of porphyrins in which eight molecules of glycine are required for the biosynthesis of one haem group (ferroporphyrin) (Meléndez-Hevia et al., 2009). In humans, an estimated 240 mg/day glycine is consumed in the production of porphyrins which are mostly used for the haem of haemoglobin or myoglobin, or for the production of catalase and peroxidase (Meléndez-Hevia et al., 2009). In this study, there was no difference in the RBC or related parameters in response to the glycine supplement. As the average RBC, haemoglobin and haematocrit values were in the higher range of what was found in the reference literature, this may indicate that the healthy cheetahs in this study were at a maximum capacity and that the additional glycine was used in the biosynthesis of other metabolites such as purines or creatine.

3.3.3. Serum biochemistry

There was a decrease in the TSP driven by the decrease in albumin from baseline to the control and glycine supplemented diets. Although the meat protein content was the same throughout, prior to the start of the study the cheetahs were fed bones, organs and skin which would provide a more diverse source of amino acids. The potential reduction in the dietary protein source, and therefore amino acids, may have led to a reduction in the serum albumin biosynthesis rate which was observed in rats fed high or low protein diets – with very little change to the catabolism rate until a critical level is reached (Kirsch et al., 1968). In human patients, the relationship between nutritional status, body composition and serum protein and albumin were investigated (Forse and Shizgal, 1980). A highly significant positive correlation was found between body cell mass (which is the total oxygen consuming cells of the body as determined by body mass and body potassium, and is accompanied by the expansion of extracellular mass) and TSP, as well as the serum albumin. The authors concluded that although nutritional status is an important determinant of serum albumin concentration, there are other factors such as malnutrition, malabsorption, liver failure, wounds, and infections that may be contributing factors. Therefore, the decrease in TSP and serum albumin is likely linked to the cheetah's weight loss and the sub-optimum diet with the supplementation of glycine not providing the variety of amino acids required to maintain the serum protein levels.

There was a significant decrease in ALP concentration from baseline to the control and glycine supplemented diet. Although not significant, the same trend was seen with ALT. Both are predominantly liver enzymes that will leak into the bloodstream when there is liver injury or damage (Fernandez and Kidney, 2007). Therefore, a decrease in these enzymes may indicate improved liver functioning or less damage in response to the control and glycine supplemented diet. However, there are various physiological factors that could increase ALP activity — such as age (with juvenile cats and dogs having higher concentrations than adults), pregnancy and lactation and a high fat diet (which has been found in humans), although these factors are not apparent in the current study (Fernandez and Kidney, 2007). Interestingly, the ALP concentration in the study by Bechert et al. (2002) was higher than what was found in the current study. This result might be partly due to older

cheetahs being included in that study or due to the higher fat content of the commercial diets. The generally lower ALP and ALT found in the current study may simply indicate that these cheetahs are indeed young and healthy.

The potassium concentration of the control diet was significantly higher than at baseline and with the glycine supplemented diet. Although all values were in the reference ranges and therefore not indicative of hypo or hyperkalaemia. The total serum calcium concentration was significantly higher at baseline compared to the glycine supplemented diet. This decrease may be directly attributed to the lower serum albumin concentration found in glycine supplemented cheetahs, which would lead to less bound calcium in the serum. Throughout the study, no bones were provided to the cheetahs which would conventionally be a major source of calcium and highlights the importance of calcium supplementation for captive cheetahs fed exclusively meat based diets (Depauw et al., 2012b). The supplement provided to the cheetahs in this study had sufficient levels of calcium and would unlikely contribute to the calcium decrease. Different combinations of immobilisation drugs, such as ketamine-medetomidine or zolazapam-tiletamine-medetomidine, have been shown to cause hypertension and hypoxaemia with the latter combination causing elevated blood calcium, potassium and glucose concentrations (Buck et al., 2022). This drug combination was used in the current study and may have attributed to the differences found in the serum biochemistry parameters, but is highly unlikely, as the same drug combinations were used during all three immobilisation events.

CHAPTER FOUR

4. URINE METABOLITES

4.1. Introduction

Urine, in mammals, is a transparent and sterile fluid generated by the kidneys and consists of a variety of compounds at a high concentration (Bouatra et al., 2013). These compounds include urea, inorganic salts, creatinine, ammonia, organic acids, various water soluble toxins and pigmented products of haemoglobin. The medical diagnostic value of urine has been explored since 4000 BC with a total of 2651 metabolites identified and quantified in the urine of humans (Bouatra et al., 2013). In metabolomics studies, urine is the most common sample used due to its metabolite abundance, easy non-invasive collection, limited sample preparation and limited sample pretreatment owing to its lower complexity and protein content (Zhang et al., 2012). Urinary metabolites, which are end products of normal and pathological cellular processes in humans and animals, are closely linked to phenotypes and act as an ideal proxy for the metabolic activity of the organs (i.e. kidney) from which it is derived (Wishart, 2019; Zhang et al., 2012). Urine metabolomics has successfully been used in a range of fields, such as physiology, diagnostics, functional genomics, pharmacology, toxicology and nutrition (Zhang et al., 2012). Importantly, urine metabolomics is emerging as a significant tool in the discovery of biomarkers and the diagnosis of many diseases including jaundice syndrome, various cancers, chronic obstructive pulmonary disease (COPD), renal injury and diabetes (Zhang et al., 2012).

There have been a limited number of studies investigating the urine metabolites of captive or freeranging cheetahs. Burger et al. (2006) utilised GC-MS to investigate the volatile constituents in the urine headspace vapour of captive male and female cheetahs and found that elemental sulphur (S₆ and S₈) was present in all samples. The authors concluded that this may be another mechanism of sulphur excretion or that it allows cheetahs to avoid detection from prey and predators due to its odourless secretion. The study by Tordiffe et al. (2017) detected 339 organic acids in the urine of healthy captive cheetahs, of which, phenolic compounds linked to the anaerobic fermentation of aromatic amino acids in the intestine and glycine conjugates were present in high concentration. Furthermore, Tordiffe and Mienie (2019) identified 38 amino acids in the urine of captive cheetahs with arginine having nearly twice the concentration of the next amino acid, glutamine. Arginine is an essential amino acid which cats are unable to produce endogenously and, therefore, require substantial amounts from the diet in order to maintain the ornithine reserves for ammonia detoxification in the urea cycle. Cystine, which is reduced in urine to form two cysteine molecules, had the highest fractional excretion (22%) which indicates that a high percentage of cystine is excreted and not reabsorbed by the body. This may be due to exogenous methionine and serine from the consumed animal tissue meeting the high metabolic demand of glutathione, taurine, CoA and general protein synthesis.

Glycine has many physiological functions and, as a result, there are many glycine linked metabolites that may be found in urine, most notably those associated with glycine conjugation (Badenhorst et al., 2014). Hippurate (or hippuric acid) is the most abundant amino acid conjugate found in the urine of mammals and is formed with the conjugation of glycine and benzoate. Although benzoate, salicylate, 4-hydroxybenzoate, 3-hydroxybenzoate, 4-aminobenzoate and 2-furoate are the natural substrates for conjugation, the major source of glycine conjugation substrates seems to be metabolites of dietary polyphenols produced by microorganisms in the gut (Badenhorst et al., 2014). These compounds include phenylpropionate, 3-hydroxyphenylpropionate and 4-hydroxyphenylpropionate. The amount of glycine conjugates excreted in urine depends on the dietary intake of polyphenolic compounds and their extent of fermentation in the colon (Badenhorst et al., 2014). Glycine forms part of the biosynthesis of carnitine and creatine which can also be excreted in urine, the latter as creatinine (Meléndez-Hevia et al., 2009). Glutathione and benzoate metabolism are indicators of glycine deficiency, with increased excretion of pyroglutamic acid (5-oxoproline) and hippurate acting as markers, respectively (Meléndez-Hevia et al., 2009).

Considering the limited urinary metabolomics studies conducted in cheetahs and the potential benefits of using this approach to investigate the metabolic effect of glycine, the aims of this chapter are to:

- identify and quantify (through relative concentrations) the metabolites present in the urine of 10 captive cheetahs at baseline, on a control diet and on a glycine supplemented diet;
- highlight and discuss the identified metabolites of importance.

4.2. Results

4.2.1. Identified metabolites and relative concentration

ASICS identified a total of 151 metabolites across all 30 urine samples. After data cleaning, 25% of the metabolites were removed and 112 metabolites remained. The 30 most abundant metabolites for each group and their relative concentrations are indicated in Table 6 and all identified metabolites are shown in Supplementary Table S2. The most abundant metabolite across all three groups was creatinine with a five to six times higher relative concentration than allantoin, the next abundant metabolite. L-cysteine and malonic acid were also abundant across all three groups, while L-ornithine (baseline), TMAO (control) and glycolic acid (glycine diet) were abundant in the respective diets. As displayed in Table 6, glycolic acid and guanidinoacetic acid were abundant in the glycine supplemented diet in 7 individuals, but interestingly, was present in less than 5

individuals in the baseline and control diet. It was, therefore, omitted from the baseline and control groups in Table 6.

Baseline			Con	trol d	iet	Glycine supplemented diet		
Metabolite	n	Relative concentration	Metabolite	n	Relative concentration	Metabolite	n	Relative concentration
Creatinine	10	64.1 ± 47.4	Creatinine	10	72.8 ± 37.6	Creatinine	10	56.9 ± 46.2
Allantoin	10	14.3 ± 3.2	Allantoin	10	13.7 ± 2.9	Allantoin	10	15.2 ± 3.8
L-Cysteine	10	10.5 ± 0.7	Malonic acid	10	13.2 ± 4.9	L-Cysteine	10	10.1 ± 1.6
Malonic acid	10	7.8 ± 4.7	ΤΜΑΟ	10	10.2 ± 7.2	Glycolic acid*	7	9.6 ± 6.2
L-Ornithine	10	6.1 ± 1.3	L-Cysteine	10	9.9 ± 0.9	Malonic acid	10	9.6 ± 4.0
L-Proline	10	4.5 ± 0.4	L-Carnitine	10	7.3 ± 2.6	ΤΜΑΟ	10	7.3 ± 4.7
DHA	10	4.3 ± 1.1	L-Ornithine	10	5.8 ± 0.9	Guanidinoacetic acid*	7	6.8 ± 5.2
L-Carnitine	10	3.9 ± 2.3	L-Proline	10	5.5 ± 0.7	L-Ornithine	10	6.4 ± 1.2
ΤΜΑΟ	10	3.9 ± 3.6	Taurine	10	4.4 ± 1.4	L-Proline	10	5.1 ± 0.5
1,3-DAP	10	3.4 ± 0.3	DHA	9	3.9 ± 0.9	L-Carnitine	10	4.9 ± 2.2
L-Tryptophan	8	3.1 ± 0.6	Choline chloride	10	3.8 ± 1.2	DHA	10	4.1 ± 0.8
L-Arginine	10	3.1 ± 0.5	Phosphocholine	10	3.5 ± 0.9	Taurine	10	3.8 ± 1.1
Taurine	10	3.0 ± 1.1	α-Ketoglutaric acid	10	3.4 ± 0.4	L-Glutamine	10	3.2 ± 0.8
Choline chloride	10	3.0 ± 1.1	L-Glutamine	10	3.4 ± 1.1	Choline chloride	10	3.2 ± 1.1
α-Ketoglutaric acid	10	2.7 ± 0.4	D-Glucose	10	3.2 ± 0.6	1,3-DAP	10	3.1 ± 0.3
D-Maltose	10	2.6 ± 0.4	D-Maltose	10	3.2 ± 0.5	L-Arginine	10	3.0 ± 0.3
D-Glucose	10	2.6 ± 0.4	L-Arginine	10	3.2 ± 0.7	α-Ketoglutaric acid	10	2.9 ± 0.3
L-Lysine	9	2.6 ± 0.4	Dimethylglycine	10	3.2 ± 2.1	L-Tryptophan	10	2.8 ± 0.6
Phosphocholine	10	2.6 ± 1.0	Creatine	7	3.1 ± 2.8	Phosphocholine	10	2.8 ± 0.7
L-Citrulline	10	2.6 ± 0.2	1,3-DAP	10	2.9 ± 0.3	D-Maltose	10	2.6 ± 0.6
L-Glutamine	10	2.5 ± 0.9	Threonic acid	8	2.7 ± 1.0	Threonic acid	9	2.6 ± 0.7
D-Galactose	9	2.5 ± 1.1	L-Tryptophan	9	2.7 ± 0.6	L-Lysine	8	2.5 ± 0.4
β-Alanine	10	2.4 ± 0.3	β-Alanine	10	2.7 ± 0.2	D-Glucose	10	2.5 ± 0.6

Table 6. The 30 most abundant metabolites identified by NMR in the urine of 10 captive cheetahs at baseline, on a control diet and on a glycine supplemented diet.

D-Gluconic acid	8	2.4 ± 0.8	D-Gluconic acid	7	2.6 ± 0.9	β-Alanine	10	2.5 ± 0.3
Threonic acid	9	2.0 ± 0.7	1-Methyl- hydantoin	10	2.6 ± 2.3	L-Citrulline	10	2.5 ± 0.3
Glycero- phosphocholine	10	1.9 ± 0.6	L-Citrulline	10	2.6 ± 0.4	Glycero- phosphocholine	10	2.1 ± 0.5
Phenethylamine	10	1.8 ± 0.5	L-Asparagine	9	2.6 ± 1.1	Phenethylamine	10	1.9 ± 0.3
Glycogen	10	1.8 ± 0.2	L-Lysine	8	2.5 ± 0.4	Glycogen	10	1.8 ± 0.3
Lactose	10	1.7 ± 0.5	Glycero- phosphocholine	10	2.4 ± 0.5	Lactose	10	1.8 ± 0.8
L-Cystine	10	1.6 ± 0.3	Phenethylamine	10	2.2 ± 0.9	L-Cystine	10	1.7 ± 0.3

Data are expressed as mean \pm SD. n indicates the number of samples where the metabolite was detected. Only those present in >5 samples were included in the table. * indicates metabolites that were present at high relative concentration in <5 samples of the baseline and control diets. Relative concentration is an arbitrary unit. TMAO, Trimethylamine oxide; 1,3-DAP, 1,3-Diaminopropane; DHA, dehydroascorbic acid

4.2.2. Metabolites identified by univariate analysis

MetaboAnalyst identified 31 significant metabolites with a one-way repeated measures ANOVA. The significant metabolites as well as their post hoc analysis conducted in JMP are shown in Table 7. Importantly, these metabolites are not necessarily those that were the most abundant in Table 6. Dimethylamine had the lowest adjusted p-value (p<0.001) with the control and glycine supplemented diet significantly greater than at baseline. The relative concentrations of the control diet were greater than baseline for 22 of the identified metabolites, with only 1,3-diaminopropane having a higher concentration at baseline compared to the control diet. Similarly, the relative concentrations of the control diet were higher than the glycine supplemented diet for 11 metabolites with the glycine diet being greater than the control for fructose and CMP. For most of the metabolites, the baseline and glycine supplemented diets were the same with the exceptions being dimethylamine (higher in the glycine diet) and methylmalonic acid (higher at baseline). While L-cystine, isovaleric acid, phosphocholine, pimelic acid and hypotaurine had an overall significance identified with ANOVA, but this was not shown in the post hoc analysis between groups presumably due to the different statistical programmes.

			Tukey-Kramer p-value			
Metabolite	F-value	Adjusted p-value	Baseline- Control	Baseline- Glycine	Control- Glycine	
Dimethylamine	28.56	0.0003	<0.01 (C>B)	0.01 (G>B)	NS	
α-Ketoglutaric acid	19.64	0.0016	<0.01 (C>B)	NS	0.02 (C>G)	
Methylguanidine	15.91	0.0037	< 0.01 (C>B)	NS	<0.01 (C>G)	
L-Carnitine	14.86	0.0037	0.01 (C>B)	NS	NS	
Dimethyl sulphone	14.53	0.0037	<0.01 (C>B)	NS	<0.01 (C>G)	
L-Proline	12.57	0.0064	<0.01 (C>B)	NS	NS	
L-Cystine	12.35	0.0064	0.05	NS	NS	
Pyroglutamic acid	11.95	0.0066	<0.01 (C>B)	NS	NS	
Trimethylamine	10.96	0.0084	<0.01 (C>B)	NS	<0.01 (C>G)	
Methylmalonic acid	10.89	0.0084	NS	0.04 (B>G)	NS	
L-Asparagine	10.30	0.0100	0.02 (C>B)	NS	0.01 (C>G)	
Trans-Aconitic acid	9.71	0.0122	0.02 (C>B)	NS	NS	
L-Carnosine	9.51	0.0124	<0.01 (C>B)	NS	NS	
D-Fructose	9.31	0.0125	NS	NS	<0.01 (G>C)	
D-Maltose	9.19	0.0125	0.03 (C>B)	NS	0.02 (C>G)	
Taurine	8.32	0.0183	0.04 (C>B)	NS	NS	
Isovaleric acid	7.41	0.0280	NS	NS	NS	
Malonic acid	7.25	0.0288	0.04 (C>B)	NS	NS	
β-Alanine	6.94	0.0324	0.03 (C>B)	NS	NS	
Acetoacetic acid	6.84	0.0327	0.01 (C>B)	NS	NS	
L-Valine	6.53	0.0357	0.04 (C>B)	NS	0.01 (C>G)	
3-Hydroxyphenylacetic acid	6.52	0.0357	NS	NS	0.04 (C>G)	
Phosphocholine	6.41	0.0365	NS	NS	NS	
Dimethylglycine	6.17	0.0396	<0.01 (C>B)	NS	<0.01 (C>G)	
Pimelic acid	6.13	0.0396	NS	NS	NS	
D-Glucose	5.72	0.0483	0.03 (C>B)	NS	0.02 (C>G)	
1-Methylhydantoin	5.67	0.0483	<0.01 (C>B)	NS	0.02 (C>G)	
Hypotaurine	5.57	0.0496	NS	NS	NS	
CMP	5.49	0.0496	NS	NS	0.04 (G>C)	
ΤΜΑΟ	5.46	0.0496	0.04 (C>B)	NS	NS	
1,3-DAP	5.40	0.0499	0.04 (B>C)	NS	NS	

Table 7. Significant metabolites identified by univariate analyses in the urine of captive cheetahs at baseline, on a control diet and glycine supplemented diet (n=10)

NS, not significant; CMP, cytidine monophosphate; TMAO, Trimethylamine oxide; 1,3-DAP, 1,3-Diaminopropane

4.2.3. Metabolites identified by multivariate analysis

The PCA and PLS-DA score plots are shown in Figure 4. Principal component (PC) 1 accounts for 27.5% of the variation seen across the data, while PC 2 accounts for 9.5% of the variation. The PCA plot shows overlap with all three diets, with the control diet having a greater spread. Based on the spread of the data, the metabolites driving the variation in PC 1 (i.e. the spread on the x axis) are contributing more to the baseline and glycine diet, while the metabolites driving the variation in PC 2 (i.e. the spread on the y axis) are contributing similarly to PC 1 for the control diet. All the metabolites would contribute to the variation, however the loadings or contribution of the top 10 metabolites of PC 1 and PC 2 are shown in Table 8. Pantothenic acid, L-carnitine, malonic acid and choline chloride were the metabolites with the highest loading scores for PC 1, therefore contributing the most to the axis distribution. In contrast, UDPG, glutaconic acid, trimethylamine and D-fructose had the highest loadings for PC 2.

The separation seen on the PLS-DA plot between the diets is noticeably small, with the control diet overlapping with both the baseline and glycine diet. This is confirmed with the negative Q2 values (Supplementary Figure S1) which indicate that this model is not appropriately predicted or is overfitted for each diet. As with the PCA plot, the control diet has a wide spread of data. The discriminant variables of component 1 accounts for 18.4% of the variation between the diets, while component 2 accounts for 15.1% of the variation seen between diets. The difference between the glycine supplemented and baseline diets account for the greatest separation along component 1 and component 2, respectively. The overall VIP scores and the coefficient variables of each diet for the first 10 metabolites are shown in Table 8. Dihydrothymine, methylmalonic acid, dimethylamine and pyroglutamic acid were the variables with the greatest importance in both component 1 and 2 in the PLS-DA model. Due to the discriminatory nature of PLS-DA in separating according to groups, the coefficients of regression indicate the contribution of each metabolite to the predicted model for each diet. Dihydrothymine, dimethyl sulphone and dimethylamine had the highest weighting at baseline. Methylguanidine, dimethyl sulphone and trimethylamine contributed to the control diet. Similarly to the baseline diet, dihydrothymine had a large contribution towards the glycine supplemented diet. This was followed by dimethyl sulphone and N-(2-Furoyl)Glycine. A more extensive list for the multivariate analyses is provided in Supplementary Table S3.



Figure 4. A two dimensional PCA (top) and PLS-DA (bottom) scores plot of the metabolites identified in the urine of 10 captive cheetahs at baseline, on the control and glycine supplemented diets. *Each point represents an individual cheetah at baseline (red), control (blue) and glycine supplemented (green) with a corresponding 95% confidence ellipse for each diet.*

A. PCA Metabolite loadings					B. PLS-DA Metabolite VIP Scores				
PC 1	PC 1 PC 2			C 1		C 2			
Pantothenic acid	-0.17	UDPG	-0.22		Dihydrothymine	2.63	Dihydrothymine	2.19	
L-Carnitine	0.17	Glutaconic acid	-0.21		Methylmalonic acid	2.40	Methylmalonic acid	1.79	
Malonic acid	0.17	Trimethylamine	0.20		Dimethylamine	2.36	Dimethylamine	1.75	
Choline chloride	0.16	D-Fructose	-0.19		Pyroglutamic acid	2.13	Pyroglutamic acid	1.65	
Acetoacetic acid	0.16	Spermidine	0.19		L-Proline	2.02	GMP	1.59	
L-Proline	0.16	Dimethylglycine	0.19		GMP	1.94	N-(2-Furoyl)Glycine	1.57	
Isovaleric acid	-0.16	Methylguanidine	0.18		1,3-DAP	1.93	L-Proline	1.55	
Acetaminophen	0.15	Lactic acid	-0.18		D-Fucose	1.89	Adipic acid	1.52	
Taurine	0.15	L-Histidine	-0.18		Dimethyl sulphone	1.69	1,3-DAP	1.47	
Ethanolamine	0.15	2-Deoxycytidine	0.17	-	L-Glutamine	1.61	UMP	1.45	

Table 8. The (A) PCA loadings, (B) PLS-DA VIP scores and (C) PLS-DA diet regression coefficients for the top 10 urinary metabolites in 10 captive cheetahs.

C. PLS-DA metabolite coefficients for each diet									
Baseline		Control	Glycine						
Dihydrothymine	97.8	Methylguanidine	71.9	Dihydrothymine	100.0				
Dimethyl sulphone	88.1	Dimethyl sulphone	68.1	Dimethyl sulphone	79.8				
Dimethylamine	84.4	Trimethylamine	60.9	N-(2-Furoyl)Glycine	75.8				
N-(2-Furoyl)Glycine	71.7	Dimethylglycine	56.2	GMP	74.1				
1,3-DAP	71.7	α-Ketoglutaric acid	52.1	Pimelic acid	68.3				
Methylmalonic acid	71.2	D-Fructose	50.1	Dimethylamine	68.0				
GMP	70.9	1-Methylhydantoin	49.1	Methylmalonic acid	64.8				
Pimelic acid	63.7	D-Glucose	46.9	1,3-DAP	60.6				
L-Proline	61.5	Dimethylamine	46.8	Adipic acid	59.4				
UMP	60.4	L-Valine	44.6	L-Tryptophane	54.8				

Loadings are the weight of each metabolite and are expressed as positive or negative number depending on its direction from the PC. Coefficients are the weighted sum of the PLS-DA regression for each group. PCA, principle component analysis; PLS-DA, partial least squares discriminant analysis, VIP, variable importance in projection; PC, principle component; C, component; TMAO, trimethylamine oxide; UDPG, uridine 5'-diphosphate glucose; GMP, guanosine monophosphate; UMP, uridine monophosphate; CMP, cytidine monophosphate; 1,3-DAP, 1,3-diaminopropane

4.2.4. Summary of potentially important metabolites

After the univariate and multivariate approach, the initial 112 urinary metabolites were reduced to 65 metabolites (Supplementary Table S4). These metabolites were present in the top 20% of at least one of the analyses. Ten metabolites were identified in 5 or more analyses – dimethyl sulphone, L-proline, D-fructose, dimethylamine, trimethylamine, pyroglutamic acid, 1,3-diaminopropane, dihydrothymine, methylmalonic acid and pimelic acid. The normalised concentration of the 10 identified metabolites at baseline, on the control and glycine supplemented diet are shown in Figure 5. Methylmalonic acid and 1,3-diaminopropane had significantly higher

concentrations at baseline compared to the glycine and control diets, respectively. L-proline and pyroglutamic acid had significantly higher concentrations on the control diet compared to baseline, while fructose had a higher concentration on the glycine diet compared to the control diet. Dimethyl sulphone and trimethylamine had higher concentrations on the control diet compared to both the baseline and glycine supplemented diets. Dimethylamine had a higher concentration on the control and glycine supplemented diet compared to the baseline diet. Significance was not identified univariately between groups for dihydrothymine and pimelic acid. Interestingly, dihydrothymine was identified in 9 cheetahs on the glycine diet, but only 5 and 6 individuals on the baseline and control diets. Similarly, fructose was identified in 10 cheetahs on the glycine diet, but in only 8 and 3 cheetahs on the baseline and control diets, respectively.





Figure 5. The normalised concentration of 10 potentially important urinary metabolites identified in 10 captive cheetahs at baseline (red), on a control diet (blue) and on a glycine supplemented diet (green). The dots represent individual values, the bar represents the mean normalised concentration and the grey line represents one standard deviation from the mean. * and # indicates p<0.05 between groups with matching symbols. Zero values were replaced by 20% of the lowest value during data scaling. Normalised concentrations may be a positive or negative number because it is scaled relative to the metabolite mean for all samples.

4.3. Discussion

4.3.1. Overview of identified metabolites in urine

In the current study, a total of 151 metabolites were identified in the urine of captive cheetahs of which 39 metabolites were removed because they were present in less than 50% of the samples in each group. Due to the untargeted approach, these metabolites span multiple classes of compounds, such as amino acids, organic acids and compounds, fatty acids, acylcarnitines and sugars. The identification of glycogen and acetaminophen as listed in Table 6 and 8 were somewhat surprising as the former is a large polysaccharide and the latter is paracetamol that is often administered as a drug. Cheetahs have been found to excrete relatively large molecules, which may include polysaccharides like glycogen although this hasn't been found previously (Tordiffe et al., 2017). Alternatively, this may be a misidentification of another glucose derivative. Similarly, paracetamol can occur naturally in plants which may be ingested by cheetahs in their enclosures, but whether this would present in their urine is not apparent. None of these metabolites were highlighted in multiple analyses. The identification of 112 metabolites in the current study aligns with previous human studies in which an average of 100 urine metabolites are identified per study across different platforms (Bouatra et al., 2013). Additionally, sex was not specifically investigated in the present study due to the randomised cross-over design which allowed for matched comparison between groups and reduced any confounding effects caused by sex. Tordiffe et al. (2016, 2017, 2019) found no significant difference in urinary amino acids, serum amino acids and serum fatty acids between male and female captive cheetahs. For urinary organic acids, 6 organic acids out of a potential 199 were highlighted as contributing significantly towards the separation of males and females using PCA and PLS-DA - namely N-acetylphenylalanine, 2-hydroxy-5methoxybenzoic acid, N-hexanoylglycine, N-tiglylglycine, 2,3-dihydroxybenzoic acid and furoylglycine (Tordiffe et al., 2017). Interestingly, these are all associated with glycine or benzoic acid, but they were not identified in other analyses and, therefore, may simply be indicative of the high abundance of glycine conjugates found in the study. Overall, these three studies found that captivity and age had a greater association with the identified metabolites and that the influence of sex was minor.

Creatinine is commonly used to control for the effects of urine volume on total urine concentration because, in the absence of kidney disease, it is believed to be excreted at relatively constant and unchanged amounts from the kidneys (Ryan et al., 2011). However, there are challenges with this approach, primarily regarding the day-to-day variability in excretion due to exogenous factors such as diet, sex, age, time of day and disease. Secondly, considering the large number of metabolites identified in metabolomics studies at varying concentrations, it may not be suitable to normalise an entire metabolome to a single metabolite, i.e. a more holistic approach should be taken (Ryan et al., 2011). In healthy captive cheetahs, the urine creatinine concentration was highly variable and largely influenced by age with a peak reached between four and six years old (Tordiffe et al., 2017).

Tordiffe et al. (2017) concluded that normalising urine metabolite concentrations to creatinine would potentially lead to the overestimation of metabolite concentrations in older and younger cheetahs, and suggested the use of urine specific gravity where the variability was lower due to the tight regulation of water in carnivorous mammals. Since the creatinine variability in the current study was quite low ($186 \pm 25 \mu mol/L$) across all three diets and the urine specific gravity fairly constant ($1069 \pm 6 au$) urine metabolites were not normalised to these values. Instead, the urine data was normalised during the pre-processing steps using the PQN method which adjusts the individual metabolites in relation to the NMR spectrum. This procedure was performed after the removal of the unwanted water and urea regions in the spectrum (Dieterle et al., 2006).

There were a total of 65 urine metabolites identified across multiple analyses of which 31 were significant in univariate analysis. The sections below will focus on the 10 identified metabolites of interest, which were highlighted in five or more analyses, and discuss their potential link to glycine supplementation.

4.3.2. Dimethyl sulphone (DMSO₂)

DMSO₂ (also known as methylsulfonylmethane, MSM) was present in the urine of all 10 cheetahs at baseline, on the control diet and on the glycine supplemented diet, and was identified as important across both univariate and multivariate analyses. DMSO₂ is a natural organic sulphur compound that was identified in normal human urine in 1966 and has since been identified in the urine, blood, cerebrospinal fluid, skin, sweat and earwax of various species including cows, rats, bobcats, African wild dogs, red deer and cheetahs (Burger et al., 2006; He and Slupsky, 2014). The concentrations of DMSO₂ in the biofluids of humans are highly influenced by diet with the consumption of foods such as onions, grains and fibres, cabbage, meats, eggs, dairy and fish leading to increased concentrations of DMSO₂ (He and Slupsky, 2014; Trimigno et al., 2020). In a comprehensive urinary NMR metabolomics study conducted in a Danish population, DMSO₂ was a key discriminatory metabolite of the conventional Danish diet characterised by sugary products, refined grains (pasta and rice included), meat, dairy, and low-fibre vegetables (Trimigno et al., 2020). The authors concluded that the notable presence of $DMSO_2$ was likely due to the contrast in gut microbial metabolism between the conventional Danish diet and the New Nordic diet rich in fish, whole grains and organic fruit and vegetables. Additionally, in a recent study investigating the serum metabolome of patients with chronic kidney disease-mineral and bone disorder it was found that increased DMSO₂ was the most notable metabolite in patients with high bone turnover (Baptista et al., 2020). The authors hypothesised that this may be due to its role in osteogenesis which was found in earlier *in vitro* studies, but the direct link to bone turnover and mineralisation is not clear. Interestingly, glycine was also increased in patients with high bone turnover and is likely due to increased collagen turnover (Baptista et al., 2020).

The origin of DMSO₂ is believed to be microbial and, while the pathways are poorly understood, evidence suggests that DMSO₂ production is linked to the decomposition of methionine by microbiota in the gastrointestinal tract (He and Slupsky, 2014). Dietary methionine that escapes absorption in the small intestine is then subject to absorption and catabolism by colonic bacteria which converts it to the highly reactive methanethiol. Methanethiol is either non-enzymatically converted to dimethyl disulphide or dimethyl trisulfide, or enzymatically (via the host enzymes) converted to dimethyl sulphide. The latter pathway ultimately leads to the production of DMSO₂ which is one of the major end products of the methionine degradation pathway (He and Slupsky, 2014). In patients with inflammatory bowel disease, DMSO₂ was earmarked as an important biomarker as it was significantly lower in the serum of these patients compared to healthy controls (Dawiskiba et al., 2014). This potentially indicates the role of the microbial methionine degradation pathways in intestinal disease. Glycine was also highlighted as a key biomarker as it was increased in inflammatory bowel disease patients.

DMSO₂ has previously been identified in the urine headspace vapour of five captive cheetahs using GC-MS (Burger et al., 2006). The authors hypothesised that the excretion of organic sulphur compounds such as DMSO₂, which is an odourless compound, is linked to territory marking and allows cheetahs to be undetected in close proximity to prey and predators. However, this study did not investigate the role of diet and disease in this small sample. In the current study, DMSO₂ was significantly higher in the urine of the control diet compared to both the baseline and glycine supplemented diets. The amount of methionine received by the cheetahs would have remained constant throughout the study and other potential sources of related organic sulphur compounds would be negligible (Table 1). Therefore, the increased excretion of DMSO₂ in the control diet is likely linked to the increased catabolism of methionine by the gut microbiota. This effect is potentially corrected when cheetahs undergo glycine supplementation leading to reduced production of DMSO₂ as the exogenous source of glycine reduces the pressure on endogenous glycine biosynthesis, in which methionine is an intermediate required for the biosynthesis of glycine from choline (Wang et al., 2013). The baseline diet represents a more optimum metabolite picture because the cheetahs received organ mince, skin and bones three weeks prior to the sampling. Therefore, the lower DMSO₂ reflects the remnants of an optimal diet in captivity, much like lower urine concentrations reflected the healthier Nordic diet in a Danish population (Trimigno et al., 2020). In the current study, DMSO₂ was the most prevalent metabolite identified across multiple analyses which highlights its importance in this captive cheetah population. This is particularly important when the high prevalence of gastrointestinal disease in captive cheetahs is considered and its potential link to diet (Lane et al., 2012; Munson et al., 1999).

4.3.3. L-Proline

Proline was identified in the urine of all 10 cheetahs across the baseline, control and glycine supplemented diet and was significantly higher in the control diet compared to baseline. The individuals on the glycine diet had a similar normalised concentration distribution to those on the control diet (Figure 5). Proline is a rigid amino acid with a unique cyclic structure which allows it to play a vital structural role in collagen along with other metabolic and physiological roles such as an antioxidant and signalling molecule (Li and Wu, 2018). Along with glycine and hydroxyproline (which is a metabolite of proline), proline plays a direct role in the biosynthesis of collagen which is the most abundant protein in mammals. Proline can conventionally be endogenously synthesised from arginine (from ornithine), glutamate and glutamine, but the lack of pyrroline-5-carboxylate synthase in cats does not allow proline production from glutamate and glutamine (Wu et al., 2011). Therefore, dietary proline is essential for growth and health maintenance. Dietary protein bound proline and hydroxyproline are hydrolysed by specific proline peptidases in the gastrointestinal tract of the small intestine to yield free proline or hydroxyproline (Wu et al., 2011). Roughly 40% of the luminal proline is catabolised by enterocytes and bacteria with the remaining amount entering portal circulation.

Proline has previously been identified in the serum (108 to 760 µmol/L) and urine (11 to 219 µmol/L) of captive cheetahs using GC-MS (Tordiffe and Mienie, 2019). In urine, the fractional excretion rates of proline-hydroxyproline (5.5%) and glycine-proline (1.7%) were high compared to many other amino acids, while urinary glycine and proline-hydroxyproline concentrations also significantly decreased with age. The authors concluded that since the excretion of free and bound hydroxyproline reflect the breakdown of mature collagen, along with the high concentration of glycine and proline, this may indicate subclinical nutritional deficiencies in young growing cheetahs. In the current study, hydroxyproline (identified as trans-4-Hydroxy-L-proline) was found in the urine across all three diets, but was only identified as an important contributor to variation in PC 2 which would potentially be contributing to the control diet. The higher urinary excretion of proline in the control group may be linked to the sub-optimum diet and collagen breakdown because arginine and ornithine (from which proline can be synthesised) was found at similar concentrations across all three diets.

4.3.4. D-Fructose

Fructose, a simple sugar, was found in the urine of 8 and 10 cheetahs at baseline and on the glycine supplemented diet, respectively. However, it was only identified in three cheetahs on the control diet, which likely contributed to the significant difference found between the control and glycine supplemented diet. The significance of fructose is somewhat surprising considering the carnivorous nature and high protein diet of cheetahs. Fructose can be endogenously synthesised from glucose and sorbitol by the polyol pathway in mammals with its production and metabolism suggested in the pathogenesis of metabolic syndrome and renal disease (Andres-Hernando et al., 2019).

Domestic cats express a hepatic fructokinase enzyme, which metabolises dietary fructose at a higher level than canine liver (Schermerhorn, 2013). This high expression is unexpected considering the low quantity of fructose in the strict carnivore diets of cats, but does align with their ability to digest and absorb fructose in the intestine (Schermerhorn, 2013). In cheetahs, fructose and glucose are important fuel substrates for spermatozoa and are metabolised for energy (Weiner et al., 2019). While fructose has not been identified in relation to cheetahs in any other study, the use of fructose in spermatozoa highlights the potential role of this sugar in reproduction, which may indicate an overall metabolic significance. Cheetahs, like domestic cats, are reliant on the breakdown of gluconeogenic amino acids (such as glycine) and the movement of these products into gluconeogenic pathways to produce energy (Schermerhorn, 2013). Urinary fructose was higher in the glycine supplemented diet compared to the control diet, while glucose was higher in the control group compared to the glycine and baseline group. Glycine supplementation has been shown to play a role in glucose homeostasis by reducing glucose production, increasing glucose tolerance and improving insulin sensitivity (Imenshahidi and Hossenzadeh, 2022). This would explain why the glucose concentration was lower in the glycine and baseline diet, however the inverse link to fructose concentration is not apparent.

4.3.5. Dimethylamine (DMA) and Trimethylamine (TMA)

The secondary and tertiary amine organic compounds, namely DMA and TMA, were identified at low concentrations in the urine of all cheetahs at baseline (excluding two cheetahs for TMA), on the control and glycine supplemented diets. TMA can be synthesised from dietary choline, trimethylglycine, phosphatidylcholine and carnitine, and either oxidised to TMAO or degraded to DMA and formaldehyde by gut microbial enzymes (Chhibber-Goel et al., 2016). DMA can also be synthesised by the post translational modification of arginine in certain proteins which ultimately produces DMA amongst other compounds (Tsikas, 2020).

In the current study, the DMA concentration was significantly lower at baseline compared to both the control and glycine supplemented diets, which aligns with its noticeable contribution to the variation in the baseline group in the PLS-DA model. For TMA, the concentration was significantly higher on the control diet compared to both the baseline and the glycine diet. The concentration pattern of TMA matches DMA which is expected considering that TMA is a major precursor for DMA. Fish and seafood are an important exogenous source of DMA and TMA through choline, although in the cheetah diet the exogenous source would likely be from the carnitine present in high protein diets (Mitchell et al., 2008; Trimigno et al., 2020). In an early study conducted on choline and gut bacteria deficient rats, the amount of urinary DMA excreted was the same as control rats which indicate that choline is not an essential precursor and that intestinal bacteria were not essential for the formation of DMA (Zeisel et al., 1985). This highlights that there may be multiple pathways in mammals that allow the endogenous biosynthesis of DMA. Intermediary metabolic

pathways such as the breakdown of creatinine or the transmethylation of methylamine from sarcosine or glycine have been suggested as endogenous sources of DMA (Mitchell et al., 2008; Zeisel et al., 1985). In contrast, TMA excretion was significantly reduced in choline deficient and bacteria deficient rats, which indicates a reliance on choline and the gut microbiome (Zeisel et al., 1985).

The presence of DMA and TMA in urine have been associated with various diseases such as kidney damage, cardiovascular disease, diabetes, prostate cancer, rheumatoid arthritis and reduced bone volume (Baptista et al., 2020; Chhibber-Goel et al., 2016; Tsikas, 2020). Therefore, the increased presence of both DMA and TMA in the control diet may be as a result of the poor diet and the potential detrimental health effects thereof. These substances have not been identified in the urine of cheetahs, but TMA has been identified in the anal sac secretions of domestic cats (Banik et al., 2021). The authors concluded that cat urine has the potential to produce TMA with its distinctive "fishy" odour and this may contribute to scent marking. While both DMA and TMA have distinctive "fishy" odours and may play a role in the scent marking of cheetahs, the significantly higher concentrations on the control diet is interesting and indicates an effect by glycine. Choline chloride and dimethylglycine, which are precursors for TMA, were significantly higher in the control diet too and could be driving the increased production of TMA, but not necessarily that of DMA. Choline and dimethylglycine are also part of the glycine biosynthesis and degradation pathways and, therefore, when glycine is supplemented in the diet, it may potentially reduce the pressure on this pathway leading to a reduction in DMA, TMA and related metabolites. This could have major health benefits for the cheetah although the mechanisms are not yet clear. Importantly, the relative concentrations of DMA and TMA are very low for all three diets and, because it is not absolute, cannot be compared to literature that describes the concentration range of detrimental urinary associations to diseases.

4.3.6. Pyroglutamic acid

Pyroglutamic acid, also known as 5-oxoproline, is a cyclic amide of glutamic acid discovered in 1882 and is an intermediate metabolite in the glutathione cycle (Kumar and Bachhawat, 2012). Pyroglutamic acid is formed by the degradation of glutathione, the spontaneous cyclisation of phosphorylated glutamic acid or the degradation of proteins that contain pyroglutamic acid at the N terminus (Kumar and Bachhawat, 2012). Increased pyroglutamic acid excretion has been shown to indicate insufficient glycine availability in healthy humans after ingesting benzoic acid to deplete glycine stores (Jackson et al., 1987). Glycine was diverted to the conjugation of benzoic acid to form hippuric acid and diverted away from glutathione biosynthesis. In contrast, oral glycine supplementation in malnourished children led to decreased pyroglutamic acid excretion and increased blood glutathione concentration, which suggests that glycine is a rate limiting nutrient in growing children (Persaud et al., 1996). There is a clear link between urinary pyroglutamic acid,
glutathione biosynthesis (and therefore antioxidant status) and the role of glycine conjugation in these pathways (Meléndez-Hevia et al., 2009).

Pyroglutamic acid has previously been identified in the range of 20 to 75 µmol/L in the urine of captive cheetahs (Tordiffe and Mienie, 2019). In the current study, pyroglutamic acid was identified in the urine of all 10 cheetahs across the baseline, control and glycine supplemented diet. It contributed to the baseline diet in the PLS-DA model and was significantly lower at baseline compared to the control diet. This aligns with the abovementioned studies because the control diet would have no glycine present and, therefore, the urinary excretion of pyroglutamic acid is increased as an indicator of this deficiency. When glycine was added to their diet, the urinary excretion decreased (albeit not significantly) to levels closer to baseline. Although the cheetahs received a basic mince diet for 3 weeks prior to baseline, the low pyroglutamic acid concentration indicates the remnants of a more collagen rich diet with a potentially higher glycine content. This was also shown in Jackson et al. (1987) where individuals had different rates of glycine depletion (as indicated by increased benzoic acid ingestion) which the authors concluded may be due to the efficiency with which glycine can be mobilised for metabolic interaction. Therefore, three weeks on a sub-optimum diet would not be sufficient to naturally deplete the crucial glycine content in captive cheetahs, with four to seven weeks being the likely range as per the control diet. A four week glycine supplementation period was enough to reduce the pyroglutamic acid levels and a longer period or higher concentration may be more beneficial.

4.3.7. 1,3-Diaminopropane (1,3-DAP)

1,3-DAP is a naturally occurring polyamine along with other molecules like putrescine, cadaverine, spermidine and spermine (Moinard et al., 2005). Polyamines play a variety of physiological roles mostly in immune response such as cell growth, synthesis of proteins and nucleic acids, and cell signalling (Moinard et al., 2005). Putrescine is synthesised from ornithine by ornithine carboxylase (a rate limiting enzyme in polyamine biosynthesis) and gives rise to the other amines and their derivatives. 1,3-DAP is a close homologue of putrescine and *in vivo* acts as a repressor of ornithine carboxylase (Persson and Rosengren, 1984). The mechanisms of 1,3-DAP biosynthesis and breakdown are not well described in mammals, but in plants 1,3-DAP is synthesised from spermine and spermidine, which leads to the production of β -alanine, an important substrate for coenzyme A production (Parthasarathy et al., 2019). Increased urinary excretion of polyamines have been observed in catabolic conditions such as low calorie diets, multiple trauma and various cancers with increased plasma 1,3-DAP found in rat models of lung cancers (Liu et al., 2017; Moinard et al., 2005).

1,3-DAP was identified in the urine of all 10 cheetahs across the diets and was significantly higher at baseline compared to the control diet with the glycine diet having a similar distribution to the control diet. Polyamines can be obtained from meat and plant based food in the diet, but also from endogenous gastrointestinal secretions, desquamation of enterocytes and bacterial synthesis (Moinard et al., 2005). While the dietary influence on 1,3-DAP, as a specific polyamine, is not clear from the literature, it may be linked to the availability of ornithine and arginine which is essential in cats (MacDonald et al., 1984; Tordiffe and Mienie, 2019). If 1,3-DAP is an inhibitor of ornithine decarboxylase in cheetahs, the higher urinary concentration at baseline may indicate a greater production of 1,3-DAP which leads to the inhibition of the polyamine pathways, as putrescine production would decrease. However, putrescine was not identified in this study and while spermidine was identified, it was fairly constant across all three diets. Additionally, the higher 1,3-DAP levels at baseline may indicate that polyamines were adequately obtained from other exogenous or endogenous sources. In contrast, the lower DAP on the control diet and glycine diet (albeit not significantly) may allow additional production of putrescine and its related metabolites in order to elicit an immune response. Due to the production of polyamines from ornithine, the lower 1,3-DAP may also be linked to the significantly higher proline (which is also produced from ornithine) in the control diet.

4.3.8. Dihydrothymine

Dihydrothymine, derived from thymine, was identified in the urine of 5 cheetahs at baseline, 6 cheetahs on the control diet and 9 cheetahs on the glycine diet. It was not identified via ANOVA in MetaboAnalyst, presumably due to the lower identification of this metabolite in the baseline and control diet, which in itself is interesting. However, when taking a multivariate approach, dihydrothymine was the greatest contributor to the diet separation in the PLS-DA model, particularly driving the baseline and glycine diet.

Dihydrothymine is formed as an intermediate after the breakdown of thymine by dihydropyrimidine dehydrogenase, ultimately forming β -aminoisobutyric acid by dihydropyrimidinase (DHP) (Löffler et al., 2005). As pyrimidines, thymine along with uracil and cytosine play an essential role in mammalian metabolism as nucleotide bases in DNA and RNA, activators in phospholipid and polysaccharide synthesis, and glycosylation of proteins and lipids (Löffler et al., 2005). There are numerous inborn errors of pyrimidine metabolism, of which high levels of urinary dihydrothymine is associated with a defect in the *DHP* gene with symptoms such as neurological abnormalities, seizures, impaired growth and gastrointestinal problems (Löffler et al., 2005; Philip et al., 2012). A single case of this condition has been reported in a malnourished stray male cat who presented with high amounts of urinary dihydrouracil and dihydrothymine, and moderate amounts of uracil and thymine, suggesting a DHP deficiency (Philip et al., 2012). In the current study, data suggests that dihydrothymine is playing a role in relation to the glycine diet compared to the baseline diet. Thymine biosynthesis leads to an equimolar glycine production of roughly 6.2 mmol/day (Meléndez-Hevia et al., 2009). Therefore, the additional glycine supplementation potentially leads to a build-

74

up of thymine, which leads to a greater excretion of dihydrothymine. Similarly, *de novo* biosynthesis of pyrimidines occur from glutamine in the cytosol and cheetahs have been shown to have a high urine concentration of glutamine (Tordiffe and Mienie, 2019). Although not significant, the current study demonstrated that glutamine was also an abundant metabolite showing a higher trend in the control and glycine supplemented diet compared to baseline. Glycine plays a key role in purine (such as adenine and guanine) production, which works alongside pyrimidines to produce DNA and RNA. This potentially points to a need for increased pyrimidine biosynthesis to match the purine biosynthesis (Meléndez-Hevia et al., 2009). The link between dihydrothymine and glycine is not clear, but it may be linked to excess glutamine or purine biosynthesis.

4.3.9. Methylmalonic acid (MMA)

MMA is a dicarboxylic acid that is a vital intermediate in fat and protein metabolism. Specifically, the catabolism of isoleucine, valine, methionine, and threonine as well as of odd chain fatty acids and cholesterol to propionate, ultimately leading to the entry of their metabolites into the Citric Acid Cycle (Hörster and Hoffmann, 2004). Methylmalonyl-CoA mutase (MCM) is an essential mitochondrial enzyme in this process. Defects in this enzyme lead to an accumulation of MMA in the urine and biofluids, called methylmalonic aciduria which is characterised by vomiting, lethargy, acidosis, ketosis, and coma (Hörster and Hoffmann, 2004). Vitamin B₁₂ (generally obtained from dietary animal tissue) produces an essential cofactor for MCM, which catalyses the conversion of methylmalonyl-CoA to succinyl-CoA. Therefore, MMA and vitamin B₁₂ have an inverse relationship often being used as diagnostic markers of vitamin B₁₂ deficiency, methylmalonic aciduria and other gastrointestinal conditions (Sobczyńska-Malefora et al., 2021).

MMA was identified in the urine of all 10 cheetahs across the diets with cheetahs on the glycine supplemented diet having a significantly lower concentration than at baseline. This indicates that MCM is functioning optimally and more metabolites of the respective protein and fat metabolism are able to enter the Citric Acid Cycle for energy production. Glycine plays an important role in the conjugation of detrimental by-products resulting from the bacterial fermentation of high protein diets. This improved conjugation may lead to better absorption of vitamin B₁₂ and catabolism of excess proteins to succinyl-CoA (Badenhorst et al., 2014; Sobczyńska-Malefora et al., 2021). MMA and vitamin B₁₂ (along with other metabolites) have been suggested as biomarkers of gastrointestinal disease in cheetahs due to the high prevalence of these disease in this population and its success as markers in domestic cats and dogs (Fox et al., 2021). While the results of the study were largely inconclusive, it was the first study investigating potential markers in cheetahs. The absolute concentrations of MMA were not determined in the current study, but the mechanism of glycine in reducing the relative concentrations warrants further investigation as this could contribute to the knowledge of their gastrointestinal health.

4.3.10. Pimelic acid

Pimelic acid (also known as α, ω -heptanedioic acid) was found in the urine of all 10 cheetahs across the baseline, control and glycine supplemented diet. A significant difference between diets was not identified by post hoc analysis, but pimelic acid contributed to the PLS-DA model specifically driving the baseline and glycine diets. Pimelic acid, a simple carboxylic acid, is the precursor for biotin (vitamin B₇) biosynthesis in microbes, fungi and plants with mammals obtaining biotin from their diet and intestinal microflora (Lin and Cronan, 2011). The biosynthesis of pimelic acid is not fully understood, but believed to be synthesised by a modified fatty acid synthetic pathway (Lin and Cronan, 2011). Biotin is an essential, water soluble vitamin that can be obtained from natural food such as meat, eggs, liver and vegetables (Zempleni et al., 2009). In the current study, biotin was not specified in the nutrient breakdown of horse meat consumed by the cheetahs, but it was provided in the supplement at 100 µg/day throughout the study (Table 1). Biotin plays a role in cell signalling, epigenetic regulation of genes, chromatin structure and as a covalently bound coenzyme for five carboxylases (Zempleni et al., 2009). Acetyl-CoA carboxylase 1 and 2 are involved in fatty acid synthesis and oxidation, and pyruvate carboxylase is a key enzyme in gluconeogenesis. Propionyl-CoA carboxylase catalyses the metabolism certain amino acids and fatty acids to the citric acid cycle and β-Methylcrotonyl-CoA carboxylase which catalyses an important step in leucine metabolism (Zempleni et al., 2009). All these pathways are crucial in energy production, with propionyl-CoA carboxylase closely linked to the discussion of MMA as it is the first enzyme to catalyse propionyl-CoA (propionate) to methylmalonyl-CoA (Hörster and Hoffmann, 2004). Therefore, increased pimelic acid may lead to increased propionyl-CoA carboxylase activation through biotin biosynthesis.

Pimelic acid has previously been identified in the urine of cheetahs at $4.5 \pm 3.0 \text{ mmol/L}$ (Tordiffe et al., 2017). At baseline, the pimelic acid concentrations were higher than both the control and glycine diet, with the difference between the baseline and control diet likely contributing to the overall significance. A greater excretion of pimelic acid may indicate sufficient biotin being obtained from the diet, and therefore a reduced need for biosynthesis of biotin from pimelic acid. On the control diet, the lower excretion may indicate the need for increased biotin biosynthesis from pimelic acid by gastrointestinal bacteria. The direct biotin consumed by the cheetahs were constant throughout the study. This finding indicates that remnants of a collagen rich diet (baseline) and glycine supplementation are influencing the production of pimelic acid or the biosynthesis of biotin by the microflora. Cats and dogs have a similar intestinal microflora to humans with changes in the microbial communities shown to be associated with acute and chronic gastrointestinal disease (Honneffer et al., 2014). Although the specific microbiome of cheetahs is still unknown, glycine may be conjugating toxic by-products influencing the production of pimelic acids and other microbial products.

4.4. Summary

A total of 151 metabolites were identified in the urine of 10 captive cheetahs at baseline, on a control diet and on a glycine supplemented diet. This was reduced to 10 metabolites that had the greatest contribution towards the metabolic differences between the diets. The change in proline, pyroglutamic acid and 1,3-diaminopropane at baseline likely reflect the influence of a historic collagen diet and not glycine supplementation. Glycine supplementation had a similar effect to the baseline diet for dimethyl sulphone, trimethylamine, dimethylamine and pimelic acid – which indicates that glycine may be responsible for the beneficial effects on the gut microbiome. On the other hand, the effects of glycine supplementation exceeded the baseline diet for fructose, dihydrothymine and methylmalonic acid. This indicates an important role of glycine in energy metabolism, gastrointestinal health and pyrimidine biosynthesis for captive cheetahs.

CHAPTER FIVE

5. SERUM METABOLITES

5.1. Introduction

Blood is the primary transporter of molecules around the body and is composed of two components - the cellular part (contains RBC, WBC and platelets) and the liquid plasma (Psychogios et al., 2011). Plasma is obtained from a blood sample when an anticoagulant is added, while serum is obtained when blood is allowed to clot and therefore the resulting supernatant contains no clotting factors (Yu et al., 2011). Both plasma and serum contain a variety of substances including proteins, nutrients, electrolytes and metabolic by-products, and serves as an indicator of all the different molecules being secreted and excreted by surrounding organs and tissues in normal and pathophysiological conditions (Psychogios et al., 2011). Due to its importance and ease of accessibility, the chemical analysis of blood has been performed for over 70 years generating normal reference ranges for many blood gases, ions and metabolites of humans and other animals. More recently, metabolomics techniques have been utilised to characterise the blood metabolome with 4229 compounds confirmed in humans (Psychogios et al., 2011). Plasma and serum metabolite analysis are both reproducible and produce equally relevant results for clinical and biological studies, although higher concentrations are found in serum (Yu et al., 2011). The metabolomics focus in animals is more limited with approximately 50 studies across different species, generally domestic animals and rodents, most of which utilised serum, plasma or urine samples (Carlos et al., 2020; Tran et al., 2020).

Although several studies have investigated a number of blood parameters in cheetahs, only two from the same laboratory have used metabolomics techniques (Tordiffe et al., 2016; Tordiffe and Mienie, 2019). The former study investigated the serum fatty acid profiles of captive and free ranging cheetahs using GC-MS. It was found that serum monounsaturated and polyunsaturated fatty acids were lower in free ranging cheetahs, likely due to the differences in dietary fat intake and composition. Tordiffe and Mienie (2019) identified 36 serum amino acids with glutamine (1624 μ mol/L), alanine (772 μ mol/L), arginine (670 μ mol/L) and glycine (479 μ mol/L) being the highest concentrations in the cheetah. These studies form the foundation of serum metabolite data for captive and free ranging cheetahs.

Glycine along with the many products related to glycine metabolic pathways have been identified in the blood of humans, pigs, rodents and birds (El Hafidi et al., 2004; Gannon et al., 2002; Imenshahidi and Hossenzadeh, 2022; Wang et al., 2014, 2013). Therefore, using a metabolomics approach to investigate the effect of glycine supplementation on the serum of cheetahs would be beneficial especially considering the high concentration of metabolically active compounds, the use of serum in nutritional studies and the limited data available on cheetah serum metabolites. The aims of this chapter are to:

- identify and quantify (through relative concentrations) the metabolites present in the serum of 10 captive cheetahs at baseline, on a control diet and on a glycine supplemented diet;
- highlight and discuss the identified metabolites of importance.

5.2. Results

5.2.1. Identified metabolites and relative concentration

ASICS identified 60 metabolites in the serum across the 10 cheetahs in each diet. These were reduced to 48 metabolites after data reduction (20% reduction) which removed metabolites present in less than 50% of the samples in all three groups. The relative concentrations of the 20 most abundant metabolites are reported in Table 9. D-glucose was the most abundant metabolite identified in the serum across all three groups – with approximately five to six times the concentration of the next metabolite. L-glutamine, lactic acid, taurine and guanidinoacetic acid followed glucose as the most abundant for each diet. The same 20 metabolites were consistently identified across all three diets, with the only exception of L-lysine which was only abundant in the control diet. All 48 metabolites are shown in Supplementary Table S5.

Baseline			Control diet			Glycine supplemented diet			
Metabolite	n	Relative concentration	Metabolite	n	Relative concentration	Metabolite	n	Relative concentration	
D-Glucose	10	23.9 ± 2.9	D-Glucose	10	25.0 ± 5.4	D-Glucose	10	23.4 ± 3.0	
L-Glutamine	10	4.9 ± 0.6	L-Glutamine	10	4.9 ± 0.5	L-Glutamine	10	4.7 ± 0.4	
Lactic acid	10	3.5 ± 0.6	Lactic acid	10	3.7 ± 0.4	Lactic acid	10	3.8 ± 0.5	
Taurine	10	2.7 ± 1.0	Taurine	10	3.5 ± 0.9	Taurine	10	2.9 ± 0.9	
Guanidinoacetic acid	10	2.7 ± 0.6	Guanidinoacetic acid	10	2.9 ± 0.8	Guanidinoacetic acid	10	2.6 ± 0.5	
D-Mannose	10	2.6 ± 0.3	D-Mannose	10	2.4 ± 0.3	G-6-P	10	2.5 ± 0.2	
G-6-P	10	2.5 ± 0.3	D-Glucuronic acid	10	2.4 ± 0.5	D-Mannose	10	2.4 ± 0.3	
Glyceric acid	10	2.3 ± 0.4	G-6-P	10	2.3 ± 0.2	D-Glucuronic acid	10	2.2 ± 0.4	
D-Glucuronic acid	10	2.3 ± 0.4	Glyceric acid	9	2.3 ± 0.4	Glyceric acid	10	2.2 ± 0.4	
D-Galactose	9	1.9 ± 0.3	D-Galactose	7	2.0 ± 0.3	D-Galactose	10	1.8 ± 0.3	
L-Carnitine	10	1.8 ± 0.4	L-Alanine	10	1.7 ± 0.2	L-Alanine	10	1.7 ± 0.3	

Table 9. The 20 most abundant metabolites identified by NMR in the serum of 10 captive cheetahs at baseline, on a control diet and on a glycine supplemented diet

L-Alanine	10	1.7 ± 0.2	L-Carnitine	10	1.7 ± 0.6	L-Carnitine	10	1.6 ± 0.5
D-Maltose	10	1.6 ± 0.2	D-Maltose	9	1.6 ± 0.2	D-Maltose	10	1.6 ± 0.1
L-Glycine	10	1.5 ± 0.3	L-Glycine	10	1.5 ± 0.5	L-Cystine	10	1.5 ± 0.2
L-Cystine	10	1.5 ± 0.1	L-Cystine	10	1.5 ± 0.2	L-Glycine	10	1.4 ± 0.3
D-Fructose	7	1.2 ± 0.4	D-Fructose	7	1.5 ± 0.4	L-Proline	10	1.3 ± 0.2
L-Proline	10	1.2 ± 0.2	Ethanolamine	10	1.2 ± 0.3	Ethanolamine	10	1.2 ± 0.2
Ethanolamine	10	1.1 ± 0.1	L-Proline	10	1.2 ± 0.3	Creatinine	10	1.1 ± 0.2
Creatinine	10	1.1 ± 0.2	L-Lysine	6	1.1 ± 0.2	D-Sorbitol	8	1.1 ± 0.2
D-Sorbitol	9	1.0 ± 0.2	Creatinine	10	1.1 ± 0.1	D-Fructose	6	1.1 ± 0.4

Data are expressed as mean \pm SD. n indicates the number of samples where the metabolite was detected. Relative concentration is an arbitrary unit. G-6-P, D-Glucose-6-Phosphate

5.2.2. <u>Metabolites identified with univariate analysis</u>

MetaboAnalyst did not identify any significant metabolites after FDR correction in the ANOVA. Only L-glutamic acid (F = 6.14, raw p = 0.01, adjusted p = 0.42) and α -aminobutyric acid (F = 3.92, raw p = 0.04, adjusted p = 0.65) were significant prior to the adjustment.

5.2.3. Metabolites identified with multivariate analysis

The PCA and PLS-DA plots are shown in Figure 6. PC 1 accounted for 32% variation in the serum data, while PC 2 accounted for 14%. Most of the data for all three diets are clustered, with three clear outliers – the baseline and control diet outliers represent the same individual. The 10 metabolites contributing the most to the variation of PC 1 and PC 2 are reported in Table 10. L-glycine, hypotaurine, L-carnitine, D-glucose and guanidinoacetic acid had the highest loading scores for PC 1 and, therefore, were the metabolites that had the highest weighting contribution towards the variation in PC 1. Based on the PCA plot, the metabolites contributing to PC 1 are driving most of the variation across the baseline, control and glycine diet. The metabolites contributing the most to PC 2 are lactic acid, L-threonine, L-valine, L-lysine and L-glutamine.

In the PLS-DA plot, there is a clear overlap with the baseline, control and glycine diet. This indicates poor separation of the diets in the model, which is supported by the negative Q2 values (Supplementary Figure S1) indicating that the model lacks predictive power. The discriminant variables of component 1 account for 15.4% of the variation between diets, while component 2 is 23.2%. Therefore, component 2 is a better axis for maximising the separation between diets, which

is not ideal and, again, highlights the limited predictability of this PLS-DA model. The VIP and coefficient scores for each diet is reported in Table 10. The top VIP in component 1 and 2 were Lglutamic acid, propionic acid and α -aminobutyric acid. Similarly, L-glutamic acid had the highest coefficient of regression for the baseline, control and glycine diets. This was followed by α aminobutyric acid and propionic acid (baseline), taurine and ethanolamine (control), taurine and Dglucose-6-phosphate (G-6-P) (glycine supplemented).



Figure 6. A two dimensional PCA (top) and PLS-DA (bottom) scores plot of the metabolites identified in the serum of 10 captive cheetahs at baseline, on the control and glycine supplemented diets. *Each point represents an individual cheetah at baseline (red), control (blue) and glycine supplemented (green) with a corresponding 95% confidence ellipse for each diet.*

Table 10. The (A) PCA loadings, (B) PLS-DA VIP scores and (C) PLS-DA diet regression coefficients for the top 10 serum metabolites in 10 captive cheetahs

A. PC	A Meta	bolite loadings		B. PLS-DA Metabolite VIP scores					
PC 1		PC 2		C 1		C 2			
L-Glycine	0.25	Lactic acid	0.29	L-Glutamic acid	2.89	L-Glutamic acid	2.63		
Hypotaurine	0.25	L-Threonine	0.28	Propionic acid	2.49	Propionic acid	2.12		
L-Carnitine	0.24	L-Valine	0.28	α -Aminobutyric acid	1.94	α-Aminobutyric acid	1.65		
D-Glucose	0.24	L-Lysine	0.28	Lactose	1.75	Lactose	1.54		
Guanidinoacetic acid	0.24	L-Glutamine	-0.27	Choline chloride	1.75	Choline chloride	1.50		
L-Cystine	0.24	Cadaverine	-0.25	L-Threonine	1.61	L-Threonine	1.36		
D-Glucuronic acid	0.24	Creatine	0.24	D-Mannose	1.49	D-Mannose	1.34		
Ethanolamine	0.23	Creatinine	-0.24	L-Methionine	1.46	L-Methionine	1.30		
N-Acetylglycine	0.22	D-Gluconic acid	-0.22	Lactic acid	1.09	Ethanolamine	1.28		
Propylene glycol	0.22	Phenethylamine	-0.18	L-Lysine	1.02	G-6-P	1.02		

C. PLS-DA metabolite coefficients for each diet								
Baseline		Control		Glycine				
L-Glutamic acid	100.0	L-Glutamic acid	47.6	L-Glutamic acid	32.6			
α-Aminobutyric acid	66.1	Taurine	40.4	Taurine	23.6			
Propionic acid	63.2	Ethanolamine	35.5	G-6-P	20.8			
Choline chloride	55.8	G-6-P	35.2	Ethanolamine	20.3			
L-Methionine	54.9	L-Threonine	34.3	D-Maltose	19.5			
Cadaverine	44.8	L-Valine	34.1	L-Threonine	18.7			
Taurine	44.5	D-Maltose	33.4	L-Methionine	17.8			
L-Lysine	44.1	Ascorbic acid	29.0	Propionic acid	17.6			
D-Mannose	43.3	D-Galactose	28.8	L-Valine	17.5			
G-6-P	41.5	D-Glucuronic acid	26.6	D-Galactose	16.1			

Loadings are the weight of each metabolite and are expressed as positive or negative number depending on its direction from the PC. Coefficients are the weighted sum of the PLS-DA regression for each group. PCA, principle component analysis; PLS-DA, partial least squares discriminant analysis, VIP, variable importance in projection; PC, principle component; C, component. G-6-P, D-Glucose-6-phosphate

5.2.4. Summary of important metabolites

The 48 serum metabolites identified were reduced to 32 metabolites that were present in the top 20% of at least one component of the univariate or multivariate analysis (Supplementary Table S6). This was further reduced to seven metabolites that were present in at least four out of the eight analyses. The seven important serum metabolites were L-glutamic acid, L-threonine, α -aminobutyric acid, G-6-P, ethanolamine, L-methionine and propionic acid. The normalised concentrations of these seven metabolites are shown in Figure 7. None of the metabolites were significantly different between diets after FDR adjustment, but they still contributed to the variation

observed between the diets. α-Aminobutyric acid, G-6-P and ethanolamine were identified in all 10 cheetahs across the three diets. However, the remaining metabolites were not consistently identified across diets – most notably, L-glutamic acid was identified in nine cheetahs at baseline, three on the control diet and two on the glycine diet.



Figure 7. The normalised concentration of 7 potentially important serum metabolites identified in 10 captive cheetahs at baseline (red), on a control diet (blue) and on a glycine supplemented diet (green). The dots represent individual values, the bar represents the mean normalised concentration and the grey line represents one standard deviation from the mean. Zero values were replaced by 20% of the lowest value during data scaling. Normalised concentrations may be a positive or negative number because it is scaled relative to the metabolite mean for all samples.

Glycine

Baseline

Control

Glycine

-2.0

Baseline

0

Control

Glycine

Baselin

Control

5.3. Discussion

5.3.1. Overview of identified metabolites in serum

A total of 60 metabolites were identified in the serum of cheetahs, of which 12 were removed after data cleaning, resulting in a final number of 48 metabolites. This aligns with previous NMR data on humans where approximately 49 compounds were identified on average in human serum (Psychogios et al., 2011). Blood is the primary transporter of molecules throughout the body, and is therefore vital for homeostasis. Considering this homeostatic importance, the identification of less metabolites and the lower number of univariate significantly different metabolites in the serum of cheetahs compared to the urine, is expected. Additionally, the cross-over design allowing each cheetah to have repeated measures across the baseline, control and glycine diets and the generally conservative endogenous metabolome (which is highly responsive to external stimuli such as diet) likely contribute towards the noticeable metabolic similarities between dietary groups (Wishart, 2019). This is emphasised by the clear overlap in the PCA and PLS-DA graphs which conventionally enhances the differences or separation between groups (Figure 6).

Glucose was the most abundant metabolite identified in the serum of the cheetahs in the current study. It has previously been identified at varying concentrations in cheetahs - 6.3 ± 0.2 mmol/L (Bechert et al., 2002) and 8.5 ± 1.6 mmol/L and 11.8 ± 2.9 mmol/L with different immobilisation drugs (Buck et al., 2022). This high concentration of glucose in cheetahs aligns with what has been found using NMR in healthy humans with glucose having the second highest serum concentration (5 mmol/L) after urea (6 mmol/L) (Psychogios et al., 2011). Due to potential spectrum interference with other metabolites, urea was excluded from the spectra analysis in the current study. Glutamine was the most abundant amino acid identified in the serum, and has previously been found in the serum of cheetahs at a concentration of 1.6 ± 0.6 mmol/L (Tordiffe and Mienie, 2019).

Despite the lack of significance found using univariate methods, the PCA and PLS-DA analysis highlighted metabolites in the serum that were contributing towards potential variation between the components and diets. This analysis was used to reduce the 48 metabolites to 7 metabolites of interest which will be discussed in the sections below as well as their potential association with glycine supplementation.

5.3.2. L-Glutamic acid

Glutamic acid, often called by its stable anion glutamate, is a highly abundant amino acid with critical roles in metabolism, signalling and nutrition (Brosnan and Brosnan, 2013). Dietary glutamic acid is used almost exclusively as a metabolic fuel that is completely oxidised to CO_2 in the intestines of rats, pigs and humans (Brosnan and Brosnan, 2013). Therefore, *in vivo* biosynthesis of glutamic acid is essential and circulating glutamic acid is tightly regulated at low concentrations. Firstly, glutamic acid can be synthesised from α -ketoglutaric acid, a citric acid cycle intermediate,

by glutamate dehydrogenase and this process is reversible. Secondly, glutamic acid can be synthesised from the other "glutamate family" amino acids glutamine, arginine, proline and histidine (Brosnan and Brosnan, 2013). Along with protein synthesis, glutamic acid has many functions such as the direct production of glutathione and as a major excitatory neurotransmitter in the central nervous system (CNS) with its product y-aminobutyric acid (GABA) being a major inhibitor (Brosnan and Brosnan, 2013; Wu, 2009). Glycine and glutamic acid are co-agonists for the N-methyl-Daspartic acid (NMDA) receptors which are ligand gated ion channels distributed throughout the CNS for fast excitatory synaptic transmission (Yu and Lau, 2018). These receptors are needed for learning and memory and are drug targets for treating Alzheimer's disease, depression, epilepsy and schizophrenia (Yu and Lau, 2018). Indeed, serum glutamic acid concentration was higher in schizophrenic patients compared to healthy controls which may be due to an imbalance in the dopamine and glutamate systems and hypofunction of the NMDA receptors (Ivanova et al., 2014). In a recent study using HPLC, glutamate, glutamine and glycine were measured in the serum and CSF of the clinical spectrum of Alzheimer's patients (Nuzzo et al., 2021). It was found that glutamate increased in pre-clinical Alzheimer's patients compared to healthy controls, while serum glycine remained constant across the clinical spectrum. The former finding suggests that glutamate concentration tends to decrease with disease progression and that glycine may not be an ideal serum or CSF biomarker for Alzheimer's disease. Although these findings do not directly influence the current study, it indicates a link between glycine and glutamate albeit at a neurological level (Nuzzo et al., 2021; Yu and Lau, 2018).

In the current study, glutamic acid was identified in the serum of nine cheetahs at baseline, and only three and two cheetahs on the control and glycine diet, respectively. While glutamic acid was not significantly different between groups after FDR adjustment, prior to that adjustment, it was significantly higher in the baseline group compared to the control and glycine diet, likely due to the difference in identification. Glutamic acid was the biggest contributor to the VIP scores in the separation of the diets, contributing primarily to the baseline diet. Glutamic acid has previously been identified in the serum of cheetahs at a concentration of $174 \pm 67 \mu mol/L$ (Tordiffe and Mienie, 2019).

Considering the conventional abundance and strict regulation of glutamic acid, it was surprising that this amino acid was not identified in most of the cheetahs on the control and glycine diet. However, it may be present at very low concentrations that were not detected with ASICS (Tardivel et al., 2017). Glutamine concentrations remained constant between the groups so it is unlikely that any change in glutamic acid is linked to a change in glutamine. Proline was the only pre-cursor to be identified in the serum, but there was no change in concentration. Conversely, proline had a significantly lower urinary excretion at baseline which may indicate greater utilisation in other pathways such as glutamate synthesis. Indeed cats lack pyrroline-5-carboxylate synthase which is needed to convert glutamic acid to proline, therefore, allowing a greater build-up of glutamic acid

or utilisation in other pathways (Wu et al. 2011). Due to the similarity between the control and glycine group, it is unlikely that any change is due to the glycine supplementation but rather the detrimental effects of a sub-optimum diet. Based on the studies in Alzheimer's and schizophrenia, the seemingly lower glutamic acid concentration may be a positive indicator of normal neurological function. Glutamic acid and glycine are required for glutathione, an important antioxidant, synthesis and reduction in glutamic acid may be an indicator of increased glutathione production due to the sub-optimum diet, even though glutathione was not identified in the cheetah serum in the current study. Pyroglutamic acid, which is an intermediate of glutathione production and is produced from glutamic acid, was highlighted as an important urinary metabolite that was significantly lower at baseline. Glutamic acid is a highly functional amino acid and the apparent difference in serum may be linked to multiple metabolic roles, such as glutathione production or energy production through α -ketoglutaric acid.

5.3.3. L-Threonine

Threonine is an essential amino acid that cannot be produced endogenously and must be obtained from the diet (Tang et al., 2021). Although not significantly different, threonine was identified at a seemingly higher concentration in the serum of ten cheetahs on the glycine supplemented diet as opposed to eight cheetahs at baseline and on the control diet, respectively. Threonine contributed more to the control and glycine diets, according to the PLS-DA model, compared to the baseline diet. Threonine plays a crucial role in lipid metabolism, protein synthesis (particularly of mucins in epithelial tissue) and maintaining intestinal health (Tang et al., 2021). Catabolism of dietary threonine occurs via three pathways. Firstly, threonine is degraded to α -ketobutyric acid and ammonia by liver threonine dehydratase which is then decarboxylated to propionyl-CoA and enters the citric acid cycle as succinyl-CoA - this is the major pathway of threonine degradation accounting for 93% of the total flux in humans (Meléndez-Hevia et al., 2009; Tang et al., 2021). Secondly, threonine dehydrogenase converts threonine to the unstable 2-amino-3-oxybutyrate which is degraded to glycine and acetyl-CoA, the latter entering the citric acid cycle for energy. This pathway only accounts for 7% of the threonine flux equating to approximately 0.3 mmol/day of glycine in adult humans (Meléndez-Hevia et al., 2009). Lastly, threonine is degraded to acetaldehyde and glycine by threonine aldolase, however, this enzyme is inactive in mammals and, therefore, this pathway is unlikely to produce glycine in the cheetah. Threonine has been identified in the serum of cheetahs at a concentration of $154 \pm 67 \mu mol/L$ (Tordiffe and Mienie, 2019).

In broiler chickens, supplemented glycine has been shown to reduce the activity of all three catabolic threonine enzymes (threonine aldolase, threonine dehydrogenase and threonine dehydratase) increasing the amount of available threonine for physiological processes (Bernardino et al., 2011). In a study by Ospina-Rojas et al. (2013), growing broiler chickens on a low protein diet were supplemented with different concentrations of glycine and serine (which is a precursor for

glycine biosynthesis) and digestible threonine. Higher concentrations of glycine and serine led to improved growth which was enhanced by greater threonine supplementation suggesting that threonine may reduce the requirement for dietary glycine. Increasing levels of glycine and serine also increased intestinal mucosa secretion, which may directly be attributed to the glycine-rich central domain of mucin or indirectly due to the improved integrity of the gastrointestinal tract by preventing threonine degradation. Reducing threonine degradation may result in reducing the formation of volatile fatty acids (e.g. acetic acid, propionic acid, and butyric acid) by commensal bacteria, therefore, improving intestinal function (Tang et al., 2021). This indicates that glycine supplementation may be essential for the re-synthesis of mucin, which has a high intestinal turnover, thereby supporting the direct role of threonine in mucin production (Ospina-Rojas et al., 2013). In thoroughbred colts and mares that received threonine supplementation for five days, plasma glycine concentration was significantly higher on the supplemented diet compared to the cross-over control diet which likely reflects the catabolism of threonine to glycine (Mastellar et al., 2016). These studies highlight the connection between glycine and threonine, especially regarding mucin biosynthesis.

Although the higher relative concentration of the glycine supplemented diet was not significantly different to the control or baseline, the trend in the current study supports what has been found in earlier studies. As threonine can only be obtained from the diet, the only direct source for the cheetahs would be the horse meat consumed which would contain approximately 0.84 g threonine/100g meat (Table 1). The amount of meat consumed in the current study remained fairly constant and, therefore, glycine supplementation may have a sparing effect on threonine. This would then allow more threonine to be directed towards propionyl-CoA for energy in the citric acid cycle or towards intestinal mucin production which would improve gastrointestinal health. Although the threonine catabolic enzyme activities were not investigated in the current study, a reduction in their respective activities may account for the increase in threonine with glycine supplementation, as was reported for broiler chickens (Bernardino et al., 2011). Overall, an increase in threonine concentration has been associated with improved gastrointestinal health, which may be crucial for cheetahs in captivity.

5.3.4. <u>α-Aminobutyric acid (AABA)</u>

ABBA is a non-proteinogenic amino acid synthesized from methionine and threonine degradation to α -ketobutyric acid (Chiarla et al., 2011). α -Ketobutyric acid can either be converted to AABA or decarboxylated to propionic acid to enter the citric acid cycle (Chiarla et al., 2011; Tang et al., 2021). The well-known isomer of AABA is GABA, which is a major inhibitory neurotransmitter in the CNS (Wang et al., 2020). Although the exact function and pathways of ABBA are poorly understood, it has been suggested as a biomarker in many conditions including severe sepsis, liver disease, osteoporosis and COVID-19 in humans (Atila et al., 2021; Chiarla et al., 2011; Wang et al., 2020).

In mice, it was found that AABA increases intracellular glutathione production through AMPK, therefore, regulating oxidative stress defence (Irino et al., 2016).

ABBA has previously been identified in the serum of cheetahs at 22 ± 12 µmol/L (Tordiffe and Mienie, 2019). In the current study, ABBA was identified at low relative concentrations in the serum of all 10 cheetahs across diets. Prior to adjustment, significance was found between diets in which the baseline diet appeared to be higher than the control and glycine supplemented diets. ABBA had a high VIP score for both component 1 and 2, which indicates its contribution to the PLS-DA model. The amount of threonine and methionine consumed by the cheetahs would have been consistent for all three diets, and the similarity between the control and glycine groups indicate that glycine supplementation may not be directly affecting the production of ABBA. In sepsis patients, there was a positive correlation between high ABBA concentrations and increasing glycine concentrations (Chiarla et al., 2011). During the conversion of choline to glycine, homocysteine utilises a methyl group to convert to methionine – this indicates a link between a glycine synthesis pathway and ABBA production from methionine (Wang et al., 2013). Additionally, glycine can be synthesised from threonine which can also produce ABBA. Therefore, ABBA could potentially increase or decrease depending on whether the endogenous or exogenous glycine is utilised, respectively. This could attribute to the consistent concentrations for the control and glycine diets. However, the baseline diet which likely contains a wider diversity of amino acids may lead to a greater production of ABBA. This metabolite appears to be an indicator of the catabolism of methionine (for which cats have a high demand) and threonine, although studies have also suggested ABBA as an indicator for glutathione production, and does not appear to have a clear functional role especially regarding glycine supplementation.

5.3.5. D-Glucose-6-Phosphte (G-6-P)

G-6-P was identified in the serum of all 10 cheetahs at baseline, on the control and glycine diet. It contributed towards the baseline and control diet PLS-DA models, and indeed appears to have a higher relative concentration at baseline compared to the control diet. G-6-P is an important saccharide formed when glucose, from the blood, enters the cell (such as hepatocytes) and is immediately phosphorylated to G-6-P to prevent the diffusion of free glucose out of the cell (Rajas et al., 2019). G-6-P has many intracellular fates including glycolysis, pentose phosphate pathway, glycogen synthesis and glucose production (Rajas et al., 2019). For carnivores, such as domestic cats and cheetahs, there is a close link between protein and glucose metabolism because the former provides substrates that channel into gluconeogenic pathways (Schermerhorn, 2013). Unlike other mammals, cats lack glucokinase and it is hypothesized that the conversion of intracellular glucose to G-6-P is driven by hexokinase I (Schermerhorn, 2013). Hexokinase has kinetic features which make it an unsuitable replacement for glucokinase, largely due to its inhibition by its product, G-6-P, which is continuously being produced by gluconeogenesis in feline

hepatocytes. Therefore, it is believed that there is a "pull effect" where G-6-P is being redirected away from gluconeogenic pathways and towards glycogen biosynthesis. This enhances G-6-P flux and reactivates hexokinase which allows an increase in glucose phosphorylation (Schermerhorn, 2013). The efficient phosphorylation of hepatic glucose is an area that requires more research.

G-6-P does not appear to be regularly identified in the blood of humans or animals, with glucose and G-6-P dehydrogenase or related enzymes being the target of investigations. Whether this is due to its primarily intracellular role or simply because it is not considered a conventional indicator of glucose metabolism, is not apparent. However, in a recent study the serum concentrations of various simple sugars and their derivatives were investigated in Humboldt penguins using GC-MS (Schaeffer et al., 2020). G-6-P was identified as a key sugar in discriminating between different nesting habitats presumably due to different energy requirements. In the current study, the seemingly higher G-6-P concentrations at baseline followed by the glycine diet, may be due to the gluconeogenic role of glycine particularly with pyruvate production (Wang et al., 2013). The serum glucose concentration remained constant across diets, although the glucose excretion was significantly higher on the control diet compared to the baseline and glycine diet indicating higher production on the control diet. This indicates an inverse relationship between glucose and G-6-P which aligns with its interconversion. Additionally, glycine is directly involved in the biosynthesis of creatine which is important for rapid energy production and glycine supplementation has been shown to reduce glucose production and insulin sensitivity (Imenshahidi and Hossenzadeh, 2022; Wang et al., 2013). Therefore, glycine supplementation may promote other pathways of energy production such as creatine and glycolysis, and reduce the need for endogenous glucose production. This reduced pressure may decrease the flux through the glucose and G-6-P pathways which leads to a build-up of G-6-P that further inhibits the activity of hexokinase. The hepatic enzyme activities were not measured in this study and may provide a useful indicator of the carbohydrate metabolism of the cheetah.

5.3.6. Ethanolamine

Ethanolamine is an organic compound that is abundantly present in the cell membranes of mammalian cells primarily as a key component of the phospholipid, phosphatidylethanolamine (PE) (Zhou et al., 2017). Ethanolamine is an essential nutrient obtained from dietary lipids, the turnover of enterocytes and bacterial cells, and indirectly via the decarboxylation of phosphatidylserine (PS) to PE resulting in a base exchange between serine and ethanolamine. PE and phosphatidylcholine (PC) account for more than half of the total phospholipid content in eukaryotic cells and, as a result, play an important role in cellular structure and function (Zhou et al., 2017). PE is produced via the CDP-ethanolamine pathway in the endoplasmic reticulum in which ethanolamine is phosphorylated by ethanolamine kinase ultimately producing PE. Along with PE production, ethanolamine can also be catabolised by intestinal bacteria producing many products including acetaldehyde, ammonia,

90

acetyl CoA, ethanol and acetic acid – most of which (excluding acetyl CoA) are harmful to the intestine and microflora (Zhou et al., 2017). Therefore, ethanolamine plays a role in regulating lipid metabolism directly and indirectly though the intestine microflora, as well as increasing the concentration of short chain fatty acids (acetic and butyric acid) which are end products of bacterial fermentation.

In the current study, ethanolamine was identified in the serum of all cheetahs across the three diets and is seemingly higher in the control diet compared to the baseline, which is also reflected by its contribution to the control diet in the PLS-DA model. Captive cheetahs would generally consume a high protein and moderate fat diet with horse meat providing an estimated 6 to 12% lipid composition (Badiani et al., 1997; Bechert et al., 2002). In an early rat model, it was found that glycine is not directly converted to ethanolamine but is converted indirectly via serine as an intermediate (Greenberg and Harris, 1950). As the cheetahs would receive the same amount of fat throughout the study, differences would likely be linked to the higher collagen at baseline or glycine on the glycine diet. However, the concentration of ethanolamine was seemingly higher on the control diet – which may be due to the turnover of intestinal cells and bacteria. In domestic cats, 144 serum metabolites were identified that were related to lipid metabolism with decreasing concentrations of ethanolamine derivatives associated with early weight loss (Pallotto et al., 2021). Interestingly, glycine derivatives (such as sarcosine and N-acetylglycine) increased and decreased at various stages of weight loss, but was associated throughout the 16 week study. In the current study, cheetahs had significantly lower weight on the glycine diet which may account for the lower ethanolamine concentration and the effect that glycine has on adiposity, insulin sensitivity and βoxidation (El Hafidi et al., 2004). However, there was no significance and baseline appeared to have a lower concentration even though the cheetahs weighed more. The utilization of ethanolamine by gut bacteria has been shown to be beneficial for their survival in the host with this cycle leading to greater proliferation of epithelial cells ultimately producing more ethanolamine and its products (Zhou et al., 2017). This may account for the higher concentration on the control diet, whereby more bacteria are utilizing ethanolamine leading to increased proliferation and turnover of epithelial cells and bacteria. Although acetic acid and other phospholipid metabolites were not identified, the notably identification of ethanolamine indicates a link between lipid metabolism, the biosynthesis and degradation of ethanolamine by the gut microbiome and the importance of dietary collagen and glycine.

5.3.7. L-Methionine

Methionine was identified at low concentrations in 2 cheetahs at baseline, and 5 cheetahs on the control and glycine supplemented diet. The lower identification at baseline is likely driving its contribution. Methionine, the sulphur containing essential amino acid, has previously been identified in the serum of cheetahs at a concentration of $63 \pm 28 \mu mol/L$ (Tordiffe and Mienie, 2019).

Methionine must be obtained from dietary protein and is necessary for normal growth and development in mammals as it is the precursor for S-adenosylmethionine (SAM, which is utilised for the biosynthesis of the polyamines spermidine and spermine), homocysteine, succinyl-CoA and phosphatidylcholine (phospholipid) (Finkelstein, 1990). Homocysteine is an important substrate in the recycling of intracellular folates, the catabolism of choline and betaine; and the biosynthesis of cystathionine, cysteine and glutathione (Finkelstein, 1998). The conversion of available methionine to SAM via methionine adenosyltransferase is the major pathway for methionine metabolism and is produced at higher concentrations in the liver compared to other tissues (Finkelstein, 1990).

Cats have a high dietary demand for methionine and cysteine due to its gluconeogenic role, along with all the other functions (MacDonald et al., 1984). In a recent study investigating the metabolic differences between domestic cats at varying risks for diabetes, it was found that serum methionine was significantly higher in Maine Coon cats which have moderate risk for diabetes compared to the low risk Birman cat (Mu et al., 2021). Interestingly, increased dimethylglycine was also found in the Maine Coon, which is a precursor for glycine synthesis and also the product formed during the conversion of homocysteine to methionine. In domestic shorthair cats fed a high choline diet, the plasma methionine concentration was the only amino acid that was significantly higher – likely due to the increased betaine and re-methylation of homocysteine (Verbrugghe et al., 2021). Reduced plasma methionine and metabolites of the methionine cycle have also been shown to be associated with longevity when the metabolic profile of species with different lifespans were investigated (Mota-Martorell et al., 2021). These studies indicate that methionine concentration appears to change according to different breeds or species and conditions. The seemingly higher (or more identified) methionine concentration on the control and glycine diets compared to the baseline in the current study, may be due to the length of time on a diet and the dietary change, and not specifically the effect of glycine supplementation. However, the routes of increased methionine concentration may be different as the pathway of endogenous glycine biosynthesis from choline does lead to methionine formation. Alternatively, glycine supplementation may reduce pressure on glutathione production and allow the recycling of homocysteine to methionine. Methionine plays an extensive role in the body and concentrations may vary in the cheetah according to different dietary conditions, like in the domestic cat, but the direct link to glycine supplementation is unclear.

5.3.8. Propionic acid

Propionic acid, or propionate, was identified at low concentrations in the serum of 8 cheetahs at baseline, 6 cheetahs on the control diet and 3 cheetahs on the glycine supplemented diet. The low identification was likely due to the low relative concentrations of this metabolite in the serum. Despite this, propionic acid had a high VIP score and noticeably contributed towards the baseline diet. Propionic acid is formed by the fermentation of undigested food (mostly starch and fibre, but also proteins and long chained fatty acids) by the intestinal microbiota, and along with acetic acid

and butyric acid is grouped as a short chain fatty acid (SCFA) (AI-Lahham et al., 2010). These substrates are broken down and converted via several potential pathways to pyruvate which is ultimately converted to propionic acid and propionyl-CoA, where the latter enters the citric acid cycle as succinyl-CoA (AI-Lahham et al., 2010).

Cheetahs on a beef based diet have been shown to have higher faecal ammonia, indole, phenol and SCFA concentrations compared to other exotic captive felids (Vester et al., 2008). In a similar study comparing the differences between horse and beef based diets in captive cheetahs, the beef diet resulted in two to three times higher concentrations of all the aforementioned products (Vester et al., 2010). In both studies, acetic acid (~70%) was the SCFA with the highest concentration followed by propionic acid (~22%) and butyric acid (~8%). Additionally, on a rabbit carcass diet cheetahs produced lower concentrations of SCFA and putrefactive compounds, which the authors suggest is due to the beneficial effects of poorly digestible animal tissue such as hair and bones (Depauw et al., 2013). In the current study, propionic acid was identified more frequently in the baseline diet compared to the control diet with the glycine diet having the lowest identified relative concentration. These cheetahs were fed horse meat which, based on the study by Vester et al. (2010), would produce lower concentrations of SCFA compared to another meat diet - but interestingly the baseline diet which would likely contain remnants of a more collagen rich and optimum diet had the highest concentration. This supports what was proposed by Depauw et al. (2011) that the poorly digestible animal tissue is causing the beneficial SCFA effects and not necessarily any metabolic benefits that a historic collagen rich diet may provide. In a short report, it was found that patients with propionic acidaemia had normally lower plasma glycine (664 µmol/L) compared to times when they experienced severe episodes of hyperammonaemia and acidosis (1200 µmol/L). The authors suggested that this may be due to increased glycine conjugation of propionic acid (forming propionylglycine) during periods of acidosis (Al-Hassnan et al., 2003). Indeed, in the current study, glycine supplementation may lead to increased conjugation of propionic acid and other detrimental intestinal products which results in lower concentrations, therefore, improving gastrointestinal health. Acetic and butyric acid were not identified in the serum so it is difficult to draw conclusions relating to the SCFA, but glycine may be beneficial for those acids too.

5.4. Summary

In summary, 60 metabolites were identified in the serum of ten cheetahs at baseline, on a control and glycine supplemented diet. This was reduced to 7 metabolites of interest that contributed most notably to the metabolic differences, based on multivariate analyses, between cheetahs in the three groups. Glutamic acid, α -aminobutyric acid and methionine contributed primarily to the baseline diet and likely do not reflect specific changes to glycine supplementation. However, threonine,

glucose-6-phosphate, ethanolamine and propionic acid concentrations were influenced by glycine supplementation. These metabolites play various roles in energy pathways, glucose metabolism, lipid metabolism and intestinal microbiome fermentation. Specifically, the increased threonine and decreased propionic acid concentrations on the glycine supplemented diet reflect positive changes in the gastrointestinal tract.

CHAPTER SIX

6. GENERAL DISCUSSION AND CONCLUSION

Cheetahs are a unique and vulnerable felid species, currently only occupying 9% of their global historic range resulting in captivity being an important tool for their conservation (Durant et al., 2017). However, in captivity they are prone to an array of diseases that are not seen in their free ranging counterparts such as chronic lymphoplasmacytic gastritis, glomerulosclerosis and venoocclusive disease (Munson, 1993; Munson et al., 1999). Various factors of captivity, namely stress, lack of physical activity and diet have been suggested as possible causes of the increased disease prevalence. To date, no studies have investigated the link between physical activity and disease and the evidence regarding the effects of stress is somewhat inconsistent. Therefore, diet has become an increasingly important avenue of investigation over the last 20 years. This is particularly important when considering the contrast of feeding in the wild, in which cheetahs would capture small to medium prey and consume entire carcasses, compared to feeding in captivity, in which cheetahs would generally consume raw muscle meat (with limited carcass components) and vitamin and mineral supplementation (Hayward et al., 2006; Whitehouse-Tedd et al., 2015). Incomplete digestion of this high meat protein diet may lead to increased bacterial fermentation in the intestine elevating the production of putrefactive compounds such as SCFA, ammonia, phenolic and indolic compounds (Depauw et al., 2013). Carcass components such as hair, skin and bone have been shown to be poorly fermentable substrates and, therefore, may act as a bulking agent filling up the intestines and forming a barrier between the bacteria and substrates (Depauw et al., 2013, 2012a). Additionally, captive cheetahs have been shown to produce more detrimental compounds compared to other captive felids (Vester et al., 2008). This indicates a link between the captive diet and the unique metabolism and physiology of the cheetah.

The major difference between the captive and wild diet lies in the presence of carcass components such as cartilage, bones, ligaments, tendons and skin – all of which are rich in collagen. Glycine is the most abundant amino acid present in collagen, which may indicate a deficiency of this amino acid in a captive diet (Wang et al., 2013). Indeed, a third of the most abundant organic acids excreted in the urine of cheetahs are associated with glycine conjugates which indicates a substantial need for dietary glycine (Tordiffe et al., 2017). Therefore, the role of glycine in collagen biosynthesis and in the conjugation of organic acids which allows the excretion of detrimental compounds, is particularly important when the gut microbiome and gastrointestinal health of captive cheetahs is considered. Glycine has many other physiological roles in the body such as the biosynthesis of creatine, glutathione, purines, haem and as a neurotransmitter which may be beneficial for captive cheetahs as has been shown in rodents, pigs and humans (Imenshahidi and Hossenzadeh, 2022; McCarty et al., 2018; Wang et al., 2013).

Therefore the purpose of this study was to investigate the metabolic effects of glycine supplementation in captive cheetahs fed a raw muscle meat diet. A metabolomics approach using urine and serum was utilised in order to obtain a holistic idea of the many metabolites and pathways involved. This was used alongside body measurements, haematology and serum biochemistry.

An overview of the important metabolites identified in the urine and serum, their potential pathways and the link to glycine are displayed in Figure 8. Although only 10 urine and 7 serum metabolites were discussed previously, Figure 8 shows the 31 important metabolites identified in the urine and 18 metabolites identified in the serum (the list of metabolites are included in Supplementary Table S4 and S6). Due to the large number of metabolites identified in the urine, only those that were identified with both univariate and multivariate analyses were included in this figure – this approach reduced 65 important metabolites to 31 metabolites. For serum, the metabolites identified with more than one analyses were included in the overview figure. Only maltose, valine and taurine overlapped in the urine and serum. This figure shows changes in metabolite concentration in relation to the control diet (not necessarily just univariate significant differences) and, therefore, highlights three potential influences. Firstly, the effect of a more collagen rich diet represented by the baseline diet (red box). It is clear from the previous chapters that in several metabolites the baseline diet is eliciting a response beyond that of glycine supplementation. This may be due to the historic or remnant effect of a collagen diet containing high levels of glycine or potentially due to other aspects of a carcass diet such as the indigestible components. Secondly, the effect of glycine supplementation alone (green box) and, thirdly, the matched effect of a collagen diet or a glycine supplemented diet (yellow box).

In previous chapters, the metabolites and the effect of glycine was described individually and not as a collective pathway. However, based on Figure 8 it is clear that there is notable overlap and connections between the identified metabolites with four specific pathways highlighted: (1) methionine associated pathways, (2) threonine associated pathways, (3) carbohydrate and fat metabolism, and (4) glutamic acid associated pathways. Each of these are linked to specific physiological functions of glycine namely collagen biosynthesis, glutathione production, creatine and energy production, DNA and RNA biosynthesis, and digestion and conjugation. These functions are highlighted by different colour areas in the figure. The discussion below will focus on each of these pathways. For ease of reference, it would be beneficial to move outwards from methionine and threonine when looking at Figure 8 as they are the starting points of the discussion. However, the figure is interconnected and can be read from any starting point.



fructose-1-phosphate; MMA, methylmalonic AABA, α-aminobutyric DMSO₂, dimethyl 1,3-diaminopropane; SAM, S-adenosyl methionine; THF, tetrahydrofolate; TMP, thymidine monophosphate; UMP, uridine monophosphate; CMP, cytidine

6.1. Methionine associated pathways

Methionine is an essential amino acid with a high dietary demand in cats, although cats are less tolerant of excess methionine compared to other amino acids indicating a high utilisation in the body (MacDonald et al., 1984). In the study by Tordiffe and Mienie (2019), methionine was identified at low concentrations relative to other amino acids in the urine and serum of cheetahs. This was also found in the current study with the baseline diet having a lower concentration in both the serum and urine compared to the control and glycine diet, although the concentration was low across all diets. Despite these low concentrations, methionine is clearly a hub of metabolism with many important metabolites identified in this study forming downstream substrates or products of methionine pathways. In terms of the gut microbiome, the lower methionine concentration at baseline led to lower production and excretion of DMSO₂ produced from methionine degradation (He and Slupsky, 2014). Similarly, TMA (and subsequently DMA) concentrations were also lower, likely due to the lower concentration of its precursor's choline, dimethylglycine and carnitine. Interestingly, glycine supplementation also led to lower concentrations of both DMSO₂ and TMA despite methionine and choline only being lower at baseline. This may indicate that glycine supplementation is affecting the production of DMSO₂ and TMA through the reduction of other sources (such as carnitine and dimethylglycine). Additionally, glycine in itself may be acting as a substrate leading to the reduced utilisation of other amino acids and producing different products, or that the bacterial population of a collagen and glycine rich diet is similar allowing the same bacteria to survive and produce similar products (Dai et al., 2011). In a recent study by Baptista et al. (2020), DMSO₂ and TMA had a strong association with bone turnover and volume, and although the mechanisms are not understood, it supports what was found in the current study with both metabolites being influenced by dietary collagen and glycine, which would also have a strong link to bone. Overall, glycine appears to be providing the same gut bacterial benefits as a more collagen rich diet.

The conversion of methionine to SAM is the major methionine catabolic pathway (Finkelstein, 1990). SAM transfers a propylamine moiety to putrescine or spermidine to yield spermidine or spermine which are important polyamines. While the exact route of 1,3-DAP biosynthesis is unclear, it is a close homologue of putrescine and, therefore, may also be linked to SAM along with ornithine (Moinard et al., 2005). Polyamines are involved in cell proliferation, signalling and overall immune response, and although no other polyamine was significantly identified in the current study, the higher concentration of 1,3-DAP on the baseline diet alone indicates a collagen effect (or another factor of the pre-study diet such as other poorly digestible components) that is not necessary replicated with glycine.

The subsequent conversion of SAM to homocysteine allows the corresponding conversion of guanidinoacetic acid to creatine through the donation of a methyl group (Wyss and Kaddurah-Daouk, 2000). Glycine is directly involved in this process as the transfer of the amidino group of

arginine to glycine yields ornithine and guanidinoacetic acid which represents the first step of creatine biosynthesis (Wyss and Kaddurah-Daouk, 2000). The reversible conversion of creatine to phosphocreatine via creatine kinase is crucial for rapid ATP production, both of which can spontaneously be degraded to creatinine which is excreted from the body. Creatine and creatinine can also be converted by gut bacteria to methylguanidine which is a uremic toxin, and along with other guanidino compounds (such as creatinine and creatine) could have neurotoxic effects if allowed to accumulate in the body (Falconi et al., 2021; Wyss and Kaddurah-Daouk, 2000). Similarly, 1-methylhydantoin can also be produced from the degradation of creatinine by gut bacteria and is either taken up by the body or degraded further by bacteria producing sarcosine which can lead to glycine biosynthesis (Wyss and Kaddurah-Daouk, 2000). Although quanidinoacetic acid, creatine and creatinine were identified in the current study, the concentrations were similar across diets. However, methylguanidine and 1-methylhydantoin were both significantly lower on the baseline and glycine diets compared to the control diet. This in itself may have positive health implications for the cheetah, but it may also indicate that more creatine is being produced from the collagen and glycine rich diets allowing the interconversion to phosphocreatine thereby producing more ATP. In the wild, cheetahs require rapid energy production in order to chase and capture prey, and this finding indicates that a collagen or glycine rich diet (which would be reflected in their wild diet) may provide them with this crucial high energy phosphate to replenish ATP and also protect them from detrimental metabolites formed during the bacterial degradation of creatine and creatinine. Despite the lower physical activity experienced in captivity, these effects would undoubtedly still be beneficial for other metabolic processes.

Homocysteine, which is a key substrate produced from SAM, is involved in several important biological processes (Finkelstein, 1998). Firstly, homocysteine allows the re-synthesis of methionine through the utilisation of betaine as a methyl donor forming dimethylglycine. This is the only route for the degradation of betaine which is an intermediate in choline catabolism. Dimethylglycine ultimately forms glycine and accounts for approximately 1.4 mmol/day glycine in humans (Meléndez-Hevia et al., 2009). The lower concentrations of dimethylglycine on the baseline and glycine diet indicates that this route is not utilised to the same extent when exogenous glycine is provided or when collagen concentrations are high. Additionally, choline concentrations were lower on the baseline diet which would indicate decreased catabolism to betaine, however on the glycine diet choline may be re-directed towards phosphocholine biosynthesis. The second important homocysteine methylation reaction, which also converts to methionine, allows the recycling of 5-methyltetrahydrofolate (methyl-THF) to THF which constitutes the folate cycle (Finkelstein, 1998). Crucially, THF to methylene-THF via glycine hydroxymethyltransferase allows the biosynthesis of glycine from serine - this is the major pathway for endogenous glycine biosynthesis in mammals accounting for 34 mmol/day glycine in humans (Meléndez-Hevia et al., 2009). Therefore, these two pathways highlight the close connection between glycine biosynthesis

99

and the methionine cycle. Lastly, homocysteine can be irreversibly converted to cysteine and finally taurine. The decreased concentrations of taurine and cystine (oxidised cysteine) on the baseline and baseline and glycine diet, respectively, indicates lower flux through this pathway when exogenous glycine is provided. This is potentially due to reduced homocysteine degradation, due to reduced flux though the methionine cycle as the need for endogenous glycine is reduced, or potentially homocysteine redirection towards SAM production or the folate cycle.

Phosphocholine, produced from choline, was lower in the baseline and glycine diets which would indicate a decreased production of PC. Similarly, ethanolamine was an important metabolite highlighted in the serum of the baseline and glycine diets with decreased concentrations leading to decreased PE production as well as PC production through the CDP-ethanolamine pathway (Zhou et al., 2017). Together, it appears that cheetahs on the collagen and glycine rich diets have a lower source of phospholipids. This may also be due to the close link between lipid metabolism and the gut microbiome, in which the turnover of enterocytes and bacterial cells is a rich lipid source that can be converted to ethanolamine and PE (Zhou et al., 2017). The utilisation of ethanolamine by gut bacteria allows their survival and growth, therefore a lower concentration may indicate reduced bacterial colony growth and, along with the potential reduced biosynthesis of acetic acid, may indicate improved gastrointestinal health. Interestingly, CMP, which can be produced from phosphocholine, had a higher urinary concentration in the glycine diets. This highlights the redirection from choline towards cytidine biosynthesis, a pyrimidine component of RNA. β-alanine can be produced via several pathways, including from gut bacteria, but Figure 8 is highlighting the potential production from cytidine to uracil, ultimately producing β-alanine (Tiedje et al., 2010). βalanine is a neuromodulator and can potentially be considered as a small molecule neurotransmitter with glycine and GABA binding sites. The largest concentration of β -alanine in the body is found as carnosine, formed from β-alanine and histidine, which can act as a neuroprotective agent and as an intramuscular pH buffer (Tiedje et al., 2010). The concentration of carnosine from biosynthesis is directly proportional to the concentration of β -alanine, therefore, the reduced concentration of both these metabolites on the baseline and glycine diets potentially indicates a decrease in the catabolism of pyrimidines. This is echoed by the higher concentrations of dihydrothymine, a derivative of the pyrimidine thymine, which was higher in the glycine diet and can be formed from the folate cycle using methylene-THF and, therefore, has a close relationship with glycine biosynthesis from serine as thymine biosynthesis leads to equimolar glycine production (Meléndez-Hevia et al., 2009). The endogenous and exogenous source of glycine may be leading to this potential increase in thymine (and subsequently dihydrothymine biosynthesis). Although glycine is known to play a direct role in purine biosynthesis, this data also suggests a close relationship with pyrimidine biosynthesis as downstream products of glycine biosynthesis. Pyrimidines and purines form the base of DNA and RNA and, therefore, play an essential role in the body with defects in these processes being fatal for mammals (Löffler et al., 2005). In this study, the relative concentrations of the identified pyrimidine metabolites were low and changes are unlikely to indicate any fatal effects on cellular processes, but it does indicate that the glycine component of a collagen diet may be the major contributor towards the biosynthesis of these crucial compounds in cheetahs.

6.2. Threonine associated pathways

Threonine is an essential amino acid in cats and its degradation is a minor source of glycine in humans (Meléndez-Hevia et al., 2009). The increased concentration of threonine on the glycine diet is likely due to the exogenous glycine reducing the amount of endogenous glycine required by threonine degradation, allowing the redirection of threonine elsewhere. This could lead to increased mucosal protein synthesis which accounts for approximately 70% of threonine utilisation (which is not depicted in Figure 8) and would have intestinal benefits (Tang et al., 2021). It may also lead to increased flux towards α -ketobutyric acid and propionyl-CoA. Therefore, glycine supplementation may have a sparing effect on threonine, simulating dietary threonine supplementation which has been shown to be beneficial in rodents, broilers and pigs (Tang et al., 2021).

At baseline, there is an increased conversion of α-ketobutyric acid to AABA which is not replicated by the glycine diet. The function of AABA is still largely unknown. Therefore, this would lead to increased production of propionyl-CoA on the glycine diet. Unfortunately, the concentration of propionyl-CoA was not determined in this study, but the concentration of propionic acid may be used as an indicator. Propionic acid was lower on the glycine diet which may indicate increased propionyl-CoA. Similarly, the increased urinary MMA concentrations at baseline (with glycine being significantly lower compared to baseline) indicates that glycine may be influencing the flux towards propionyl-CoA and methylmalonyl-CoA towards succinyl-CoA, and not necessarily the remnants of a collagen rich diet. The increased concentration of pimelic acid on the baseline diet (and to a lesser extent on the glycine diet) may indicate a greater production of biotin which is a coenzyme for propionyl-CoA carboxylase that catalyses the conversion of propionyl-CoA to methylmalonyl-CoA (Zempleni et al., 2009). Alternatively, biotin could be used as a coenzyme in other pathways. There are a number of additional sources of propionyl-CoA including valine, isoleucine and odd chain fatty acids, all of which may be influencing the concentration. The reduced concentration of by-products of this pathway on the glycine diet may indicate greater flux towards the citric acid cycle. On the other hand, the baseline diet may also be leading to greater flux through these pathways but is producing increased by-products such as MMA and propionic acid (which may be detrimental). The effect of the baseline diet on propionyl-CoA is not likely to be caused by changes in threonine but potential other routes of propionyl-CoA production.

Propionic acid was the only SCFA identified in this study, but the identification of 3hydroxyphenylacetic acid (decreased) and isovaleric acid (increased) may be indicative of other SCFAs. Despite the structural similarity between acetic acid and 3-hydroxyphenylacetic acid, it may

not be linked to SCFA production. Nonetheless, it likely has a microbial origin as has been found with 4-hydroxyphenylacetic acid in cheetahs (Tordiffe et al., 2017). The reduced concentration of propionic acid on the glycine diet in the current study supports what was found by Depauw et al. (2013), but it also highlights that the effect of glycine specifically leading to a decreased concentration and not necessarily the long lasting remnants of a collagen diet. Alternatively, the actual carcass tissue components may elicit the same response, but that would be over a shorter period and cannot be concluded from this study. Isovaleric acid (a BCFA) has previously been identified in cheetahs by Depauw et al. (2013) in which the concentration was lower, along with all the other SCFAs and BCFAs, on a carcass diet compared to a minced beef diet. The increased concentration of isovaleric acid in the current study does not match what was found previously, although this BCFA contributes a very small percentage compared to the total SCFA (Depauw et al., 2013; Vester et al., 2010). Overall, considering that all cheetahs consumed a similar amount of muscle meat throughout the study, the reduced production of products of protein digestion and fermentation on the glycine diet compared to both the baseline and control diet, indicate a positive gastrointestinal effect of glycine. This may be due to an increase in threonine allowing intestinal mucosa production, thereby improving intestinal integrity, as well as improved glycine conjugation of organic acids and other detrimental products formed during digestion and fermentation of the high protein diet consumed by captive cheetahs.

6.3. Carbohydrate and fat metabolism

As obligate carnivores, the primary source of glucose in cheetahs is from proteins that link into gluconeogenic pathways (Schermerhorn, 2013). The identification of mannose, lactose, maltose and galactose as important metabolites in the serum is somewhat surprising considering the very low carbohydrate ingestion in cheetahs, especially with the sole consumption of horse muscle meat which would have no carbohydrates (Badiani et al., 1997). While no carbohydrate requirements have been shown for cats, they are generally able to efficiently utilise most common carbohydrates added to their diets (MacDonald et al., 1984). Interestingly, glucose and galactose covalently bind to collagen, and connective tissue also contains glycosaminoglycans - this may account for the slightly higher concentrations of these sugars on the baseline diet (Depauw et al., 2012a). All these identified sugars contain glucose monomers and, therefore, they may contribute towards the glucose concentration in the cheetah. Despite these potential contributions, the excretion of glucose was lower in the baseline and glycine diets. However, this may be due to increased conversion to G-6-P and flux through glycolysis. There is also a decreased production of glucuronic acid, which is a derivative of glucose, on the glycine and baseline diet which may also highlight flux through other pathways. Fructose concentration was higher in the glycine and baseline diet compared to the control diet, which may be contributing to the increased G-6-P concentration. This G-6-P could then lead to increased pyruvate and increased acetyl-CoA production to enter the citric acid cycle. The slightly higher lactic acid on the glycine diet may indicate increased flux through glycolysis (producing lactic acid from pyruvate). The concentrations of pyruvate remained low and fairly constant between diets, which also supports that any increase in glycolysis caused by glycine leading to pyruvate would be quickly converted to lactic acid or acetyl-CoA. Glycine supplementation in humans has been shown to reduce glucose concentrations and improve insulin sensitivity, and therefore has recently been suggested as a potential therapeutic treatment for obesity and diabetes (Imenshahidi and Hossenzadeh, 2022). This may be similar in cheetahs with lower glucose found on a collagen rich and glycine diet, despite the generally higher concentrations of potential glucose sources (such as mannose, lactose and maltose). This may be due to increased flux through glycolysis indicated by the increased G-6-P concentrations, although few intermediates of this pathway were identified. The similarities between the glycine and baseline diet also indicate that glycine is a major contributor and not simply the historic effects of collagen, although the presence of connective tissue which binds glucose and may act as a source of carbohydrates in cheetahs could lead to increased glucose when cheetahs are fed a carcass diet.

Malonic acid, along with its coenzyme malonyl-CoA, is involved with fatty acid biosynthesis. No fatty acids were identified in this study. Therefore, the lower concentration of malonic acid on the baseline and glycine diet may provide an indication of reduced fatty acid biosynthesis. Although cats and cheetahs have reduced capacity to synthesise certain fatty acids (such as linoleic acid), it is challenging to draw any conclusions around fatty acid biosynthesis and ingestion from this current study (Bauer, 1997). Carnitine transports fatty acids across the mitochondrial membrane to undergo β-oxidation and increased circulating and urine concentrations have been found in cats with feline hepatic lipidosis, but whether this shift is appropriate in magnitude for the metabolic disturbance, is unclear (Verbrugghe and Bakovic, 2013). In the current study, carnitine was lower on the baseline and glycine diet which may indicate reduced transport due to reduced fatty acids. The ketone body, acetoacetic acid, was also lower on the baseline and glycine diets. This may be a result of reduced utilisation of fatty acids and ketone bodies and reduced β-oxidation. Increasing acetoacetic acid concentrations was found in obese cats during a 16 week weight loss period, indicating a conversion to ketone bodies for energy rather than storing fats as triglycerides (Pallotto et al., 2021). The reduced concentrations of malonic acid, carnitine and acetoacetic acid on the baseline and glycine diets indicate a reduced synthesis, transport and oxidation of fatty acids. This may simply be indicative of a healthy body weight in contrast to what has been observed in obese cats and cats with hepatic lipidosis (Pallotto et al., 2021; Verbrugghe and Bakovic, 2013). In terms of body weight, the cheetahs in the current study had a significant decrease in weight from the baseline to glycine diet, which may be linked to fat loss, although both diets appeared to have a similar response to fat metabolism as highlighted by the abovementioned metabolites. Therefore, the increased weight loss may be linked to the role of glycine in the biosynthesis of bile salts as well as its role in the conjugation to bile acids improving excretion from the body – these are important for

the digestion and absorption of dietary fat (Meléndez-Hevia et al., 2009; Wang et al., 2013). Markers of bile acid metabolism, such as cholate, decreased with weight loss in cats which may be due to the role of bile acids in lipid digestion but also as signalling molecules that regulate metabolism and inflammation (Pallotto et al., 2021).

α-Ketoglutaric acid was the only significant metabolite identified that is directly involved in the citric acid cycle. The concentration was lower in the baseline and glycine diet, which is likely due to its involvement in glutamic acid biosynthesis (which will be discussed in the next section). Transaconitic acid, which is an intermediate linked to citric acid and isocitric acid, was also lower in the baseline and glycine diets. It is difficult to draw any conclusions regarding the citric acid cycle, but it appears that a historic collagen diet and a glycine diet may be causing similar effects. These effects may be shifting the entry points of compounds into the citric acid cycle, causing increased flux through succinyl-CoA and acetyl-CoA (from pyruvate) which was described earlier. This would then allow other metabolites, such as α -ketoglutaric acid and oxaloacetate, to be re-directed towards other functions such as glutamic acid production and aspartic acid production while still maintaining appropriate flux and producing ATP. Overall, it appears that a collagen and glycinerich diet elicited similar responses to carbohydrate and fat metabolism with reduced glucose production, increased flux through glycolysis, reduced fatty acid biosynthesis and β-oxidation. It also appears that energy production from the citric acid cycle may be shifted towards succinyl-CoA and acetyl-CoA (from pyruvate) to balance the redirection of oxaloacetate and α -ketoglutaric acid towards aspartic acid and glutamic acid production.

6.4. Glutamic acid associated pathways

The conversion of oxaloacetate, from the citric acid cycle, to aspartic acid allows the corresponding reversible conversion of α -ketoglutaric acid to glutamic acid (Brosnan and Brosnan, 2013). Similarly, glutamic acid can also be produced from proline, although this conversion is not reversible as cats lack pyrroline-5-carboxylate synthase (Wu et al., 2011). Glutamic acid had a high concentration on the baseline diet which corresponds with the lower concentration of proline and α -ketoglutaric acid, indicating a greater conversion to glutamic acid. Glycine, proline and hydroxyproline collectively make up the vitally important structural protein, collagen (Wang et al., 2013). Therefore, the lower urinary proline excretion on the baseline diet likely indicates a reduced turnover of collagen due to higher collagen ingestion and maintenance, while the opposite is seen on the control diet which is collagen and glycine deficient. The excretion of hydroxyproline was also slightly higher in the control group, although it was not identified as significant. Together, these indicate that the control group had a higher collagen turnover due to collagen deficiency, and has also been found in young cheetahs with potential nutritional deficiencies (Tordiffe and Mienie, 2019). Neither glycine nor any of its derivatives or conjugates were identified as important across the analyses. It is apparent that the baseline diet, which still contains remnants of a collagen rich

104

diet, has the greatest influence and that any impact that glycine supplementation may be having directly on collagen is minimal. It is interesting that despite the 3 week habituation period where no carcass components other than muscle meat were consumed by the cheetahs, there is still a clear metabolic influence on collagen that cannot be replicated by 4 weeks of glycine supplementation. This could be due to the collagen effect build up over a long period of time (cheetahs received mixed carcass components for at least 6 months prior to the study) or potentially the re-direction of glycine towards other physiological functions and not collagen biosynthesis. The latter could also emphasise that 30 g of glycine per kg meat (which would equal between 30 g to 60 g per serving) provided in this study is insufficient when the diet is completely collagen deficient. Additionally, this could also indicate that the baseline contains higher levels of glycine than would be expected.

Aspartic acid can also lead to lysine biosynthesis which degrades to the odorous amine cadaverine. Despite the lower lysine concentration at baseline, the baseline diet produced a higher amount of cadaverine. Cadaverine derivatives were previously identified at high concentration in urine of cheetahs, which the authors suggested may be due to the high muscle meat intake (Tordiffe et al., 2017). In the current study, the muscle meat content remained the same, but surprisingly the control and glycine diet yielded a lower cadaverine concentration. This may indicate that a collagen deficient diet (independent of glycine deficiency) redirects aspartic acid towards homoserine and methionine metabolism and away from lysine and asparagine. The decreased cadaverine concentration may be beneficial for the cheetah.

Glutamic acid can also be formed from glutamine, which occurs during the conversion of aspartic acid to asparagine. Asparagine excretion was significantly lower in the baseline and glycine groups, which may indicate that the high glutamic acid concentration on the baseline diet is primarily a result of proline and α -ketoglutaric acid. Although not highlighted in this study, urinary glutamine was slightly higher on the control and glycine diets which supports the lower glutamic acid found on those diets. Glutamic acid, glycine and cysteine are substrates required to produce the major antioxidant glutathione (Brosnan and Brosnan, 2013). Glycine and cysteine were similar across diets, while the higher glutamic acid concentration at baseline likely indicates a greater production of glutathione. This is reflected by the lower concentrations of pyroglutamic acid. This potential increased glutathione production would protect against oxidative damage and convey an improved immunity- this may explain the higher eosinophil and basophil WBC count in the control and glycine groups. Additionally, the lower pyroglutamic acid also indicates that there is adequate glycine as they have an inverse relationship (Jackson et al., 1987). This supports the above theory that there was indeed a glycine deficiency on the control diet that was not fully corrected by glycine supplementation and may lead to a decreased biosynthesis of glutathione, which would be detrimental. Overall, it appears that collagen (or historic metabolic effects of collagen) has a greater impact on glutamic acid related pathways than glycine alone and that when glycine is limited, it appears to be redirected to other pathways in cheetahs, such as pathways that promote glycine biosynthesis.

6.5. Study limitations and future directions

There were several limitations of this study, mostly linked to challenges around the study design and data analysis. Firstly, the two groups were not split equally and the length of the control diet was not the same between groups - this was purely due to logistical challenges. Secondly, the habituation period was implemented to standardise the diets for all cheetahs and it was anticipated that the 3 week habituation would also lead to a collagen and glycine deficiency. While the former occurred, the latter likely did not occur as it is apparent from the data throughout the study that the baseline diet had effects that, on many occasions, exceeded that of the glycine supplementation. This indicates that the cheetah's diet prior to the start of the study had much higher levels of collagen and glycine than was anticipated. Ideally, samples could have been collected before the habituation period which would have been a better representation of a collagen diet. As no intervention studies of this nature have been conducted in cheetahs, the length of the diet intervention (i.e. 4 weeks) and the amount of glycine (i.e. 30 g per kg meat) was deemed suitable based on studies in rodents, pigs and humans and fell within the time and funding constraints of the study. However, this may not have been long enough nor at a sufficient concentration to elicit a metabolic response similar to a collagen diet. Nevertheless, this is a realistic amount that could be implemented in captive facilities over an extended period of time.

Lastly, there were limitations associated with the NMR analysis and the ASICS software package. NMR inherently creates complex spectra in which it can be challenging to identify individual metabolites that are contributing to the spectra. The use of ASICS, as an automated software, was a simple and reliable approach to this challenge. However, ASICS has two notable limitations. Firstly, it contains a library of 190 metabolites and, therefore, only metabolites within this library could be identified in the current study. As a result, there may be a large number of metabolites that were not identified in the urine and serum of cheetahs. Secondly, the concentrations obtained were relative to the spectrum and not an absolute concentration. This has no direct implications to the data as all comparisons were made between study groups, but does limit comparison to the literature especially those using similar approaches in cats and cheetahs. These are not unique problems as all studies utilising NMR would face similar challenges.

Future studies should investigate the metabolic response of wild or captive cheetahs to a carcass diet which would provide an important nutritional reference. It may also emphasise aspects of metabolism that are influenced by the carcass diet and allow functional and feasible nutritional strategies to be implemented in captive facilities, especially when frequent carcass component diets are not possible. Future studies could also investigate the effect of glycine over an extended period

106

of time (such as six months) or the effects in a cheetah population with gastrointestinal diseases, as the current study was performed using a healthy and young cohort. Additionally, the present study has used an untargeted approach to highlight key areas and it may be beneficial to use a targeted approach and alternative separation and identification techniques (such as GC-MS) to enhance the data obtained. For example, targeted approaches could focus specifically on methionine, threonine, energy metabolism or organic compounds from bacterial fermentation as highlighted in the current study. Other molecular techniques such as such as fluorescence in situ hybridization (FISH), polymerase chain reaction (PCR) and sequencing (454-pyrosequencing and shotgun sequencing) may also be beneficial to further investigate the gut microbiome in captive cheetahs as has previously been conducted in domestic cats (Lyu et al., 2020). This approach would be particularly helpful when investigating their acute and long-term response to dietary changes (such as glycine supplementation) for three main reasons: (i) the microbiome rapidly responds to short-term macronutrient changes; (ii) long-term dietary habits are a major determinant in the bacterial composition of an individual's gut microbiota; and (iii) dietary changes have varying effects in different individuals (Lyu et al., 2020).

6.6. Conclusion

This was the first study to investigate the metabolic effects of glycine supplementation in captive cheetahs in an attempt to further understand nutrition and health in captivity. It appears that glycine supplementation had a direct effect on threonine and methionine related pathways, specifically that of threonine sparing, pyrimidine biosynthesis and decreasing the production of bacterial fermentation products which may prove to be extremely beneficial in improving gastrointestinal health. Glycine supplementation had a limited effect on collagen and glutathione production, indicating that when cheetahs experience glycine or collagen deficiency that glycine is redirected to other pathways potentially those that promote glycine biosynthesis or other metabolic pathways. Along with glycine, this study also highlighted the historic or remnant effect of a collagen rich diet which would be achieved through the consumption of bones, skin and connective tissue. This effect was similar to glycine supplementation for energy metabolism, specifically that of creatine production, carbohydrate and fat metabolism. Changes to these pathways, in particular glycolysis and succinyl-CoA biosynthesis, may indicate a shift that takes place when collagen or glycine is deficient in cheetahs. Additionally, these latter findings indicate that the effects of a collagen rich diet can be maintained in cheetahs for up to 3 weeks even with no additional sources of glycine and collagen, and that depletion only occurs between 3 to 7 weeks. The findings of this study indicate the diverse metabolic effects of glycine and collagen in captive cheetahs and future studies should use a targeted approach to further elaborate on these pathways. Glycine supplementation may be a useful tool to improve the health of captive cheetahs especially when dietary collagen is limited.

107

REFERENCES

- Al-Hassnan, Z.N., Boyadjiev, S.A., Praphanphoj, V., Hamosh, A., Braverman, N.E., Thomas, G.H., Geraghty, M.T., 2003. The relationship of plasma glutamine to ammonium and of glycine to acid– base balance in propionic acidaemia. J. Inherit. Metab. Dis. 26, 89–91.
- Al-Lahham, S.H., Peppelenbosch, M.P., Roelofsen, H., Vonk, R.J., Venema, K., 2010. Biological effects of propionic acid in humans; metabolism, potential applications and underlying mechanisms. Biochim. Biophys. Acta Mol. Cell Biol. Lipids 1801, 1175–1183. https://doi.org/https://doi.org/10.1016/j.bbalip.2010.07.007
- 3. Andres-Hernando, A., Johnson, R.J., Lanaspa, M.A., 2019. Endogenous fructose production: what do we know and how relevant is it? Curr. Opin. Clin. Nutr. Metab. Care 22, 289.
- Atila, A., Alay, H., Yaman, M.E., Akman, T.C., Cadirci, E., Bayrak, B., Celik, S., Atila, N.E., Yaganoglu, A.M., Kadioglu, Y., Halıcı, Z., Parlak, E., Bayraktutan, Z., 2021. The serum amino acid profile in COVID-19. Amino Acids 53, 1569–1588. https://doi.org/10.1007/s00726-021-03081-w
- Badenhorst, C.P.S., Erasmus, E., Van Der Sluis, R., Nortje, C., Van Dijk, A.A., 2014. A new perspective on the importance of glycine conjugation in the metabolism of aromatic acids. Drug Metab. Rev. 46, 343–361. https://doi.org/10.3109/03602532.2014.908903
- Badiani, A., Nanni, N., Gatta, P.P., Tolomelli, B., Manfredini, M., 1997. Nutrient Profile of Horsemeat. J. Food Compos. Anal. 10, 254–269. https://doi.org/10.1006/jfca.1997.0540
- Banik, C., Koziel, J.A., Li, J.Z., 2021. Simultaneous Chemical and Sensory Analysis of Domestic Cat Urine and Feces with Headspace Solid-Phase Microextraction. Separations 8, 15. https://doi.org/https://doi.org/10.3390/ separations8020015
- Baptista, A.L., Padilha, K., Malagrino, P.A., Venturini, G., Zeri, A.C.M., dos Reis, L.M., Martins, J.S., Jorgetti, V., Pereira, A.C., Titan, S.M., Moyses, R.M.A., 2020. Potential Biomarkers of the Turnover, Mineralization, and Volume Classification: Results Using NMR Metabolomics in Hemodialysis Patients. JBMR Plus 4, 1–9. https://doi.org/10.1002/jbm4.10372
- 9. Bauer, J.E., 1997. Fatty acid metabolism in domestic cats (Felis catus) and cheetahs (Acinonyx jubatas). Proc. Nutr. Soc. 56, 1013–1024.
- Bechert, U., Mortenson, J., Dierenfeld, E.S., Cheeke, P., Keller, M., Holick, M., Chen, T.C., Rogers, Q., 2002. Diet composition and blood values of captive cheetahs (Acinonyx jubatus) fed either supplemented meat or commercial food preparations. J. Zoo Wildl. Med. 33, 16–28. https://doi.org/10.1638/1042-7260(2002)033[0016:DCABVO]2.0.CO;2
- Beckonert, O., Keun, H.C., Ebbels, T.M.D., Bundy, J., Holmes, E., Lindon, J.C., Nicholson, J.K., 2007. Metabolic profiling, metabolomic and metabonomic procedures for NMR spectroscopy of urine, plasma, serum and tissue extracts. Nat. Protoc. 2, 2692–2703. https://doi.org/10.1038/nprot.2007.376
- 12. Bernardino, V.M.P., Albino, L.F.T., Rostagno, H.S., Oliveira, M.G.A., Mendes, F.Q., Pereira, C.M.C., Ferreira, I.M., Maia, R.C., 2011. Effect of different digestible threonine: Digestible lysine ratios, with
or without glycine supplementation, on the enzymatic activity in broiler chicks. Rev. Bras. Zootec. 40, 2732–2738. https://doi.org/10.1590/S1516-35982011001200017

- Bertschinger, H.J., Meltzer, D.G.A., Van Dyk, A., 2008. Captive Breeding of Cheetahs in South Africa
 30 Years of Data from the de Wildt Cheetah and Wildlife Centre. Reprod. Domest. Anim. 43, 66– 73. https://doi.org/10.1111/j.1439-0531.2008.01144.x
- Bissett, C., Bernard, R.T.F., 2007. Habitat selection and feeding ecology of the cheetah (Acinonyx jubatus) in thicket vegetation: Is the cheetah a savanna specialist? J. Zool. 271, 310–317. https://doi.org/10.1111/j.1469-7998.2006.00217.x
- Bond, J.C., Lindburg, D.G., 1990. Carcass feeding of captive cheetahs (Acinonyx jubatus): the effects of a naturalistic feeding program on oral health and psychological well-being. Appl. Anim. Behav. Sci. 26, 373–382. https://doi.org/10.1016/0168-1591(90)90036-D
- Bouatra, S., Aziat, F., Mandal, R., Guo, A.C., Wilson, M.R., Knox, C., Bjorndahl, T.C., Krishnamurthy, R., Saleem, F., Liu, P., Dame, Z.T., Poelzer, J., Huynh, J., Yallou, F.S., Psychogios, N., Dong, E., Bogumil, R., Roehring, C., Wishart, D.S., 2013. The Human Urine Metabolome. PLoS One 8. https://doi.org/10.1371/journal.pone.0073076
- Brosnan, J.T., Brosnan, M.E., 2013. Glutamate: A truly functional amino acid. Amino Acids 45, 413– 418. https://doi.org/10.1007/s00726-012-1280-4
- Buck, R.K., Tordiffe, A.S.W., Zeiler, G.E., 2022. Ketamine-medetomidine compared to tiletaminezolazepam-medetomidine for immobilisation of semi-captive cheetahs (Acinonyx jubatus). J. S. Afr. Vet. Assoc. 93, 1–7. https://doi.org/10.36303/JSAVA.2022.93.1.489
- 19. Burger, B. V., Visser, R., Moses, A., Le Roux, M., 2006. Elemental sulfur identified in urine of cheetah, Acinonyx jubatus. J. Chem. Ecol. 32, 1347–1352. https://doi.org/10.1007/s10886-006-9056-5
- Carlos, G., dos Santos, F.P., Fröehlich, P.E., 2020. Canine metabolomics advances. Metabolomics 16, 1–19. https://doi.org/10.1007/s11306-020-1638-7
- 21. Caro, T.M., Holt, M.E., FitzGibbon, C.D., Bush, M., Hawkey, C.M., Kock, R.A., 1987. Health of adult free-living cheetahs. J. Zool. 212, 573–584. https://doi.org/10.1111/j.1469-7998.1987.tb05956.x
- Chhibber-Goel, J., Gaur, A., Singhal, V., Parakh, N., Bhargava, B., Sharma, A., 2016. The complex metabolism of trimethylamine in humans: Endogenous and exogenous sources. Expert Rev. Mol. Med. 18, 1–11. https://doi.org/10.1017/erm.2016.6
- Chiarla, C., Giovannini, I., Siegel, J.H., 2011. Characterization of alpha-amino-n-butyric acid correlations in sepsis. Transl. Res. 158, 328–333. https://doi.org/10.1016/j.trsl.2011.06.005
- Chong, J., Wishart, D.S., Xia, J., 2019. Using MetaboAnalyst 4.0 for Comprehensive and Integrative Metabolomics Data Analysis. Curr. Protoc. Bioinforma. 68, 1–128. https://doi.org/10.1002/cpbi.86
- Costantini, D., Czirják, G., Melzheimer, J., Menges, V., Wachter, B., 2019. Sex and species differences of stress markers in sympatric cheetahs and leopards in Namibia. Comp. Biochem. Physiol. -Part A Mol. Integr. Physiol. 227, 8–13. https://doi.org/10.1016/j.cbpa.2018.09.002
- 26. Costantini, D., Wachter, B., Melzheimer, J., Czirják, G., 2017. Socioecological and environmental

predictors of physiological stress markers in a threatened feline species. Conserv. Physiol. 5, 1–14. https://doi.org/10.1093/conphys/cox069

- Crook, A.A., Powers, R., 2020. Quantitative NMR-Based Biomedical Metabolomics: Current Status and Applications. Molecules 25. https://doi.org/10.3390/molecules25215128
- 28. Dai, Z.-L., Wu, G., Zhu, W.-Y., 2011. Amino acid metabolism in intestinal bacteria: links between gut ecology and host health. Front Biosci 16, 1768–1786.
- Dawiskiba, T., Deja, S., Mulak, A., Ząbek, A., Jawień, E., Pawełka, D., Banasik, M., Mastalerz-migas, A., Balcerzak, W., Kaliszewski, K., Skóra, J., Barć, P., Korta, K., Pormańczuk, K., Szyber, P., Litarski, A., Młynarz, P., Dawiskiba, T., Skóra, J., Barć, P., Korta, K., 2014. Serum and urine metabolomic fingerprinting in diagnostics of inflammatory bowel diseases. World J Gastroenterol 20, 163–174. https://doi.org/10.3748/wjg.v20.i1.163
- Depauw, S., Bosch, G., Hesta, M., Whitehouse-Tedd, K., Hendriks, W.H., Kaandorp, J., Janssens, G.P.J., 2012a. Fermentation of animal components in strict carnivores: A comparative study with cheetah fecal inoculum. J. Anim. Sci. 90, 2540–2548. https://doi.org/10.2527/jas.2011-4377
- Depauw, S., Hesta, M., Whitehouse-Tedd, K., Stagegaard, J., Buyse, J., Janssens, G.P.J., 2012b. Blood values of adult captive cheetahs (Acinonyx jubatus) fed either supplemented beef or whole rabbit carcasses. Zoo Biol. 31, 629–641. https://doi.org/10.1002/zoo.20427
- Depauw, S., Hesta, M., Whitehouse-Tedd, K., Vanhaecke, L., Verbrugghe, A., Janssens, G.P.J., 2013. Animal fibre: The forgotten nutrient in strict carnivores? First insights in the cheetah. J. Anim. Physiol. Anim. Nutr. (Berl). 97, 146–154. https://doi.org/10.1111/j.1439-0396.2011.01252.x
- Dierenfeld, E.S., 1993. Nutrition of Captive Cheetahs: Food Composition and Blood Parameters. Zoo Biol. 12, 143–150. https://doi.org/10.1002/zoo.1430120113
- Dieterle, F., Ross, A., Schlotterbeck, G., Senn, H., 2006. Probabilistic quotient normalization as robust method to account for dilution of complex biological mixtures. Application in1H NMR metabonomics. Anal. Chem. 78, 4281–4290. https://doi.org/10.1021/ac051632c
- Durant, S., Mitchell, N., Ipavec, A., Groom, R., 2015. Acinonyx jubatus. The IUCN Red List of Threatened Species 2015: e.T219A50649567. Mammal Species. https://doi.org/10.1644/1545-1410(2005)771[0001:AJ]2.0.CO;2
- Durant, S.M., Mitchell, N., Groom, R., Pettorelli, N., Ipavec, A., Jacobson, A.P., Woodroffe, R., Böhm, M., Hunter, L.T.B., Becker, M.S., Broekhuis, F., Bashir, S., Andresen, L., Aschenborn, O., Beddiaf, M., Belbachir, F., Belbachir-Bazi, A., Berbash, A., De Matos Machado, I.B., Breitenmoser, C., Chege, M., Cilliers, D., Davies-Mostert, H., Dickman, A.J., Ezekiel, F., Farhadinia, M.S., Funston, P., Henschel, P., Horgan, J., De Iongh, H.H., Jowkar, H., Klein, R., Lindsey, P.A., Marker, L., Marnewick, K., Melzheimer, J., Merkle, J., M'Soka, J., Msuha, M., O'Neill, H., Parker, M., Purchase, G., Sahailou, S., Saidu, Y., Samna, A., Schmidt-Küntzel, A., Selebatso, E., Sogbohossou, E.A., Soultan, A., Stone, E., Der Meer, E. Van, Van Vuuren, R., Wykstra, M., Young-Overton, K., 2017. The global decline of cheetah Acinonyx jubatus and what it means for conservation. Proc. Natl. Acad. Sci. U. S. A. 114, 528–533. https://doi.org/10.1073/pnas.1611122114

- Eaton, K.A., Radin, M.J., Kramer, L., Wack, R., Sherding, R., Krakowka, S., Fox, J.G., Morgan, D.R., 1993. Epizootic Gastritis Associated with Gastric Spiral Bacilli in Cheetahs (Acinonyx jubatus). Vet. Pathol. 30, 55–63. https://doi.org/10.1177/030098589303000107
- El Hafidi, M., Pérez, I., Zamora, J., Soto, V., Carvajal-Sandoval, G., Baños, G., 2004. Glycine intake decreases plasma free fatty acids, adipose cell size, and blood pressure in sucrose-fed rats. Am. J. Physiol. - Regul. Integr. Comp. Physiol. 287, 1387–1393. https://doi.org/10.1152/ajpregu.00159.2004
- Elbourne, D.R., Altman, D.G., Higgins, J.P.T., Curtin, F., Worthington, H. V., Vail, A., 2002. Metaanalyses involving cross-over trials: Methodological issues. Int. J. Epidemiol. 31, 140–149. https://doi.org/10.1093/ije/31.1.140
- Falconi, C.A., Junho, C.V. da C., Fogaça-Ruiz, F., Vernier, I.C.S., da Cunha, R.S., Stinghen, A.E.M., Carneiro-Ramos, M.S., 2021. Uremic Toxins: An Alarming Danger Concerning the Cardiovascular System. Front. Physiol. 12, 1–20. https://doi.org/10.3389/fphys.2021.686249
- 41. Fernandez, N.J., Kidney, B.A., 2007. Alkaline phosphatase: Beyond the liver. Vet. Clin. Pathol. 36, 223–233. https://doi.org/10.1111/j.1939-165X.2007.tb00216.x
- 42. Finkelstein, J.D., 1998. The metabolism of homocysteine : pathways and regulation. Eur J Pediatr 157, 40–44.
- 43. Finkelstein, J.D., 1990. Methionine metabolism in mammals. J. Nutr. Biochem. 1, 228-237.
- 44. Forse, R.A., Shizgal, H.M., 1980. Serum Albumin and Nutritional Status. J. Parenter. Enter. Nutr. 4, 450–454. https://doi.org/10.1177/014860718000400503
- Fox, L., Haefele, H., Uelmen, J., Hoppes, S., Swenson, J., Tolbert, M.K., Suchodolski, J.S., Steiner, J.M., 2021. Biomarkers of Gastrointestinal Disease in Cheetahs (Acinonyx Jubatus). J. Zoo Wildl. Med. 52, 886–892. https://doi.org/10.1638/2021-0012
- 46. Gannon, M.C., Nuttall, J.A., Nuttall, F.Q., 2002. The metabolic response to ingested glycine. Am. J. Clin. Nutr. 76, 1302–1307. https://doi.org/10.1093/ajcn/76.6.1302
- 47. Garrett, R.H., Grisham, C.M., 2012. Biochemistry. Cengage Learning.
- Gillis-Germitsch, N., Vybiral, P.R., Codron, D., Clauss, M., Kotze, A., Mitchell, E.P., 2017. Intrinsic factors, adrenal gland morphology, and disease burden in captive cheetahs (Acinonyx jubatus) in South Africa. Zoo Biol. 36, 40–49. https://doi.org/10.1002/zoo.21341
- Gosselin, S.J., Setchell, K.D.R., Harrington, G.W., Welsh, M.B., Pylypiw, H., Kozeniauskas, R., Dollard, D., Tarr, M.J., Dresser, B.L., 1989. Nutritional considerations in the pathogenesis of hepatic veno-occlusive disease in captive cheetahs. Zoo Biol. 8, 339–347. https://doi.org/10.1002/zoo.1430080404
- 50. Greenberg, D.M., Harris, S.C., 1950. Lack of direct biological conversion of glycine to ethanolamine. Proc. Soc. Exp. Biol. Med. 75, 683–684.
- 51. Guasch-Ferre, M., Bhupathiraju, S.N., Hu, F.B., 2018. Use of metabolomics in improving assessment of dietary intake. Clin. Chem. 64, 82–98. https://doi.org/10.1373/clinchem.2017.272344
- 52. Hansen, J.C., Gilman, A.P., Odland, J.Ø., 2010. Is thermogenesis a significant causal factor in

preventing the "globesity" epidemic? Med. Hypotheses 75, 250–256. https://doi.org/10.1016/j.mehy.2010.02.033

- Hawkey, C.M., Hart, M.G., 1986. Haematological reference values for adult pumas, lions, tigers, leopards, jaguars and cheetahs. Res. Vet. Sci. 41, 268–269. https://doi.org/10.1016/s0034-5288(18)30611-8
- 54. Hayward, M.W., Hofmeyr, M., O'Brien, J., Kerley, G.I.H., 2006. Prey preferences of the cheetah (Acinonyx jubatus) (Felidae: Carnivora): morphological limitations or the need to capture rapidly consumable prey before kleptoparasites arrive? J. Zool. 270, 615–627. https://doi.org/10.1111/j.1469-7998.2006.00184.x
- 55. He, X., Slupsky, C.M., 2014. Metabolic fingerprint of dimethyl sulfone (DMSO2) in microbialmammalian co-metabolism. J. Proteome Res. 13, 5281–5292. https://doi.org/10.1021/pr500629t
- Hébert, J.R., Frongillo, E.A., Adams, S.A., Turner-McGrievy, G.M., Hurley, T.G., Miller, D.R., Ockene, I.S., 2016. Perspective: Randomized controlled trials are not a panacea for diet-related research. Adv. Nutr. 7, 423–432. https://doi.org/10.3945/an.115.011023
- Hendriks, M.M.W.B., Eeuwijk, F.A. va., Jellema, R.H., Westerhuis, J.A., Reijmers, T.H., Hoefsloot, H.C.J., Smilde, A.K., 2011. Data-processing strategies for metabolomics studies. Trends Anal. Chem. 30, 1685–1698. https://doi.org/10.1016/j.trac.2011.04.019
- Honneffer, J.B., Minamoto, Y., Suchodolski, J.S., Honneffer, J.B., Minamoto, Y., Suchodolski, J.S.,
 2014. Microbiota alterations in acute and chronic gastrointestinal inflammation of cats and dogs.
 World J Gastroenterol 20, 16489–16497. https://doi.org/10.3748/wjg.v20.i44.16489
- Hörster, F., Hoffmann, G.F., 2004. Pathophysiology, diagnosis, and treatment of methylmalonic aciduria - Recent advances and new challenges. Pediatr. Nephrol. 19, 1071–1074. https://doi.org/10.1007/s00467-004-1572-3
- Hudson-Lamb, G.C., Schoeman, J.P., Hooijberg, E.H., Heinrich, S.K., Tordiffe, A.S.W., 2016. Reference intervals for selected serum biochemistry analytes in cheetahs (Acinonyx jubatus). J. S. Afr. Vet. Assoc. 87, 1–6. https://doi.org/10.4102/jsava.v87i1.1316
- Hudson, P.E., Corr, S.A., Payne-Davis, R.C., Clancy, S.N., Lane, E., Wilson, A.M., 2011a. Functional anatomy of the cheetah (Acinonyx jubatus) hindlimb. J. Anat. 218, 363–374. https://doi.org/10.1111/j.1469-7580.2010.01310.x
- Hudson, P.E., Corr, S.A., Payne-Davis, R.C., Clancy, S.N., Lane, E., Wilson, A.M., 2011b. Functional anatomy of the cheetah (Acinonyx jubatus) forelimb. J. Anat. 218, 375–385. https://doi.org/10.1111/j.1469-7580.2011.01344.x
- 63. Hunter, J.S., Durant, S.M., Caro, T.M., 2007. To flee or not to flee: Predator avoidance by cheetahs at kills. Behav. Ecol. Sociobiol. 61, 1033–1042. https://doi.org/10.1007/s00265-006-0336-4
- Imenshahidi, M., Hossenzadeh, H., 2022. Effects of glycine on metabolic syndrome components: a review. J. Endocrinol. Invest. https://doi.org/10.1007/s40618-021-01720-3
- 65. Irino, Y., Toh, R., Nagao, M., Mori, T., Honjo, T., Shinohara, M., Tsuda, S., Nakajima, H., Satomi-

Kobayashi, S., Shinke, T., Tanaka, H., Ishida, T., Miyata, O., Hirata, K.I., 2016. 2-Aminobutyric acid modulates glutathione homeostasis in the myocardium. Sci. Rep. 6, 1–10. https://doi.org/10.1038/srep36749

- Ivanova, S.A., Boyko, A.S., Fedorenko, O.Y., Krotenko, N.M., Semke, A.V., Bokhan, N.A., 2014. Glutamate Concentration in the Serum of Patients with Schizophrenia. Procedia Chem. 10, 80–85. https://doi.org/10.1016/j.proche.2014.10.015
- Jackson, A.A., Badaloo, A. V., Forrester, T., Hibbert, J.M., Persaud, C., 1987. Urinary excretion of 5oxoproline (pyroglutamic aciduria) as an index of glycine insufficiency in normal man. Br. J. Nutr. 58, 207–214. https://doi.org/10.1079/bjn19870088
- Jain, A., Li, X.H., Chen, W.N., 2019. An untargeted fecal and urine metabolomics analysis of the interplay between the gut microbiome, diet and human metabolism in Indian and Chinese adults. Sci. Rep. 9, 1–13. https://doi.org/10.1038/s41598-019-45640-y
- 69. Jones, O.A.H., Cheung, V.L., 2007. An introduction to metabolomics and its potential application in veterinary science. Comp. Med. 57, 436–442.
- Kim, S., Kim, J., Yun, E.J., Kim, K.H., 2016. Food metabolomics: from farm to human. Curr. Opin. Biotechnol. 37, 16–23. https://doi.org/10.1016/j.copbio.2015.09.004
- 71. Kirberger, R.M., Tordiffe, A.S.W., 2016. Ultrasonographic adrenal gland findings in healthy semicaptive cheetahs (Acinonyx jubatus). Zoo Biol. 35, 260–268. https://doi.org/10.1002/zoo.21285
- 72. Kirsch, R., Frith, L., Black, E., Hoffenberg, R., 1968. Regulation of Albumin Synthesis and Catabolism by Alteration of Dietary Protein. Nature 217, 578–579. https://doi.org/10.1038/217578a0
- Koester, D.C., Wildt, D.E., Maly, M., Comizzoli, P., Crosier, A.E., 2017. Non-invasive identification of protein biomarkers for early pregnancy diagnosis in the cheetah (Acinonyx jubatus). PLoS One 12, 1–21. https://doi.org/10.1371/journal.pone.0188575
- Koopman, R., Caldow, M.K., Ham, D.J., Lynch, G.S., 2017. Glycine metabolism in skeletal muscle: Implications for metabolic homeostasis. Curr. Opin. Clin. Nutr. Metab. Care 20, 237–242. https://doi.org/10.1097/MCO.00000000000383
- 75. Kumar, A., Bachhawat, A.K., 2012. Pyroglutamic acid: Throwing light on a lightly studied metabolite. Curr. Sci. 102, 288–297. https://doi.org/https://www.jstor.org/stable/24083854
- Lane, E.P., Miller, S., Lobetti, R., Caldwell, P., Bertschinger, H.J., Burroughs, R., Kotze, A., van Dyk, A., 2012. Effect of diet on the incidence of and mortality owing to gastritis and renal disease in captive cheetahs (Acinonyx jubatus) in South Africa. Zoo Biol. 31, 669–682. https://doi.org/10.1002/zoo.20431
- 77. Lefort, G., Liaubet, L., Canlet, C., Tardivel, P., Père, M.C., Quesnel, H., Paris, A., Iannuccelli, N., Vialaneix, N., Servien, R., 2019. ASICS: an R package for a whole analysis workflow of 1D 1H NMR spectra. Bioinformatics 35, 4356–4363. https://doi.org/10.1093/bioinformatics/btz248
- Lenz, E.M., Wilson, I.D., 2007. Analytical strategies in metabonomics. J. Proteome Res. 6, 443–458. https://doi.org/10.1021/pr0605217

- 79. Li, P., Wu, G., 2018. Roles of dietary glycine, proline, and hydroxyproline in collagen synthesis and animal growth. Amino Acids 50, 29–38. https://doi.org/10.1007/s00726-017-2490-6
- Li, P., Yin, Y.L., Li, D., Kim, W.S., Wu, G., 2007. Amino acids and immune function. Br. J. Nutr. 98, 237–252. https://doi.org/10.1017/S000711450769936X
- Lin, S., Cronan, J.E., 2011. Molecular BioSystems Closing in on complete pathways of biotin biosynthesis 1811–1821. https://doi.org/10.1039/c1mb05022b
- Liu, R., Li, P., Bi, C.W., Ma, R., Yin, Y., Bi, K., Li, Q., 2017. Plasma N-acetylputrescine, cadaverine and 1,3-diaminopropane: Potential biomarkers of lung cancer used to evaluate the efficacy of anticancer drugs. Oncotarget 8, 88575–88585. https://doi.org/10.18632/oncotarget.19304
- Koffler, M., Fairbanks, L.D., Zameitat, E., Marinaki, A.M., Simmonds, H.A., 2005. Pyrimidine pathways in health and disease. Trends Mol. Med. 11, 430–437. https://doi.org/10.1016/j.molmed.2005.07.003
- Lyu, Y., Su, C., Verbrugghe, A., Van de Wiele, T., Martos Martinez-Caja, A., Hesta, M., 2020. Past, Present, and Future of Gastrointestinal Microbiota Research in Cats. Front. Microbiol. 11. https://doi.org/10.3389/fmicb.2020.01661
- 85. MacDonald, M.L., Rogers, Q.R., Morris, J.G., 1984. Nutrition of the domestic cat, a mammalian carnivore. Annu. Rev. Nutr. 4, 521–562. https://doi.org/10.1146/annurev.nu.04.070184.002513
- Marker-Kraus, L., 1997. History of the Cheetah: Acinonyx jubatus in zoos 1829–1994. Int. Zoo Yearb. 35, 27–43.
- 87. Marker, L., Johnston, B., 2020. International cheetah (Acinonyx jubatus) studbook. Cheetah Conserv. Fund, Namibia.
- Marker, L.L., Dickman, A.J., 2003. Morphology, physical condition, and growth of the cheetah (Acinonyx jubatus jubatus). J. Mammal. 84, 840–850. https://doi.org/10.1644/BRB-036
- Mastellar, S.L., Moffet, A., Harris, P.A., Urschel, K.L., 2016. Effects of threonine supplementation on whole-body protein synthesis and plasma metabolites in growing and mature horses. Vet. J. 207, 147–153. https://doi.org/10.1016/j.tvjl.2015.09.026
- McCarty, M.F., O'Keefe, J.H., DiNicolantonio, J.J., 2018. Dietary glycine is rate-limiting for glutathione synthesis and may have broad potential for health protection. Ochsner J. 18, 81–87. https://doi.org/10.1043/TOJ-17-0022
- Meléndez-Hevia, E., De Paz-Lugo, P., Cornish-Bowden, A., Cárdenas, M.L., 2009. A weak link in metabolism: the metabolic capacity for glycine biosynthesis does not satisfy the need for collagen synthesis. J. Biosci. https://doi.org/10.1007/s12038-009-0100-9
- 92. Mills, E.J., Chan, A.W., Wu, P., Vail, A., Guyatt, G.H., Altman, D.G., 2009. Design, analysis, and presentation of crossover trials. Trials 10, 1–6. https://doi.org/10.1186/1745-6215-10-27
- 93. Misra, B.B., 2021. New software tools , databases , and resources in metabolomics : updates from 2020. Metabolomics. https://doi.org/10.1007/s11306-021-01796-1
- 94. Mitchell, E.P., Prozesky, L., Lawrence, J., 2018. A new perspective on the pathogenesis of chronic

renal disease in captive cheetahs (Acinonyx jubatus). PLoS One 13, 1–16. https://doi.org/10.1371/journal.pone.0194114

- 95. Mitchell, S.C., Zhang, A.Q., Smith, R.L., 2008. Dimethylamine and diet. Food Chem. Toxicol. 46, 1734–1738. https://doi.org/10.1016/j.fct.2008.01.010
- 96. Moinard, C., Cynober, L., de Bandt, J.P., 2005. Polyamines: Metabolism and implications in human diseases. Clin. Nutr. 24, 184–197. https://doi.org/10.1016/j.clnu.2004.11.001
- 97. Mota-Martorell, N., Jové, M., Berdún, R., Pamplona, R., 2021. Plasma methionine metabolic profile is associated with longevity in mammals. Commun. Biol. 4, 1–11.
- 98. Mu, E., Moazzami, A., Hermansson, U., Hansson-hamlin, H., Pettersson, A., Anderson, F., Ha, J., 2021. Differences in metabolic profiles between the Burmese, the Maine coon and the Birman cat
 Three breeds with varying risk for diabetes mellitus 1–15. https://doi.org/10.1371/journal.pone.0249322
- 99. Munson, L., 1993. Diseases of captive cheetahs (Acinonyx jubatus): Results of the cheetah research council pathology survey, 1989–1992. Zoo Biol. 12, 105–124. https://doi.org/10.1002/zoo.1430120110
- 100. Munson, L., Nesbit, J.W., Meltzer, D.G., Colly, L.P., Bolton, L., Kriek, N.P., 1999. Diseases of captive cheetahs (Acinonyx jubatus jubatus) in South Africa: a 20-year retrospective survey. J. Zoo Wildl. Med. 30, 342–7.
- 101. Munson, L., Terio, K.A., Worley, M., Jago, M., Bagot-Smith, A., Marker, L., 2005. Extrinsic factors significantly affect patterns of disease in free-ranging and captive cheetah (Acinonyx jubatus) populations. J. Wildl. Dis. 41, 542–548. https://doi.org/10.7589/0090-3558-41.3.542
- 102. Muroya, S., Ueda, S., Komatsu, T., Miyakawa, T., Ertbjerg, P., 2020. Meatabolomics: Muscle and meat metabolomics in domestic animals. Metabolites 10. https://doi.org/10.3390/metabo10050188
- 103. Nuzzo, T., Mancini, A., Miroballo, M., Casamassa, A., Di Maio, A., Donati, G., Sansone, G., Gaetani, L., Paoletti, F.P., Isidori, A., Calabresi, P., Errico, F., Parnetti, L., Usiello, A., 2021. High performance liquid chromatography determination of I-glutamate, I-glutamine and glycine content in brain, cerebrospinal fluid and blood serum of patients affected by Alzheimer's disease. Amino Acids 53, 435–449. https://doi.org/10.1007/s00726-021-02943-7
- 104. O'Brien, S.J., Roelke, M.E., Marker, L., Newman, A., Winkler, C.A., Meltzer, D., Colly, L., Evermann, J.F., Bush, M., Wildt, D.E., 1985. Genetic basis for species vulnerability in the cheetah. Science (80-.). 227, 1428–1434. https://doi.org/10.1126/science.2983425
- 105. O'Brien, S.J., Wildt, D.E., Goldman, D., Merril, C.R., Bush, M., 1983. The cheetah is depauperate in genetic variation. Science (80-.). 221, 459–462.
- 106. Oliver, S.G., Winson, M.K., Kell, D.B., Baganz, F., 1998. Systematic functional analysis of the yeast genome. Trends Biotechnol. 16, 373–378.
- 107. Ospina-Rojas, I.C., Murakami, A.E., Oliveira, C.A.L., Guerra, A.F.Q.G., 2013. Supplemental glycine and threonine effects on performance, intestinal mucosa development, and nutrient utilization of

growing broiler chickens. Poult. Sci. 92, 2724-2731. https://doi.org/10.3382/ps.2013-03171

- 108. Pallotto, M.R., Oba, P.M., de Godoy, M.R.C., Pappan, K.L., Buff, P.R., Swanson, K.S., 2021. Effects of weight loss and moderate-protein, high-fiber diet consumption on the fasted serum metabolome of cats. Metabolites 11. https://doi.org/10.3390/metabo11050324
- 109. Pang, Z., Chong, J., Zhou, G., de Lima Morais, D.A., Chang, L., Barrette, M., Gauthier, C., Jacques, P.-É., Li, S., Xia, J., 2021. MetaboAnalyst 5.0: narrowing the gap between raw spectra and functional insights. Nucleic Acids Res. 1–9. https://doi.org/10.1093/nar/gkab382
- 110. Parthasarathy, A., Savka, M.A., Hudson, A.O., 2019. The Synthesis and Role of β -Alanine in Plants. Front. Plant Sci. 10. https://doi.org/10.3389/fpls.2019.00921
- 111. Persaud, C., Forrester, T., Jackson, A.A., 1996. Urinary excretion of 5-L-oxoproline (pyroglutamic acid) is increased during recovery from severe childhood malnutrition and responds to supplemental glycine. J. Nutr. 126, 2823–2830.
- 112. Persson, L., Rosengren, E., 1984. The effect of 1,3-diaminopropane on the urinary excretion of spermidine in the mouse. Agents Actions 15, 235–237. https://doi.org/10.1007/BF01972354
- 113. Petzke, K.J., Albrecht, V., Przybilski, H., 1986. The influence of high glycine diets on the activity of glycine-catabolizing enzymes and on glycine catabolism in rats. J. Nutr. 116, 742–750. https://doi.org/10.1093/jn/116.5.742
- 114. Philip, P., Hyland, K., Chiles, J., McGavin, C., Yu, Y., Taylor, D., 2012. Partial Pyridoxine Responsiveness in PNPO Deficiency. JIMD Rep. 4, 113–116. https://doi.org/10.1007/8904
- 115. Psychogios, N., Hau, D.D., Peng, J., Guo, A.C., Mandal, R., Bouatra, S., Sinelnikov, I., Krishnamurthy, R., Eisner, R., Gautam, B., Young, N., Xia, J., Knox, C., Dong, E., Huang, P., Hollander, Z., Pedersen, T.L., Smith, S.R., Bamforth, F., Greiner, R., McManus, B., Newman, J.W., Goodfriend, T., Wishart, D.S., 2011. The human serum metabolome. PLoS One 6. https://doi.org/10.1371/journal.pone.0016957
- 116. Quirke, T., O' Riordan, R.M., 2011. The effect of different types of enrichment on the behaviour of cheetahs (Acinonyx jubatus) in captivity. Appl. Anim. Behav. Sci. 133, 87–94. https://doi.org/10.1016/j.applanim.2011.05.004
- 117. Quirke, T., O'Riordan, R., Davenport, J., 2013. A comparative study of the speeds attained by captive cheetahs during the enrichment practice of the "cheetah run." Zoo Biol. 32, 490–496. https://doi.org/10.1002/zoo.21082
- 118. Rajas, F., Gautier-Stein, A., Mithieux, G., 2019. Glucose-6 phosphate, A central hub for liver carbohydrate metabolism. Metabolites 9. https://doi.org/10.3390/metabo9120282
- 119. Razak, M.A., Begum, P.S., Viswanath, B., Rajagopal, S., 2017. Multifarious Beneficial Effect of Nonessential Amino Acid, Glycine: A Review. Oxid. Med. Cell. Longev. 2017. https://doi.org/10.1155/2017/1716701
- 120. Ryan, D., Robards, K., Prenzler, P.D., Kendall, M., 2011. Recent and potential developments in the analysis of urine: a review. Anal. Chim. Acta 684, 17–29.

- 121. Saccenti, E., Hoefsloot, H.C.J., Smilde, A.K., Westerhuis, J.A., Hendriks, M.M.W.B., 2014. Reflections on univariate and multivariate analysis of metabolomics data. Metabolomics 10, 361– 374. https://doi.org/10.1007/s11306-013-0598-6
- 122. Schaeffer, D.J., Levengood, J.M., Adkesson, M.J., 2020. Effects of nest type and sex on blood saccharide profiles in Humboldt penguins (Spheniscus humboldti): Implications for habitat conservation. PLoS One 15, 1–14. https://doi.org/10.1371/journal.pone.0233101
- 123. Schermerhorn, T., 2013. Normal glucose metabolism in carnivores overlaps with diabetes pathology in non-carnivores. Front. Endocrinol. (Lausanne). 4, 1–15. https://doi.org/10.3389/fendo.2013.00188
- 124. Sekhar, R. V., Patel, S.G., Guthikonda, A.P., Reid, M., Balasubramanyam, A., Taffet, G.E., Jahoor, F., 2011. Deficient synthesis of glutathione underlies oxidative stress in aging and can be corrected by dietary cysteine and glycine supplementation. Am. J. Clin. Nutr. 94, 847–853. https://doi.org/10.3945/ajcn.110.003483
- Setchell, K.D.R., Gosselin, S.J., Welsh, M.B., Johnston, J.O., Balistreri, W.F., Kramer, L.W., Dresser, B.L., Tarr, M.J., 1987. Dietary estrogens-A probable cause of infertility and liver disease in captive cheetahs. Gastroenterology 93, 225–233. https://doi.org/10.1016/0016-5085(87)91006-7
- 126. Sharp, N.C.C., 1997. Timed running speed of a cheetah (Acinonyx jubatus). J. Zool. 241, 493–494. https://doi.org/10.1111/j.1469-7998.1997.tb04840.x
- 127. Sobczyńska-Malefora, A., Delvin, E., McCaddon, A., Ahmadi, K.R., Harrington, D.J., 2021. Vitamin B12 status in health and disease: a critical review. Diagnosis of deficiency and insufficiency–clinical and laboratory pitfalls. Crit. Rev. Clin. Lab. Sci. 58, 399–429. https://doi.org/10.1080/10408363.2021.1885339
- 128. Tang, Q., Tan, P., Ma, N., Ma, X., 2021. Physiological functions of threonine in animals: Beyond nutrition metabolism. Nutrients 13. https://doi.org/10.3390/nu13082592
- 129. Tardivel, P.J.C., Canlet, C., Lefort, G., Tremblay-Franco, M., Debrauwer, L., Concordet, D., Servien, R., 2017. ASICS: an automatic method for identification and quantification of metabolites in complex 1D 1H NMR spectra. Metabolomics 13, 1–9. https://doi.org/10.1007/s11306-017-1244-5
- 130. Terio, K.A., Marker, L., Munson, L., 2004. Evidence for chronic stress in captive but not free-ranging cheetahs (Acinonyx jubatus) based on adrenal morphology and function. J. Wildl. Dis. 40, 259–266. https://doi.org/10.7589/0090-3558-40.2.259
- 131. Tiedje, K.E., Stevens, K., Barnes, S., Weaver, D.F., 2010. β-Alanine as a small molecule neurotransmitter. Neurochem. Int. 57, 177–188. https://doi.org/10.1016/j.neuint.2010.06.001
- 132. Tordiffe, A.S.W., Mienie, L.J., 2019. Serum and urine amino acid profiles of captive cheetahs (Acinonyx jubatus). Comp. Clin. Path. 28, 287–296. https://doi.org/10.1007/s00580-018-2829-8
- 133. Tordiffe, A.S.W., van Reenen, M., Reyers, F., Mienie, L.J., 2017. Gas chromatography-mass spectrometry profiles of urinary organic acids in healthy captive cheetahs (Acinonyx jubatus). J. Chromatogr. B 1049–1050, 8–15. https://doi.org/10.1016/j.jchromb.2017.02.018
- 134. Tordiffe, A.S.W., Wachter, B., Heinrich, S.K., Reyers, F., Mienie, L.J., 2016. Comparative serum fatty

acid profiles of captive and free-ranging cheetahs (Acinonyx jubatus) in Namibia. PLoS One 11, 1– 14. https://doi.org/10.1371/journal.pone.0167608

- 135. Tran, H., McConville, M., Loukopoulos, P., 2020. Metabolomics in the study of spontaneous animal diseases. J. Vet. Diagnostic Investig. 32, 635–647. https://doi.org/10.1177/1040638720948505
- 136. Trimigno, A., Khakimov, B., Savorani, F., Poulsen, S.K., Astrup, A., Dragsted, L.O., Engelsen, S.B., 2020. Human urine 1H NMR metabolomics reveals alterations of the protein and carbohydrate metabolism when comparing habitual Average Danish diet vs. healthy New Nordic diet. Nutrition 79– 80, 110867. https://doi.org/10.1016/j.nut.2020.110867
- 137. Tsikas, D., 2020. Urinary dimethylamine (DMA) and its precursor asymmetric dimethylarginine (ADMA) in clinical medicine, in the context of nitric oxide (no) and beyond. J. Clin. Med. 9, 1–19. https://doi.org/10.3390/jcm9061843
- 138. Tvrzicka, E., Kremmyda, L.S., Stankova, B., Zak, A., 2011. Fatty acids as biocompounds: Their role in human metabolism, health and disease - a review. part 1: Classification, dietary sources and biological functions. Biomed. Pap. 155, 117–130. https://doi.org/10.5507/bp.2011.038
- 139. Verbrugghe, A., Bakovic, M., 2013. Peculiarities of one-carbon metabolism in the strict carnivorous cat and the role in feline hepatic lipidosis. Nutrients 5, 2811–2835. https://doi.org/10.3390/nu5072811
- 140. Verbrugghe, A., Rankovic, A., Armstrong, S., Santarossa, A., Kirby, G.M., Bakovic, M., 2021. Serum lipid, amino acid and acylcarnitine profiles of obese cats supplemented with dietary choline and fed to maintenance energy requirements. Animals 11. https://doi.org/10.3390/ani11082196
- 141. Vester, B.M., Beloshapka, A.N., Middelbos, I.S., Burke, S.L., Dikeman, C.L., Simmons, L.G., Swanson, K.S., 2010. Evaluation of nutrient digestibility and fecal characteristics of exotic felids fed horse-or beef-based diets: Use of the domestic cat as a model for exotic felids. Zoo Biol. 29, 432– 448. https://doi.org/10.1002/zoo.20275
- 142. Vester, B.M., Burke, S.L., Dikeman, C.L., Simmons, L.G., Swanson, K.S., 2008. Nutrient digestibility and fecal characteristics are different among captive exotic felids fed a beef-based raw diet. Zoo Biol. 27, 126–136. https://doi.org/10.1002/zoo.20172
- 143. Wang, W., Dai, Z., Wu, Z., Lin, G., Jia, S., Hu, S., Dahanayaka, S., Wu, G., 2014. Glycine is a nutritionally essential amino acid for maximal growth of milk-fed young pigs. Amino Acids 46, 2037– 2045. https://doi.org/10.1007/s00726-014-1758-3
- 144. Wang, W., Wu, Z., Dai, Z., Yang, Y., Wang, J., Wu, G., 2013. Glycine metabolism in animals and humans: Implications for nutrition and health. Amino Acids 45, 463–477. https://doi.org/10.1007/s00726-013-1493-1
- 145. Wang, Z., Bian, L., Mo, C., Shen, H., Zhao, L.J., Su, K.J., Kukula, M., Lee, J.T., Armstrong, D.W., Recker, R., Lappe, J., Bonewald, L.F., Deng, H.W., Brotto, M., 2020. Quantification of aminobutyric acids and their clinical applications as biomarkers for osteoporosis. Commun. Biol. 3. https://doi.org/10.1038/s42003-020-0766-y
- 146. Wasserstein, R.L., Schirm, A.L., Lazar, N.A., 2019. Moving to a World Beyond "p < 0.05." Am. Stat. 73, 1–19. https://doi.org/10.1080/00031305.2019.1583913

- 147. Weiner, H.S., Crosier, A.E., Keefer, C.L., 2019. Analysis of metabolic flux in felid spermatozoa using metabolomics and 13C-based fluxomics. Biol. Reprod. 100, 1261–1274. https://doi.org/10.1093/biolre/ioz010
- 148. Wells, A., Terio, K.A., Ziccardi, M.H., Munson, L., 2004. The stress response to environmental change in captive cheetahs (Acinonyx jubatus). J. Zoo Wildl. Med. 35, 8–14. https://doi.org/10.1638/02-084
- 149. Whitehouse-Tedd, K.M., Lefebvre, S.L., Janssens, G.P.J., 2015. Dietary factors associated with faecal consistency and other indicators of gastrointestinal health in the captive cheetah (Acinonyx jubatus). PLoS One 10, 1–20. https://doi.org/10.1371/journal.pone.0120903
- 150. Williams, T.M., Dobson, G.P., Mathieu-Costello, O., Morsbach, D., Worley, M.B., Phillips, J.A., 1997.
 Skeletal muscle histology and biochemistry of an elite sprinter, the African cheetah. J. Comp. Physiol. B 167, 527–535. https://doi.org/10.1007/s003600050105
- 151. Wishart, D.S., 2019. Metabolomics for Investigating Physiological and Pathophysiological Processes.Physiol. Rev. 99, 1819–1875. https://doi.org/10.1152/physrev.00035.2018
- 152. Woc Colburn, A.M., Sanchez, C.R., Citino, S., Crosier, A.E., Murray, S., Kaandorp, J., Kaandorp, C., Marker, L., 2018. Clinical Management of Captive Cheetahs, Cheetahs: Biology and Conservation. https://doi.org/10.1016/B978-0-12-804088-1.00024-1
- 153. Wu, G., 2009. Amino acids: metabolism, functions, and nutrition. Amino Acids 37, 1–17.
- 154. Wu, G., Bazer, F.W., Burghardt, R.C., Johnson, G.A., Kim, S.W., Knabe, D.A., Li, P., Li, X., McKnight, J.R., Satterfield, M.C., Spencer, T.E., 2011. Proline and hydroxyproline metabolism: Implications for animal and human nutrition. Amino Acids 40, 1053–1063. https://doi.org/10.1007/s00726-010-0715-z
- 155. Wyss, M., Kaddurah-Daouk, R., 2000. Creatine and creatinine metabolism. Physiol. Rev. 80, 1108– 1182.
- 156. Yu, A., Lau, A.Y., 2018. Glutamate and glycine binding to the NMDA receptor. Structure 26, 1035– 1043. https://doi.org/https://doi.org/10.1016/j.str.2018.05.004
- 157. Yu, Z., Kastenmüller, G., He, Y., Belcredi, P., Möller, G., Prehn, C., Mendes, J., Wahl, S., Roemisch-Margl, W., Ceglarek, U., Polonikov, A., Dahmen, N., Prokisch, H., Xie, L., Li, Y., Wichmann, H.E., Peters, A., Kronenberg, F., Suhre, K., Adamski, J., Illig, T., Wang-Sattler, R., 2011. Differences between human plasma and serum metabolite profiles. PLoS One 6, 1–6. https://doi.org/10.1371/journal.pone.0021230
- 158. Zeisel, S.H., DaCosta, K.A., Fox, J.G., 1985. Endogenous formation of dimethylamine. Biochem. J. 232, 403–408. https://doi.org/10.1042/bj2320403
- 159. Zempleni, J., Wijeratne, S.S.K., Hassan, Y.I., 2009. Biotin. Biofactors 35, 36–46. https://doi.org/10.1002/biof.8
- 160. Zhang, A., Sun, H., Wu, X., Wang, X., 2012. Urine metabolomics. Clin. Chim. Acta 414, 65–69. https://doi.org/http://dx.doi.org/10.1016/j.cca.2012.08.016

- 161. Zhong, Z., Wheeler, M.D., Li, X., Froh, M., Schemmer, P., Yin, M., Bunzendaul, H., Bradford, B., Lemasters, J.J., 2003. L-glycine: A novel antiinflammatory, immunomodulatory, and cytoprotective agent. Curr. Opin. Clin. Nutr. Metab. Care 6, 229–240. https://doi.org/10.1097/00075197-200303000-00013
- 162. Zhou, J., Xiong, X., Wang, K., Zou, L., Lv, D., Yin, Y., 2017. Ethanolamine Metabolism in the Mammalian Gastrointestinal Tract: Mechanisms, Patterns, and Importance. Curr. Mol. Med. 17, 1–8. https://doi.org/10.2174/1566524017666170331161715
- 163. Zoran, D.L., 2002. The carnivore connection to nutrition in cats. J. Am. Vet. Med. Assoc. 221, 1559– 1567.

APPENDIX 1

Supplementary Tables

Chemical Name	Chemical Name in PubChem	PubChem ID	CAS ID
1-Methyl-L-Histidine	1-Methyl-L-histidine	92105	332-80-9
1-Methylhydantoin	1-Methylhydantoin	69217	616-04-6
1,3-Diaminopropane	1,3-diaminopropane	428	109-76-2
2-AminoAdipicAcid	2-Aminohexanedioic acid	469	542-32-5
2-AminobutyricAcid	DL-2-Aminobutyric acid	6657	2835-81-6
2-Deoxyadenosine	2'-deoxyadenosine	13730	958-09-8
2-Deoxycytidine	2'-deoxycytidine	13711	951-77-9
2-Deoxyguanosine	2'-deoxyguanosine	187790	961-07-9
2-HydroxybutyricAcid	2-Hydroxybutyric acid	11266	600-15-7
2-HydroxyphenylAceticAcid	2-Hydroxyphenylacetic acid	11970	614-75-5
2-MethylglutaricAcid	2-Methylglutaric Acid	12046	1115-81-7
2-Oxobutyrate	2-oxobutanoic acid	58	600-18-0
2-Oxoglutarate	2-ketoglutaric acid	51	328-50-7
2-Oxoisovalerate	3-Methyl-2-oxobutyric acid	49	759-05-7
2-PicolinicAcid	Picolinic acid	1018	98-98-6
2-Propanol	Isopropanol	3776	67-63-0
3-Hydroxybutyrate	3-hydroxybutyric acid	441	300-85-6
3-HydroxyphenylAceticAcid	3-Hydroxyphenylacetic acid	12122	621-37-4
3-Methyl-L-Histidine	3-Methyl-L-histidine	64969	368-16-1
3-MethyladipicAcid	3-Methyladipic Acid	12292	03-01-3058
3-Methylxanthine	3-Methylxanthine	70639	1076-22-8
3-PhenylPropionicAcid	Hydrocinnamic acid	107	501-52-0
4-AminoHippuricAcid	4-Aminohippuric acid	2148	61-78-9
4-EthylPhenol	4-Ethylphenol	31242	123-07-9
4-HydroxyphenylAceticAcid	4-hydroxyphenylacetic acid	127	156-38-7
5-AminoValericAcid	5-Aminovaleric acid	138	660-88-8
7-Methylxanthine	7-Methylxanthine	68374	552-62-5
Acetaminophen	Acetaminophen	1983	103-90-2
AceticAcid	Acetic acid	176	64-19-7
Acetoacetate	Acetoacetic acid	96	541-50-4
Acetone	Acetone	180	67-64-1
Adenine	Adenine	190	73-24-5
Adenosine	Adenosine	60961	58-61-7
AdipicAcid	Adipic acid	196	124-04-9
ADP	Adenosine 5'-diphosphate	6022	58-64-0
Allantoin	Allantoin	204	97-59-6
alpha-HydroxyisobutyricAcid	2-Hydroxyisobutyric acid	11671	594-61-6
AMP	Adenosine 5'-monophosphate	6083	61-19-8
ArgininosuccinicAcid	Argininosuccinic acid	16950	2387-71-5
AscorbicAcid	Ascorbic acid	54670067	50-81-7
АТР	Adenosine triphosphate	5957	56-65-5
Azelaic Acid	Azelaic acid	2266	123-99-9

Table S1. ASICS reference library of 190 metabolites

BenzoicAcid	Benzoic acid	243	65-85-0
Beta-Alanine	Beta-alanine	239	107-95-9
beta-HydroxyisovalericAcid	beta-Hydroxyisovaleric acid	69362	625-08-1
Betaine	Betaine	247	107-43-7
Butyrate	Butyric acid	264	107-92-6
Cadaverine	1,5-Diaminopentane	273	462-94-2
CDP	Cytidine 5'-diphosphate	6132	63-38-7
CholineChloride	Choline chloride	6209	67-48-1
CitraconicAcid	Citraconic acid	643798	498-23-7
Citrate	Citric acid	311	77-92-9
СМР	5'-cytidylic acid / Cytidine monophosphate	6131	63-37-6
Creatine	Creatine	586	57-00-1
Creatinine	Creatinine	588	60-27-5
СТР	Cytidine triphosphate	6176	65-47-4
Cytosine	Cytosine	597	71-30-7
D-Fructose	D-Fructose	5984	57-48-7
D-Fucose	6-Deoxyhexopyranose	840	3615-37-0
D-Galactose	D-Galactose	6036	59-23-4
D-GluconicAcid	Gluconic acid	10690	526-95-4
D-Glucose	D-Glucose	5793	50-99-7
D-Glucose-6-Phosphate	Glucose 6-phosphate	5958	299-31-0
D-GlucuronicAcid	D-Glucuronic acid	65041	03-12-6556
D-Maltose	Maltose	6255	69-79-4
D-Mannose	D-Mannose	18950	530-26-7
D-Sorbitol	D-Sorbitol	5780	50-70-4
dAMP	2'-deoxyadenosine-5'-monophosphate	12599	653-63-4
DehydroAscorbicAcid	Dehydroascorbic acid	440667	490-83-5
Dihydrothymine	Dihydrothymine	93556	696-04-8
Dimethylamine	Dimethylamine	674	124-40-3
Dimethylglycine	N,N-dimethylglycine	673	1118-68-9
Dimethylsulfone	Dimethyl sulfone	6213	67-71-0
Ethanolamine	Ethanolamine	700	141-43-5
EthylmalonicAcid	Ethylmalonic acid	11756	601-75-2
Formate	Formic acid	284	64-18-6
FumaricAcid	Fumaric acid	444972	110-17-8
GABA	4-aminobutyric acid	119	56-12-2
Galactitol	Galactitol	11850	608-66-2
GDP	Guanosine 5'-diphosphate	8977	146-91-8
GlutaconicAcid	Glutaconic acid	5280498	628-48-8
GlutaricAcid	Glutaric acid	743	110-94-1
GlycericAcid	DL-Glyceric acid	752	473-81-4
Glycerol	Glycerol	753	56-81-5
Glycerophosphocholine	SN-glycero-3-phosphocholine	657272	28319-77-9
Glycogen	Glycogen	439177	9005-79-2
GlycolicAcid	Glycolic acid	757	79-14-1
GlyoxylicAcid	Glyoxylic acid	760	298-12-4
GMP	5'-Guanylic acid / guanosine monophosphate	6804	85-32-5

GTP	Guanosine triphosphate	6830	86-01-01
GuanidinoaceticAcid	Glycocyamine	763	352-97-6
HippuricAcid	Hippuric acid	464	495-69-2
HomovanillicAcid	Homovanillic acid	1738	306-08-1
Hypotaurine	Hypotaurine	107812	300-84-5
Hypoxanthine	Hypoxanthine	790	68-94-0
IMP	Inosinic acid	8582	131-99-7
Indoxylsulfate	Indoxyl sulfate	10258	487-94-5
Inosine	Inosine	6021	58-63-9
Isobutyrate	Isobutyric acid	6590	79-31-2
IsocitricAcid	Isocitric acid	1198	320-77-4
IsovalericAcid	Isovaleric acid	10430	503-74-2
KynurenicAcid	Kynurenic acid	3845	492-27-3
L-Alanine	L-alanine	5950	56-41-7
L-Anserine	Anserine	112072	584-85-0
L-Arabitol	L-arabitol	439255	7643-75-6
L-Arginine	L-arginine	6322	74-79-3
L-Asparagine	L-asparagine	6267	70-47-3
L-Aspartate	L-aspartic acid	5960	56-84-8
L-Carnitine	L-carnitine	10917	541-15-1
L-Carnosine	L-Carnosine	439224	305-84-0
L-Citrulline	L-citrulline	9750	372-75-8
L-Cysteine	L-cysteine	5862	52-90-4
L-Cystine	L-cystine	67678	56-89-3
L-GlutamicAcid	L-glutamic acid	33032	56-86-0
L-Glutamine	L-glutamine	5961	56-85-9
L-Glutathione-oxidized	Oxiglutatione	65359	27025-41-8
L-Glutathione-reduced	Glutathione	124886	70-18-8
L-Glycine	Glycine	750	56-40-6
L-Histidine	L-histidine	6274	71-00-1
L-Isoleucine	L-isoleucine	6306	73-32-5
L-Leucine	L-leucine	6106	61-90-5
L-Lysine	L-lysine	5962	56-87-1
L-Methionine	L-methionine	6137	63-68-3
L-Ornithine	L-ornithine	6262	70-26-8
L-Phenylalanine	L-phenylalanine	6140	63-91-2
L-Proline	L-proline	145742	147-85-3
L-Serine	L-serine	5951	56-45-1
L-Threonine	L-threonine	6288	72-19-5
L-Tryptophane	L-tryptophan	6305	73-22-3
L-Tyrosine	L-tyrosine	6057	60-18-4
L-Valine	L-valine	6287	72-18-4
Lactate	Lactic acid	612	50-21-5
Lactose	Lactose	440995	63-42-3
Levoglucosan	Levoglucosan	2724705	498-07-7
LevulinicAcid	Levulinic acid	11579	123-76-2
MalicAcid	Malic acid	525	6915-15-7
Malonate	Malonic acid	867	141-82-2

MandelicAcid	DI-Mandelic acid	1292	90-64-2
Methanol	Methanol	887	67-56-1
Methylamine	Methanamine	6329	74-89-5
Methylguanidine	Methylguanidine	10111	471-29-4
MethylmalonicAcid	Methylmalonic acid	487	516-05-2
Myo-Inositol	Scyllo-inositol	892	87-89-8
N-(2-Furoyl)Glycine	N-(2-Furoyl)glycine	21863	5657-19-2
N-Acetyl-L-AsparticAcid	N-Acetyl-L-aspartic acid	65065	997-55-7
N-Acetylglycine	N-Acetylglycine	10972	543-24-8
NAD	Nadide	5892	53-84-9
NADP	NADP	5886	53-59-8
NicotinicAcid	Nicotinic acid	938	59-67-6
NicotinuricAcid	Nicotinuric acid	68499	583-08-4
O-Acetyl-L-Carnitine	Acetyl-L-carnitine	7045767	3040-38-8
Oxypurinol	Oxypurinol	4644	2465-59-0
PantothenicAcid	Pantothenic acid	6613	79-83-4
Phenethylamine	Phenethylamine	1001	64-04-0
PhenylglyoxylicAcid	Benzoylformic acid	11915	611-73-4
Phosphocholine	Phosphocholine	1014	107-73-3
PimelicAcid	Pimelic acid	385	111-16-0
Propionate	Propionic acid	1032	79-09-4
PropyleneGlycol	Propylene glycol	1030	57-55-6
Pyrocatechol	Pyrocatechol	289	120-80-9
PyroglutamicAcid	L-Pyroglutamic acid	7405	98-79-3
Pyruvic-Acid	Pyruvic acid	1060	127-17-3
QuinolinicAcid	Quinolinic acid	1066	89-00-9
S-Acetamidomethylcysteine	S-(Acetamidomethyl)-L-cysteine	1590100	19647-70-2
SaccaricAcid	D-glucaric acid	33037	87-73-0
Sarcosine	Sarcosine	1088	107-97-1
SebacicAcid	Sebacic acid	5192	111-20-6
Spermidine	Spermidine	1102	124-20-9
Succinate	Succinic acid	1110	110-15-6
SyringicAcid	Syringic acid	10742	530-57-4
TartaricAcid	DL-Tartaric acid	875	133-37-9
Taurine	Taurine	1123	107-35-7
Threitol	D-Threitol	169019	2418-52-2
ThreonicAcid	Threonic acid	151152	14-12-3909
ТМАО	Trimethylamine oxide	1145	1184-78-7
trans-4-Hydroxy-L-Proline	trans-4-Hydroxy-L-proline	5810	51-35-4
Trans-AcotinicAcid	Trans-Aconitic acid	444212	4023-65-8
trans-FerulicAcid	Ferulic acid	445858	537-98-4
Trigonelline	Trigonelline	5570	535-83-1
Trimethylamine		1	75 50 0
1 mileary larmie	Trimethylamine	1146	75-50-3
UDP	Trimethylamine Uridine 5'-diphosphate	1146 6031	75-50-3 58-98-0
UDP UDPG	Trimethylamine Uridine 5'-diphosphate UDP-glucose	1146 6031 8629	75-50-3 58-98-0 133-89-1
UDP UDPG UMP	Trimethylamine Uridine 5'-diphosphate UDP-glucose Uridine 5'-monophosphate / Uridine monophosphate	1146 6031 8629 6030	75-50-3 58-98-0 133-89-1 58-97-9
UDPG UMP Uracil	Trimethylamine Uridine 5'-diphosphate UDP-glucose Uridine 5'-monophosphate / Uridine monophosphate Uracil	1146 6031 8629 6030 1174	75-50-3 58-98-0 133-89-1 58-97-9 66-22-8

UrocanicAcid	Urocanic acid	736715	104-98-3
UTP	Uridine 5'-triphosphate	6133	63-39-8
Valerate	Valeric acid	7991	109-52-4
VanillicAcid	Vanillic acid	8468	121-34-6
Xylitol	Xylitol	6912	87-99-0

Table S2. Relative concentration of the 112 metabolites identified in the urine of captive cheetahs at baseline, on a control and glycine supplemented diet (n=10).

Base	Baseline Diet					Control Diet				Glycine Diet			
Metabolite	n	Mean	SD	Metabolite	n	Mean	SD	Metabolite	n	Mean	SD		
Creatinine	10	64.11	47.37	Creatinine	10	72.76	37.56	Creatinine	10	56.90	46.17		
Glycolic Acid	4	21.53	29.15	Allantoin	10	13.73	2.91	Allantoin	10	15.19	3.77		
Allantoin	10	14.26	3.18	Malonate	10	13.23	4.91	L-Cysteine	10	10.07	1.62		
L-Cysteine	10	10.46	0.74	ТМАО	10	10.17	7.17	Glycolic Acid	7	9.59	6.24		
Malonate	10	7.84	4.74	L-Cysteine	10	9.86	0.88	Malonate	10	9.57	4.04		
L-Ornithine	10	6.12	1.33	Glycolic Acid	4	9.48	5.90	ТМАО	10	7.31	4.66		
Guanidinoacetic Acid	4	5.13	4.40	L-Carnitine	10	7.32	2.64	Guanidinoacetic Acid	7	6.81	5.23		
L-Proline	10	4.54	0.39	Guanidinoacetic Acid	5	6.72	5.47	L-Ornithine	10	6.38	1.15		
DHA	10	4.30	1.08	L-Ornithine	10	5.77	0.97	L-Proline	10	5.12	0.46		
L-Carnitine	10	3.87	2.27	L-Proline	10	5.50	0.70	L-Carnitine	10	4.96	2.22		
ΤΜΑΟ	10	3.87	3.59	Taurine	10	4.42	1.36	DHA	10	4.09	0.79		
1,3-DAP	10	3.35	0.33	DHA	9	3.95	0.91	Taurine	10	3.82	1.09		
L-Tryptophane	8	3.06	0.61	Choline Chloride	10	3.75	1.24	L-Glutamine	10	3.22	0.77		
L-Arginine	10	3.05	0.50	Phosphocholine	10	3.52	0.88	Choline Chloride	10	3.15	1.09		
Taurine	10	3.03	1.10	2-Oxoglutarate	10	3.41	0.35	1,3-DAP	10	3.05	0.31		
Choline Chloride	10	2.99	1.09	L-Glutamine	10	3.35	1.14	L-Arginine	10	2.98	0.34		
2-Oxoglutarate	10	2.70	0.42	D-Glucose	10	3.23	0.58	2-Oxoglutarate	10	2.93	0.29		
D-Maltose	10	2.61	0.41	D-Maltose	10	3.20	0.48	L-Tryptophane	10	2.80	0.61		
D-Glucose	10	2.59	0.35	L-Arginine	10	3.19	0.68	Phosphocholine	10	2.79	0.71		
L-Lysine	9	2.58	0.35	Dimethylglycine	10	3.18	2.14	D-Maltose	10	2.56	0.58		
Phosphocholine	10	2.56	1.02	Creatine	7	3.07	2.76	Threonic Acid	9	2.55	0.73		
L-Citrulline	10	2.55	0.22	1,3-DAP	10	2.96	0.34	L-Lysine	8	2.53	0.40		
L-Glutamine	10	2.53	0.89	Threonic Acid	8	2.73	1.01	D-Glucose	10	2.52	0.60		
D-Galactose	9	2.48	1.12	L-Tryptophane	9	2.72	0.55	β-Alanine	10	2.51	0.29		
β-Alanine	10	2.40	0.26	β-Alanine	10	2.72	0.19	L-Citrulline	10	2.45	0.27		
D-Gluconic Acid	8	2.38	0.78	D-Gluconic Acid	7	2.64	0.97	Glycerophospho- choline	10	2.06	0.49		
ThreonicAcid	9	1.99	0.68	1- Methylhydantoin	10	2.61	2.28	Phenethylamine	10	1.85	0.34		
Glycerophospho- choline	10	1.87	0.55	L-Citrulline	10	2.58	0.35	Glycogen	10	1.83	0.29		
Phenethylamine	10	1.84	0.45	L-Asparagine	9	2.57	1.08	Lactose	10	1.78	0.77		
Glycogen	10	1.76	0.20	L-Lysine	8	2.45	0.39	L-Cystine	10	1.72	0.30		
Lactose	10	1.71	0.53	Glycerophospho- choline	10	2.38	0.47	Isocitric Acid	9	1.65	0.82		

L-Cystine	10	1.64	0.30	Phenethylamine	10	2.22	0.87	Pyroglutamic Acid	10	1.55	0.34
L-Phenylalanine	10	1.62	0.63	L-Cystine	10	2.04	0.47	2-Aminoadipic Acid	10	1.48	0.21
Isocitric Acid	10	1.55	0.55	Trans-Acotinic Acid	10	2.00	0.90	D-Galactose	6	1.41	1.27
2-Aminoadipic Acid	10	1.51	0.33	Glyceric Acid	10	1.96	0.58	L-Phenylalanine	10	1.40	0.31
G-6-P	10	1.49	0.40	L-Phenylalanine	10	1.95	1.00	D-Fructose	10	1.33	0.54
2-Deoxycytidine	10	1.49	0.42	Glycogen	10	1.88	0.30	Trans-Acotinic Acid	10	1.31	0.51
Creatine	5	1.39	2.15	G-6-P	9	1.80	0.67	Glyceric Acid	8	1.28	0.82
L-Anserine	8	1.32	0.93	2-Aminoadipic Acid	10	1.80	0.52	Hypotaurine	10	1.27	0.28
Glyceric Acid	7	1.24	0.97	Isocitric Acid	9	1.73	0.84	D-Gluconic Acid	4	1.26	1.63
Pyroglutamic Acid	10	1.22	0.26	Pyroglutamic Acid	10	1.68	0.32	L-Histidine	10	1.25	0.61
Ascorbic Acid	9	1.19	0.60	L-Anserine	10	1.64	0.31	G-6-P	8	1.22	0.73
Hypotaurine	10	1.16	0.34	Lactose	10	1.56	0.46	Hippuric Acid	9	1.19	0.52
Trans-Acotinic Acid	10	1.12	0.47	Ascorbic Acid	10	1.54	0.61	Ascorbic Acid	10	1.18	0.46
HippuricAcid	10	1.12	0.31	L-Carnosine	10	1.51	0.30	L-Anserine	10	1.18	0.49
3- Phenylpropionic Acid	10	1.08	0.33	Hypotaurine	10	1.46	0.36	L-Carnosine	10	1.16	0.27
Glutaconic Acid	10	1.03	0.13	Ethanolamine	7	1.38	1.08	Glutaconic Acid	10	1.14	0.11
UMP	10	1.02	0.13	trans-4-Hydroxy- L-Proline	10	1.36	0.32	UMP	10	1.14	0.26
L-Asparagine	7	1.01	0.74	D-Mannose	9	1.34	0.60	D-Mannose	8	1.08	0.68
L-Histidine	10	1.01	0.28	Hippuric Acid	10	1.28	0.31	2-Deoxycytidine	8	1.06	0.64
L-Carnosine	9	0.98	0.43	D-Galactose	5	1.24	1.45	3- PhenylPropionic Acid	10	1.02	0.13
Adipic Acid	10	0.97	0.20	3- Phenylpropionic Acid	10	1.20	0.37	L-Asparagine	6	0.97	0.92
dAMP	10	0.96	0.44	2-Deoxycytidine	9	1.19	0.72	5-Aminovaleric Acid	10	0.97	0.19
D-Mannose	7	0.95	0.73	4-Hydroxyphenyl Acetic Acid	10	1.16	0.52	СМР	9	0.96	0.38
trans-4-Hydroxy- L-Proline	8	0.95	0.56	Dimethylamine	10	1.07	0.18	4-Hydroxyphenyl Acetic Acid	10	0.95	0.44
D-Fucose	10	0.94	0.17	dAMP	10	1.06	0.61	Ethanolamine	5	0.94	1.04
5-Aminovaleric Acid	10	0.94	0.22	UMP	10	1.05	0.21	trans-4-Hydroxy- L-Proline	8	0.94	0.58
D-Fructose	8	0.90	0.54	Glutaconic Acid	10	1.05	0.36	Dimethylamine	10	0.93	0.14
Dimethylglycine	9	0.85	0.97	Dimethyl sulphone	10	1.02	0.25	Adipic Acid	10	0.91	0.19
СМР	9	0.84	0.35	5-Aminovaleric Acid	10	0.95	0.16	dAMP	9	0.85	0.56
N-Acetylglycine	10	0.84	0.27	Adipic Acid	10	0.94	0.20	Dimethylglycine	6	0.84	1.24
IMP	9	0.81	0.48	L-Histidine	9	0.90	0.39	D-Fucose	10	0.77	0.19
4-Hydroxyphenyl Acetic Acid	9	0.79	0.39	L-Methionine	9	0.88	0.54	N-(2- Furoyl)Glycine	5	0.76	0.93
Spermidine	10	0.76	0.15	Methylguanidine	10	0.87	0.40	Dimethyl sulphone	10	0.75	0.12
L-Tyrosine	10	0.74	0.40	Indoxylsulfate	9	0.84	0.55	L-Methionine	8	0.73	0.48
Methylmalonic Acid	10	0.73	0.20	D-Fucose	10	0.83	0.21	L-Tyrosine	10	0.70	0.36
Dimethylamine	10	0.72	0.12	L-Tyrosine	9	0.77	0.42	L-Aspartate	6	0.70	0.66
GMP	8	0.70	0.55	N-Acetylglycine	8	0.73	0.43	1- Methylhydantoin	6	0.69	0.73
Pimelic Acid	10	0.66	0.10	Spermidine	9	0.73	0.32	L-Glutathione- oxidized	9	0.64	0.35

L-Methionine	7	0.60	0.47	Betaine	9	0.73	0.60	IMP	7	0.64	0.52
L-Aspartate	5	0.59	0.64	Acetaminophen	7	0.68	0.53	Betaine	9	0.64	0.29
Valerate	10	0.59	0.12	IMP	8	0.68	0.62	GTP	8	0.63	0.34
Levoglucosan	7	0.58	0.42	Levoglucosan	7	0.67	0.48	Pimelic Acid	10	0.63	0.09
L-Alanine	9	0.57	0.34	Acetoacetate	10	0.65	0.32	N-Acetylglycine	8	0.61	0.44
Dimethyl sulphone	10	0.56	0.10	GTP	8	0.64	0.37	Spermidine	8	0.61	0.37
L-Isoleucine	10	0.56	0.09	L-Glutathione-	8	0.64	0.48	L-Isoleucine	10	0.59	0.09
Isovaleric Acid	10	0.54	0.10	Methylmalonic	10	0.60	0.15	Valerate	10	0.55	0.09
Indoxylsulfate	7	0.53	0.40	3-Methylxanthine	9	0.60	0.74	Isovaleric Acid	10	0.54	0.11
3-Methylxanthine	7	0.53	0.90	L-Isoleucine	10	0.59	0.12	4-Aminohippuric Acid	8	0.54	0.66
L-Glutathione-	7	0.51	0.37	Pimelic Acid	10	0.58	0.09	Methylmalonic Acid	10	0.54	0.16
3- Hydroxybutyrate	10	0.50	0.04	L-Alanine	9	0.55	0.28	Dihydrothymine	9	0.50	0.35
4-Ethylphenol	8	0.48	0.27	L-Aspartate	4	0.54	0.71	3- Hydroxybutyrate	9	0.49	0.19
Myo-Inositol	7	0.47	0.37	Homovanillic Acid	7	0.54	0.50	Creatine	4	0.48	0.62
GTP	6	0.46	0.41	Valerate	10	0.53	0.07	L-Alanine	9	0.46	0.20
Betaine	8	0.45	0.28	СМР	6	0.52	0.47	Indoxylsulfate	6	0.45	0.42
1- Methylhydantoin	3	0.41	0.74	GMP	6	0.52	0.64	trans-Ferulic Acid	10	0.44	0.13
N-Acetyl-L- Aspartic Acid	10	0.41	0.06	Quinolinic Acid	5	0.52	0.57	UDPG	7	0.44	0.42
Quinolinic Acid	4	0.40	0.52	3- Hydroxybutyrate	10	0.52	0.12	L-Leucine	10	0.44	0.07
L-Leucine	10	0.40	0.04	Benzoic Acid	7	0.50	0.40	Acetoacetate	10	0.43	0.19
2-Oxobutyrate	10	0.40	0.05	trans-Ferulic Acid	10	0.50	0.12	N-Acetyl-L- Aspartic Acid	10	0.42	0.05
Acetaminophen	5	0.39	0.50	O-Acetyl-L- Carnitine	10	0.49	0.15	O-Acetyl-L-	10	0.38	0.12
Benzoic Acid	6	0.38	0.35	4-EthylPhenol	9	0.48	0.22	Methylguanidine	10	0.38	0.05
trans-Ferulic Acid	9	0.37	0.15	Isovaleric Acid	10	0.47	0.13	Quinolinic Acid	4	0.38	0.49
Methylguanidine	10	0.35	0.06	N-Acetyl-L- Aspartic Acid	10	0.47	0.13	4-EthylPhenol	8	0.37	0.23
O-Acetyl-L- Carnitine	10	0.35	0.18	3-Hydroxyphenyl Acetic Acid	9	0.42	0.17	Benzoic Acid	6	0.37	0.33
N-(2- Furovi)Glycine	2	0.35	0.89	2-Oxobutyrate	10	0.37	0.02	Levoglucosan	5	0.34	0.37
Acetoacetate	10	0.33	0.16	D-Fructose	3	0.36	0.62	Acetaminophen	4	0.34	0.46
4-Aminohippuric Acid	7	0.30	0.21	L-Leucine	10	0.36	0.06	2-Oxobutyrate	9	0.34	0.12
Pantothenic Acid	8	0.29	0.17	UDPG	5	0.32	0.41	Pyruvic Acid	10	0.31	0.06
UDPG	6	0.28	0.25	Pyruvic-Acid	10	0.32	0.05	3-Methylxanthine	9	0.31	0.18
Homovanillic Acid	5	0.27	0.29	Propylene Glycol	8	0.27	0.15	Homovanillic Acid	7	0.31	0.28
Ethanolamine	1	0.26	0.82	Trimethylamine	10	0.26	0.12	2-Hydroxyphenyl Acetic Acid	8	0.30	0.18
Propylene Glycol	7	0.26	0.21	4-Aminohippuric Acid	5	0.25	0.26	Propylene Glycol	9	0.28	0.11
Pyruvic-Acid	9	0.26	0.10	Myo-Inositol	3	0.25	0.41	Pantothenic Acid	7	0.24	0.18
3-Hydroxyphenyl	6	0.23	0.21	Dihydrothymine	6	0.23	0.20	GMP	4	0.24	0.34
2-Hydroxyphenyl	5	0.21	0.23	L-Valine	7	0.17	0.12	Myo-Inositol	3	0.23	0.37
Lactate	5	0.20	0.22	Lactate	4	0.15	0.20	3-Hydroxyphenyl	6	0.22	0.20
Sebacic Acid	5	0.19	0.21	Pantothenic Acid	4	0.15	0.19	Lactate	6	0.22	0.19
				I				I			-

Dihydrothymine	5	0.17	0.18	β-Hydroxy- isovaleric Acid	10	0.13	0.01	β-Hydroxy- isovaleric Acid	10	0.15	0.06
β-Hydroxy- isovaleric Acid	8	0.11	0.06	2-Hydroxyphenyl Acetic Acid	3	0.11	0.19	Trimethylamine	10	0.13	0.06
Trimethylamine	8	0.10	0.08	N-(2- Furoyl)Glycine	1	0.10	0.31	Sebacic Acid	2	0.09	0.21
L-Valine	2	0.05	0.11	Sebacic Acid	1	0.03	0.11	L-Valine	1	0.02	0.08

Data are expressed as mean and SD. n indicates the number of samples where the metabolite was detected. Relative concentration is an arbitrary unit. Refer to Table S1 for full names of metabolites.

Ta	able	S3.	The	(A) PCA	loadings	s, (B) P	LS-DA VI	P scores	and (C)	PLS-DA	diet r	egression	coefficients	for the
to	p 20) urin	ary i	metabolite	es in 10	captive	cheetahs	. All thes	se metak	oolites co	ntribut	ted toward	s Table S4.	

A. PCA Me	etabolite I	oadings		B. PLS-DA Metabolite VIP Scores					
PC 1		PC 2		C 1		C 2			
Pantothenic acid	-0.173	UDPG	-0.220	Dihydrothymine	2.632	Dihydrothymine	2.185		
L-Carnitine	0.173	Glutaconic acid	-0.210	Methylmalonic acid	2.404	Methylmalonic acid	1.794		
Malonate	0.168	Trimethylamine	0.195	Dimethylamine	2.354	Dimethylamine	1.752		
Choline chloride	0.162	D-Fructose	-0.192	Pyroglutamic acid	2.125	Pyroglutamic acid	1.647		
Acetoacetate	0.161	Spermidine	0.189	L-Proline	2.024	GMP	1.594		
L-Proline	0.160	Dimethylglycine	0.188	GMP	1.941	N-(2- Furoyl)Glycine	1.565		
Isovaleric acid	-0.156	Methylguanidine	0.178	1,3- Diaminopropane	1.929	L-Proline	1.552		
Acetaminophen	0.155	Lactate	-0.178	D-Fucose	1.894	Adipic acid	1.521		
Taurine	0.154	L-Histidine	-0.177	Dimethyl sulphone	1.696	1,3- Diaminopropane	1.467		
Ethanolamine	0.153	2-Deoxycytidine	0.174	L-Glutamine	1.610	UMP	1.454		
Acetyl-L-carnitine	0.148	N-Acetylglycine	0.169	2-Deoxycytidine	1.557	Acetaminophen	1.446		
Quinolinic acid	0.148	Trans-4-Hydroxy- L-Proline	0.168	D-Fructose	1.482	Dimethyl sulphone	1.416		
Pyroglutamic acid	0.144	Threonic acid	-0.167	Ethanolamine	1.427	D-Fucose	1.411		
Adipic acid	0.142	L-Glutamine	-0.160	Taurine	1.426	L-Alanine	1.380		
L-Cystine	0.141	1- Methylhydantoin	0.151	Myo-Inositol	1.390	Lactose	1.337		
L-Asparagine	0.140	Levoglucosan	0.146	TMAO	1.370	Isovaleric acid	1.335		
Lactose	-0.140	Phenethylamine	0.146	D-Galactose	1.367	L-Glutamine	1.302		
TMAO	0.139	Dihydrothymine	-0.141	N-Acetylglycine	1.330	Pimelic acid	1.255		
Glycerophospho- choline	0.139	Dimethyl sulphone	0.140	UMP	1.329	L-Cystine	1.250		
D-Galactose	-0.139	D-Gluconic acid	0.134	Levoglucosan	1.315	D-Galactose	1.240		

C. PLS-DA metabolite coefficients for each diet

Baseline		Control		Glycine			
Dihydrothymine	97.8	Methylguanidine	71.91	Dihydrothymine	100.0		
Dimethyl sulphone	88.1	Dimethyl sulphone	68.06	Dimethyl sulphone	79.8		
Dimethylamine	84.4	Trimethylamine	60.96	N-(2-Furoyl)Glycine	75.8		
N-(2-Furoyl)Glycine	71.7	Dimethylglycine	56.15	GMP	74.1		
1,3-Diaminopropane	71.7	2-Oxoglutarate	52.12	Pimelic acid	68.3		

Methylmalonic acid	71.2	D-Fructose	50.13	Dimethylamine	68.0
GMP	70.9	1-Methylhydantoin	49.08	Methylmalonic acid	64.8
Pimelic acid	63.7	D-Glucose	46.90	1,3-Diaminopropane	60.6
L-Proline	61.5	Dimethylamine	46.77	Adipic acid	59.4
UMP	60.4	L-Valine	44.56	L-Tryptophane	54.8
Adipic acid	57.1	L-Carnosine	44.30	4-Aminohippuric acid	54.2
Pyroglutamic acid	54.9	3-Hydroxyphenylacetic acid	42.61	CMP	53.5
L-Alanine	54.2	L-Asparagine	40.27	UMP	51.3
CMP	52.4	Pimelic acid	38.89	L-Valine	50.6
D-Fructose	51.6	D-Maltose	38.31	L-Proline	50.6
4-Aminohippuric acid	49.4	Beta-Alanine	37.77	D-Glucose	48.3
Acetaminophen	48.7	Trans-Acotinic acid	37.32	Trimethylamine	48.2
L-Valine	47.3	L-Proline	36.98	2-Hydroxyphenylacetic acid	48.0
Trimethylamine	47.1	D-Glucose-6-Phosphate	33.69	Acetaminophen	47.8
D-Fucose	46.3	Pyroglutamic acid	32.96	D-Fructose	47.6

Loadings are the weight of each metabolite and are expressed as positive or negative number depending on its direction from the PC. Coefficients are the weighted sum of the PLS-DA regression for each group. PCA, principle component analysis; PLS-DA, partial least squares discriminant analysis, VIP, variable importance in projection; PC, principle component; C, component. Refer to Table S1 for full names of metabolites.

Table S4. The list of 65 potentially important urinary metabolites according to their frequency in the top 20% in univariate and multivariate analyses.

Metabolite	Count	Univariate	PC 1	PC 2	VIP 1	VIP 2	Coeff. Baseline	Coeff. Control	Coeff. Glycine
Dimethyl sulphone	7	1		1	1	1	1	1	1
L-Proline	7	1	1		1	1	1	1	1
D-Fructose	6	1		1	1		1	1	1
Dimethylamine	6	1			1	1	1	1	1
Pyroglutamic Acid	6	1	1		1	1	1	1	
1,3-Diaminopropane	5	1			1	1	1		1
Dihydrothymine	5			1	1	1	1		1
Methylmalonic Acid	5	1			1	1	1		1
Pimelic Acid	5	1				1	1	1	1
Trimethylamine	5	1		1			1	1	1
Acetaminophen	4		1			1	1		1
Adipic Acid	4		1			1	1		1
GMP	4				1	1	1		1
L-Valine	4	1					1	1	1
UMP	4				1	1	1		1
1-Methylhydantoin	3	1		1				1	
СМР	3	1					1		1
D-Fucose	3				1	1	1		
D-Galactose	3		1		1	1			
D-Glucose	3	1						1	1
Dimethylglycine	3	1		1				1	
Isovaleric Acid	3	1	1			1			
L-Asparagine	3	1	1					1	

L-Cystine	3	1	1			1			
L-Glutamine	3			1	1	1			
Methylguanidine	3	1		1				1	
N-(2-Furoyl)Glycine	3					1	1		1
Taurine	3	1	1		1				
ТМАО	3	1	1		1				
2-Deoxycytidine	2			1	1				
2-Oxoglutarate	2	1						1	
3-Hydroxyphenyl Acetic Acid	2	1						1	
4-Aminohippuric Acid	2						1		1
Acetoacetate	2	1	1						
β-Alanine	2	1						1	
D-Maltose	2	1						1	
Ethanolamine	2		1		1				
Lactose	2		1			1			
L-Alanine	2					1	1		
L-Carnitine	2	1	1						
L-Carnosine	2	1						1	
Levoglucosan	2			1	1				
Malonate	2	1	1						
N-Acetylglycine	2			1	1				
Trans-Acotinic Acid	2	1						1	
2-Hydroxyphenyl Acetic Acid	1								1
Choline Chloride	1		1						
D-Gluconic Acid									
D Glacome Acia	1			1					
D-Glucose-6-Phosphate	1		·	1				1	
D-Glucose-6-Phosphate Glutaconic Acid	1 1 1		·	1 1				1	
D-Glucose-6-Phosphate Glutaconic Acid Glycerophosphocholine	1 1 1 1		1	1 1				1	
D-Glucose-6-Phosphate Glutaconic Acid Glycerophosphocholine Hypotaurine	1 1 1 1	1	1	1 1				1	
D-Glucose-6-Phosphate Glutaconic Acid Glycerophosphocholine Hypotaurine Lactate	1 1 1 1 1	1	1	1 1 1				1	
D-Glucose-6-Phosphate Glutaconic Acid Glycerophosphocholine Hypotaurine Lactate L-Histidine	1 1 1 1 1 1	1	1	1 1 1 1				1	
D-Glucose-6-Phosphate Glutaconic Acid Glycerophosphocholine Hypotaurine Lactate L-Histidine L-Tryptophane	1 1 1 1 1 1 1	1	1	1 1 1 1				1	1
D-Glucose-6-Phosphate Glutaconic Acid Glycerophosphocholine Hypotaurine Lactate L-Histidine L-Tryptophane Myo-Inositol	1 1 1 1 1 1 1 1	1	1	1 1 1 1	1			1	1
D-Glucose-6-Phosphate Glutaconic Acid Glycerophosphocholine Hypotaurine Lactate L-Histidine L-Tryptophane Myo-Inositol O-AcetyI-L-Carnitine	1 1 1 1 1 1 1 1 1	1	1	1 1 1 1	1			1	1
D-Glucose-6-Phosphate Glutaconic Acid Glycerophosphocholine Hypotaurine Lactate L-Histidine L-Tryptophane Myo-Inositol O-AcetyI-L-Carnitine Pantothenic Acid	1 1 1 1 1 1 1 1 1	1	1 1 1	1 1 1 1	1			1	1
D-Glucose-6-Phosphate Glutaconic Acid Glycerophosphocholine Hypotaurine Lactate L-Histidine L-Tryptophane Myo-Inositol O-AcetyI-L-Carnitine Pantothenic Acid Phenethylamine	1 1 1 1 1 1 1 1 1	1	1 1 1	1 1 1 1	1			1	1
D-Glucose-6-Phosphate Glutaconic Acid Glycerophosphocholine Hypotaurine Lactate L-Histidine L-Tryptophane Myo-Inositol O-AcetyI-L-Carnitine Pantothenic Acid Phenethylamine Phosphocholine	1 1 1 1 1 1 1 1 1	1	1 1 1	1 1 1 1	1			1	1
D-Glucose-6-Phosphate Glutaconic Acid Glycerophosphocholine Hypotaurine Lactate L-Histidine L-Tryptophane Myo-Inositol O-AcetyI-L-Carnitine Pantothenic Acid Phenethylamine Phosphocholine Quinolinic Acid	1 1 1 1 1 1 1 1 1 1	1	1 1 1	1 1 1 1	1			1	1
D-Glucose-6-Phosphate Glutaconic Acid Glycerophosphocholine Hypotaurine Lactate L-Histidine L-Tryptophane Myo-Inositol O-AcetyI-L-Carnitine Pantothenic Acid Phenethylamine Phosphocholine Quinolinic Acid Spermidine	1 1 1 1 1 1 1 1 1 1 1	1	1 1 1	1 1 1 1 1	1			1	1
D-Glucose-6-Phosphate Glutaconic Acid Glycerophosphocholine Hypotaurine Lactate L-Histidine L-Tryptophane Myo-Inositol O-AcetyI-L-Carnitine Pantothenic Acid Phenethylamine Phosphocholine Quinolinic Acid Spermidine Threonic Acid	1 1 1 1 1 1 1 1 1 1 1	1	1 1 1	1 1 1 1 1 1 1	1			1	1
D-Glucose-6-Phosphate Glutaconic Acid Glycerophosphocholine Hypotaurine Lactate L-Histidine L-Tryptophane Myo-Inositol O-AcetyI-L-Carnitine Pantothenic Acid Phenethylamine Phosphocholine Quinolinic Acid Spermidine Threonic Acid trans-4-Hydroxy-L-Proline	1 1 1 1 1 1 1 1 1 1 1 1	1	1 1 1	1 1 1 1 1 1 1 1	1			1	1

Refer to Table S1 for full names of metabolites.

Basel	ine Di	et		Control Diet		Glycir	ine Diet				
Metabolite	n	Mean	SD	Metabolite	n	Mean	SD	Metabolite	n	Mean	SD
D-Glucose	10	23.96	2.97	D-Glucose	10	25.02	5.38	D-Glucose	10	23.41	3.01
L-Glutamine	10	4.89	0.60	L-Glutamine	10	4.93	0.51	L-Glutamine	10	4.68	0.40
Lactate	10	3.54	0.63	Lactate	10	3.66	0.43	Lactate	10	3.78	0.49
Taurine	10	2.74	1.02	Taurine	10	3.50	0.97	Taurine	10	2.89	0.87
Guanidinoacetic Acid	10	2.72	0.61	Guanidinoacetic Acid	10	2.90	0.79	Guanidinoacetic Acid	10	2.61	0.47
D-Mannose	10	2.58	0.33	D-Mannose	10	2.43	0.31	G-6-P	10	2.46	0.18
G-6-P	10	2.48	0.26	D-Glucuronic Acid	10	2.38	0.48	D-Mannose	10	2.37	0.34
Glyceric Acid	10	2.33	0.36	G-6-P	10	2.31	0.18	D-Glucuronic Acid	10	2.21	0.41
D-Glucuronic Acid	10	2.27	0.36	Glyceric Acid	9	2.28	0.40	Glyceric Acid	10	2.16	0.42
D-Galactose	9	1.96	0.27	D-Galactose	7	1.98	0.33	D-Galactose	10	1.79	0.28
L-Carnitine	10	1.76	0.43	L-Alanine	10	1.74	0.22	L-Alanine	10	1.74	0.26
L-Alanine	10	1.74	0.23	L-Carnitine	10	1.69	0.59	L-Carnitine	10	1.64	0.46
D-Maltose	10	1.61	0.17	D-Maltose	9	1.55	0.19	D-Maltose	10	1.58	0.12
L-Glycine	10	1.51	0.31	L-Glycine	10	1.48	0.47	L-Cystine	10	1.48	0.17
L-Cystine	10	1.50	0.14	L-Cystine	10	1.47	0.22	L-Glycine	10	1.40	0.32
D-Fructose	7	1.23	0.42	D-Fructose	7	1.47	0.39	L-Proline	10	1.25	0.19
L-Proline	10	1.20	0.16	Ethanolamine	10	1.24	0.27	Ethanolamine	10	1.19	0.21
Ethanolamine	10	1.14	0.14	L-Proline	10	1.24	0.29	Creatinine	10	1.10	0.18
Creatinine	10	1.10	0.19	L-Lysine	6	1.12	0.22	D-Sorbitol	8	1.08	0.17
D-Sorbitol	9	1.02	0.20	Creatinine	10	1.09	0.13	D-Fructose	6	1.07	0.41
Galactitol	10	0.98	0.19	D-Sorbitol	10	1.05	0.23	Galactitol	10	0.96	0.17
Betaine	9	0.93	0.21	Galactitol	10	1.03	0.22	Betaine	10	0.94	0.27
L-Glutamic Acid	9	0.89	0.10	Betaine	8	0.99	0.32	Phosphocholine	8	0.91	0.11
Threitol	10	0.88	0.11	Phosphocholine	9	0.95	0.22	Ascorbic Acid	10	0.87	0.14
Ascorbic Acid	10	0.87	0.11	Ascorbic Acid	10	0.93	0.18	L-Lysine	7	0.87	0.19
Phosphocholine	10	0.87	0.20	Threitol	9	0.91	0.16	Threitol	9	0.86	0.14
α-Aminobutyric Acid	10	0.87	0.10	α-Aminobutyric Acid	10	0.78	0.10	α-Aminobutyric Acid	10	0.78	0.11
D-Gluconic Acid	8	0.75	0.11	D-Gluconic Acid	8	0.76	0.13	Lactose	10	0.78	0.14
L-Leucine	10	0.72	0.17	Lactose	10	0.72	0.11	L-Leucine	10	0.68	0.07
Lactose	10	0.69	0.11	L-Leucine	10	0.67	0.06	L-Threonine	10	0.59	0.11
Hypotaurine	10	0.57	0.03	Hypotaurine	10	0.57	0.05	Hypotaurine	10	0.55	0.05
Citrate	10	0.54	0.11	N-Acetylglycine	10	0.50	0.04	D-Gluconic Acid	7	0.53	0.38
N-Acetylglycine	10	0.49	0.04	L-Threonine	8	0.44	0.25	N-Acetylglycine	10	0.48	0.03
Cadaverine	9	0.48	0.18	Pantothenic Acid	10	0.43	0.03	Citrate	8	0.46	0.26
Pantothenic Acid	10	0.45	0.03	Choline Chloride	10	0.42	0.21	Choline Chloride	10	0.45	0.28
L-Threonine	8	0.44	0.26	Citrate	8	0.42	0.23	Pantothenic Acid	10	0.43	0.03
Propylene Glycol	10	0.41	0.04	Propylene Glycol	10	0.40	0.07	Propylene Glycol	10	0.41	0.04

Table S5. Relative concentration of the 48 metabolites identified in the serum of captive cheetahs at baseline,on a control and glycine supplemented diet (n=10).

L-Valine	10	0.39	0.13	L-Valine	10	0.36	0.06	L-Valine	10	0.40	0.07
L-Lysine	4	0.37	0.50	Cadaverine	6	0.32	0.29	Cadaverine	8	0.40	0.22
Propionate	8	0.36	0.19	Creatine	7	0.32	0.23	L-Tyrosine	10	0.31	0.03
Choline Chloride	10	0.29	0.05	L-Tyrosine	9	0.28	0.10	L-Methionine	5	0.26	0.28
Phenethylamine	6	0.26	0.23	L-Methionine	5	0.27	0.29	Creatine	5	0.26	0.31
L-Tyrosine	8	0.25	0.13	1,3-DAP	5	0.26	0.28	Phenethylamine	6	0.26	0.22
Creatine	4	0.24	0.37	L-Glutamic Acid	3	0.26	0.42	L-Glutamic Acid	2	0.22	0.46
α-Hydroxyisobutyric Acid	9	0.21	0.11	Propionate	6	0.25	0.21	1,3-DAP	4	0.20	0.26
1,3-DAP	3	0.14	0.23	α-Hydroxy- isobutyric Acid	9	0.24	0.11	α-Hydroxy- isobutyric Acid	9	0.20	0.09
L-Methionine	2	0.10	0.21	Phenethylamine	5	0.21	0.22	Propionate	3	0.13	0.21
O-Acetyl-L-Carnitine	9	0.08	0.03	O-Acetyl-L- Carnitine	9	0.08	0.03	O-Acetyl-L-Carnitine	9	0.08	0.03

Data are expressed as mean and SD. n indicates the number of samples where the metabolite was detected. Relative concentration is an arbitrary unit. Refer to Table S1 for full names of metabolites.

Table S6. The list of 32 potentially important serum metabolites according to their frequency in the top 20% in univariate and multivariate analyses.

Metabolite	Count	Univariate	PC 1	PC 2	VIP 1	VIP 2	Coeff. Baseline	Coeff. Control	Coeff. Glycine
L-Glutamic acid	6	1			1	1	1	1	1
L-Threonine	5			1	1	1		1	1
α-Aminobutyric Acid	4	1			1	1	1		
D-Glucose-6-Phosphate	4					1	1	1	1
Ethanolamine	4		1			1		1	1
L-Methionine	4				1	1	1		1
Propionate	4				1	1	1		1
Choline Chloride	3				1	1	1		
D-Mannose	3				1	1	1		
L-Lysine	3			1	1		1		
L-Valine	3			1				1	1
Taurine	3						1	1	1
Cadaverine	2			1			1		
D-Galactose	2							1	1
D-GlucuronicAcid	2		1					1	
D-Maltose	2							1	1
Lactate	2			1	1				
Lactose	2				1	1			
Ascorbic Acid	1							1	
Creatine	1			1					
Creatinine	1			1					
D-Gluconic Acid	1			1					
D-Glucose	1		1						
Guanidinoacetic Acid	1		1						
Hypotaurine	1		1						
L-Carnitine	1		1						
L-Cystine	1		1						
L-Glutamine	1			1					

L-Glycine	1	1	
N-Acetylglycine	1	1	
Phenethylamine	1		1
Propylene Glycol	1	1	

Refer to Table S1 for full names of metabolites.

Group	Number	Sex	Age (y)	Siblings
	1	Male	5	*
2	2	Male	5	*
2	3	Male	5	*
	4	Female	3	#
	5	Female	5	*
	6	Male	4	No
4	7	Female	3	#
I	8	Female	3	#
	9	Female	2	+
	10	Male	2	+

Table S7. Description of cheetahs included in the study (n=10)

Age is the age of the individual at the end of the sample collection period. Matching symbols (*, #, +) indicate individuals that are siblings.



Supplementary Figures

Figure S1. PLS-DA classification for urine (left) and serum (urine) of captive cheetahs using different numbers of components where the leave-one-out validation was used. *None of the components represent a good classifier. Accuracy, predicted residual error sum of squares (PRESS); R2, sum of squares; Q2, cross-validated R2.*

APPENDIX 2

Statistical analysis of the urine and serum metabolites of the n=6 group: Methods and Results

The following analyses were conducted:

- The identification and quantification of the metabolites present in the urine and serum of 6 (Group 1) captive cheetahs at baseline, on a control diet and on a glycine supplemented diet.
- The same process was followed as described in Chapter 2 "Data analysis and Statistical analysis". The most notable difference in this process was the data cleaning and normalisation (auto scaling) which was performed in MetaboAnalyst 5.0. Data cleaning removes metabolites that are present in less than 50% of all three groups in order to reduce the number of outlying metabolites that may affect the analysis. Auto scaling makes use of the following equation to calculate the scaled value for a single metabolite of a sample: scaled value = (X₂- X₁)/SD₁ where X₂ is an individual sample metabolite concentration, X₁ is the metabolite mean for all samples, SD₁ is the metabolite standard deviation for all samples. As the sample size was different to the n=10 analysis, this would lead to different metabolite means and standard deviations, resulting in different scaled values which would then undergo the respective univariate and multivariate statistical analyses.
- Univariate (one way repeated measures ANOVA) and multivariate (PCA and PLS-DA) analyses were performed and the metabolites of interest were identified as described in Chapter 2 *"Identifying metabolites of interest"*.
- The urine and serum PCA and PLS-DA plots were compared between the combined n=10 group (from Chapters 4 and 5) and the split n=6 group. The identified metabolites of interest were evaluated to determine whether this process would lead to different overall conclusions.

The results of the additional analyses are described below.

Urine metabolites

After cleaning, there were 117 identified urinary metabolites remaining compared to the 112 metabolites in the n=10 analysis. MetaboAnalyst identified 10 significant metabolites with a one-way repeated measures ANOVA after FDR adjustment, compared to the 31 significant metabolites identified previously. These 10 metabolites were dimethyl sulphone, α -ketoglutaric acid, methylguanidine, L-leucine, trimethylamine, dimethylamine, dimethylglycine, D-fructose, pimelic acid and TMAO (Table S8). L-leucine was the only metabolite not previously identified in the n=10 group analysis.

		Adjusted	Tu	key-Kramer p-val	ue
Metabolite	F-value	p-value	Baseline- Control	Baseline- Glycine	Control- Glycine
Dimethyl sulphone	72.78	0.0001	<0.001 (C>B)	NS	<0.001 (C>G)
α-ketoglutaric acid	54.72	0.0002	<0.01 (C>B)	NS	<0.05 (C>G)
Methylguanidine	38.84	0.0007	<0.001 (C>B)	NS	<0.001 (C>G)
L-Leucine	33.52	0.0010	NS	<0.05 (G>B)	<0.001 (G>C)
Trimethylamine	17.91	0.0110	<0.001 (C>B)	NS	<0.01 (C>G)
Dimethylamine	14.09	0.0159	0.02 (C>B)	NS	NS
Dimethylglycine	13.53	0.0159	<0.01 (C>B)	NS	<0.001 (C>G)
D-Fructose	11.20	0.0370	NS	NS	<0.01 (G>C)
Pimelic acid	11.19	0.0370	NS	NS	NS
ΤΜΑΟ	9.96	0.0444	NS	NS	NS

Table S8. Significant metabolites identified by univariate analyses in the urine of captive cheetahs at baseline, on a control diet and glycine supplemented diet (n=6)

NS, not significant; TMAO, Trimethylamine oxide.

The PCA and PLS-DA plots for the baseline, control and glycine supplemented diet of the n=6 and n=10 group are shown in Figure S2. For the n=6 group, PC 1 accounts for 29.9% and PC 2 accounts for 13.2% of the variation observed. This was similar to the combined n=10 group with 27.5% (PC 1) and 9.5% (PC 2) of the variation. For the PLS-DA of the n=6 group, component 1 and 2 explained 18.2% and 17.8% of the variation, respectively. It is apparent from Figure S2 (right panel) that the spread of scores for each diet are more compact for the n=6 group compared to the n=10 group and when viewed together may indicate a slightly better separation. Although the cross validation Q2 values were still negative (data not shown) which indicates that this still has poor predictive power. When separated by diet, there is clear overlap and similar pattern between the groups - with the control diet having the greatest spread of data. There were slightly different metabolites driving the variation in the n=6 group for the PCA and PLS-DA with the top three metabolite in each analyses as follows: pantothenic acid, choline chloride, L-carnitine (PC 1 loadings); L-leucine, dimethyl sulphone, methylguanidine (PC 2 loadings); ethanolamine, L-proline, dimethylamine (VIP 1); ethanolamine, allantoin, L-proline (VIP 2); ethanolamine, dimethylamine, allantoin (baseline coefficient); dimethyl sulphone, methylguanidine, dimethylglycine (control coefficient); dimethyl sulphone, L-leucine, dimethylamine (glycine coefficient). This slight difference in individual metabolites was expected because of the difference in scores due to the different normalisation values and sample size compared to the n=10 group.

Seven metabolites of interest were identified in the top 20% in 5 or more analyses – namely L-leucine, D-fructose, spermidine, dimethyl sulphone, dimethylamine, L- proline and guanosine monophosphate (GMP). Fructose, dimethyl sulphone, dimethylamine and L-proline were identified previously as metabolites of interest and clearly highlighted and discussed in Chapter 4 and 6. Spermidine was not previously highlighted to the same extent, although it was discussed in Chapter 4 and 6 in relation to



Figure S2. A two dimensional PCA (left) and PLS-DA (right) scores plot of the metabolites identified in the urine of captive cheetahs at baseline (top panel), on the control (middle panel) and glycine supplemented (bottom panel) diets. The combined n=10 group is reflected in blue and the split n=6 group is reflected in red. *Each point represents an individual cheetah with a corresponding 95% confidence ellipse for each group.*

1,3-diaminopropane as a polyamine in the "methionine associated pathway". The purine ribonucleotide, GMP, was previously identified in 4 urinary analyses (Table S4), but was not discussed. However, it may have a very close link to the pyrimidines, dihydrothymine and CMP, which were highlighted and discussed in Chapter 4 and 6. This would likely provide a further support for the role that glycine plays in purine synthesis and, therefore, DNA and RNA synthesis which was discussed in the aforementioned chapters. The essential amino acid L-leucine was not identified in any of the previous analyses, likely due to its fairly constant urinary concentration across the combined n=10 group (Table S2). However, in the n=6 group there was a significantly higher concentration of L-leucine in the glycine supplemented diet compared to the baseline and control diet. L-leucine is closely linked to protein synthesis (through activation of mTOR) and the activation of glutamate dehydrogenase which catalyses the inter-conversion of glutamic acid to α -ketoglutaric acid (Wu, 2009). Although it was not specifically discussed, it has a close connection to the discussion of glutamic acid as well as energy production through its role as a ketogenic amino acid contributing towards acetyl-CoA and acetoacetate.

Serum metabolites

After cleaning, there were 50 identified serum metabolites remaining compared to the 48 metabolites in the previous n=10 analysis. MetaboAnalyst did not identify any significant metabolites with a one-way repeated measures ANOVA after FDR adjustment, this was similar to the previous analysis. However, L-glutamine (F = 10.09; raw p = 0.004; adjusted p = 0.188), L-glutamic acid (F = 7.26; raw p = 0.011; adjusted p = 0.259), L-valine (F = 6.02; raw p = 0.019; adjusted p = 0.259), D-glucuronic acid (F = 5.69, raw p = 0.022; adjusted p = 0.259), L-carnitine (F = 5.14; raw p = 0.029; adjusted p = 0.273) and lactate (F = 4.23; raw p = 0.047; adjusted p = 0.368) were significant prior to FDR adjustment.

The serum PCA and PLS-DA plots of the combined n=10 and split n=6 group for each diet are shown in Figure S3. For the n=6 group, PC 1 accounts for 34.1% of the variation explained while PC 2 accounts for 13.1%. Similarly, in the n=10 group PC 1 accounted for 32.3% and PC 2 accounted for 13.1%. When the diet is considered through PLS-DA, component 1 accounts for 15.5% and component 2 accounts for 27.3% of the variation – this was similar to what was found in the previous analysis with clear overlap between diets. Therefore, this PLS-DA model did not have high predictive power for the different diets. There is clear overlap for the n=6 and n=10 group for each diet with potential outliers driving the spread of data despite the apparent clustering of the data. The top three metabolites driving the variation observed in multivariate analyses of the serum are as follows: guanidinoacetic acid, L-cystine, hypotaurine (PC 1 loadings); galactitol, choline chloride, D-sorbitol (PC 2 loadings); L-glutamic acid, galactitol, D-mannose (baseline coefficient); L-valine, L-glutamine, D-



glucose-6-phosphate (control coefficient); L-valine, L-glutamic acid, L-glutamine (glycine coefficient). The slight difference in metabolites would be the same as previously described for urine.

Figure S3. A two dimensional PCA (left) and PLS-DA (right) scores plot of the metabolites identified in the serum of captive cheetahs at baseline (top panel), on the control (middle panel) and glycine supplemented (bottom panel) diets. The combined n=10 group is reflected in blue and the split n=6 group is reflected in red. *Each point represents an individual cheetah with a corresponding 95% confidence ellipse for each group.*

Seven metabolites of interest were identified in the top 20% of 5 analyses – namely L-glutamic acid, D-glucuronic acid, D-sorbitol, galactitol, L-carnitine, L-threonine and L-valine. Glutamic acid and threonine were identified previously (to the same extent) and formed a crucial part of the discussion in Chapters 5 and 6. Indeed, their respective pathways were highlighted extensively under "glutamic acid and threonine associated pathways" and were influenced by glycine supplementation or deficiency. D-glucuronic acid was previously identified to a lesser extent, although it was briefly discussed in Chapter 6 in relation to glucose metabolism. The findings of D-sorbitol and galactitol are somewhat surprising in the cheetah with the former being discussed briefly in Chapter 4 in relation to fructose metabolism as it can be used to synthesis fructose through the polyol pathway in mammals. Both are sugar alcohols which can be reduced from glucose and galactose and, therefore, have a strong link to "carbohydrate and fat metabolism" which was extensively discussed in Chapter 6 despite these specific metabolites not being discussed. Carnitine was previously highlighted in the urine analysis and therefore discussed in Chapter 4 in relation to dimethylamine and trimethylamine. It was also discussed in Chapter 6 in relation to synthesis, transport and oxidation of fatty acids, and therefore under the broader pathway of "carbohydrate and fat metabolism". The essential amino acid, L-valine, was highlighted previously in the urine (Table 7, Table 8) and, to a lesser extent, in the serum (Table 10). It was discussed in Chapter 6 as a potential source of propionyl-CoA and under the larger pathway of "threonine associated pathways". Most of the serum metabolites of interest were linked to energy metabolism either through carbohydrate or fat metabolism which is notably less diverse than what was found in the n=10 analysis.

Summary and concluding thoughts on the combined n=10 and split n=6 group analysis

It is evident from the above that using a different grouping of the samples resulted in the identification of slightly different metabolites of interest. For the urine analysis, less metabolites were identified through univariate analysis, but these metabolites had a greater difference between diets i.e. a larger effect. For the serum analysis, more metabolites were identified by univariate analysis for the n=6 group, but none were significant after FDR adjustment. In the urine and serum PCA and PLS-DA, the two groups had a clear overlap indicating a similar metabolome albeit with slightly different scores (due to the different sample size) and metabolites driving the variation. Most of the key metabolites identified were found previously either to a similar or lesser extent. Therefore, the same or similar conclusions can be drawn in relation to specific pathways specifically that of glucose and fat metabolism, and purine and pyrimidine biosynthesis.

The combination of Group 1 (n=6) and Group 2 (n=4) is a clear study limitation as they experienced different lengths of time on the respective diets and when analysed separately produced slightly different results. However, utilising the combined n=10 group with a greater statistical power allowed the identification of a greater diversity of metabolites in the urine and the well-regulated and conserved

serum – providing a better representation of our untargeted approach. Therefore, we believe that utilising the combined n=10 group is a fair approach to investigate the holistic effect of glycine supplementation in captive cheetahs and address the aims of the study. It is able to provide us with more diverse information than using the n=6 group, which is notably less powered although it has the potential for a greater effect size.



Faculty of Veterinary Science Animal Ethics Committee

5 March 2020

Approval Certificate New Application

 AEC Reference No.:
 REC231-19

 Title:
 The metabolic effects of glycine supplementation in captive cheetahs (Acinonyx jubatus).

 Researcher:
 Ms KM Van Boom

 Student's Supervisor:
 Prof ASW Tordiffe

 Dear Ms KM Van Boom,
 Frof ASW Tordiffe

The **New Application** as supported by documents received between 2020-01-15 and 2020-02-24 for your research, was approved by the Animal Ethics Committee on its quorate meeting of 2020-02-24. Please note the following about your ethics approval:

1. The use of species is approved:

Species	Number			
Cheetah	16			
Samples				
Semen (depending on number of males)	24 (0.5-1ml)			
Serum	48 (20 ml each)			
Skeletal muscle (Triceps brachii)	48 (40-100mg)			
Urine	48 (5 ml each)			

- 2. Ethics Approval is valid for 1 year and needs to be renewed annually by 2021-03-05.
- 3. Please remember to use your protocol number (REC231-19) on any documents or correspondence with the AEC regarding your research.
- Please note that the AEC may ask further questions, seek additional information, require further modification, monitor the conduct of your research, or suspend or withdraw ethics approval.
- 5.

Ethics approval is subject to the following:

The ethics approval is conditional on the research being conducted as stipulated by the details
of all documents submitted to the Committee. In the event that a further need arises to change
who the investigators are, the methods or any other aspect, such changes must be submitted
as an Amendment for approval by the Committee.

We wish you the best with your research.

Yours sincerely

Prof Naidoo

Prot A Naidoo CHAIRMAN: UP-Animal Ethics Committee



Faculty of Veterinary Science Animal Ethics Committee

11 May 2020

Approval Certificate Amendment1

AEC Reference No.:	REC231-19
Title:	The metabolic effects of glycine supplementation in captive cheetahs
	(Acinonyx jubatus).
Researcher:	Ms KM Van Boom
Student's Supervisor:	Prof ASW Tordiffe

Dear Ms KM Van Boom,

The **Amendment** as supported by documents received between 2020-03-12 and 2020-05-04 for your research, was approved by the Animal Ethics Committee on its quorate meeting of 2020-05-04.

Please note the following about your ethics approval has been granted:

1. The second venue will be Cheetah Experience in Bloemfontein (1 Maluti Avenue, Bloemfontein, Free State).

The study will be replicated at a) Cango Wildlife Ranch and b) Cheetah Experience. (The animal numbers will not change)

- 2. Please remember to use your protocol number (REC231-19) on any documents or correspondence with the AEC regarding your research.
- 3. Please note that the AEC may ask further questions, seek additional information, require further modification, monitor the conduct of your research, or suspend or withdraw ethics approval.

Ethics approval is subject to the following:

• The ethics approval is conditional on the research being conducted as stipulated by the details of all documents submitted to the Committee. In the event that a further need arises to change who the investigators are, the methods or any other aspect, such changes must be submitted as an Amendment for approval by the Committee.

We wish you the best with your research. Yours sincerely

র্দের্চা V Naidoo CHAIRMAN: UP-Animal Ethics Committee



Faculty of Veterinary Science Animal Ethics Committee

11 May 2020

Approval Certificate Amendment 2

AEC Reference No.:	REC231-19
Title:	The metabolic effects of glycine supplementation in captive cheetahs
	(Acinonyx jubatus).
Researcher:	Ms KM Van Boom
Student's Supervisor:	Prof ASW Tordiffe

Dear Ms KM Van Boom,

The **Amendment** as supported by documents received between 2020-04-14 and 2020-05-04 for your research, was approved by the Animal Ethics Committee on its quorate meeting of 2020-05-04.

Please note the following about your ethics approval:

1. The use of species is approved:

Species	Number Available
Cheetah	
Samples Semen (depending on number of males)	24 (0.5-1ml)
Serum	48 (20 ml each)
Skeletal muscle (Triceps brachii)	48 (40-100mg)
Urine	48 (5 ml each)
Additional samples approved: Gastric mucosal biopsies (typically 2mm x 2mm x 2mm) will be collected from each animal	9 per animal (three from each area in the stomach). Total 144
This will be done by means of a 1.5m endoscope (Stortz, Germany).	

- 2. Please remember to use your protocol number (REC231-19) on any documents or correspondence with the AEC regarding your research.
- 3. Please note that the AEC may ask further questions, seek additional information, require further modification, monitor the conduct of your research, or suspend or withdraw ethics approval.

Ethics approval is subject to the following:

• The ethics approval is conditional on the research being conducted as stipulated by the details of all documents submitted to the Committee. In the event that a further need arises to change who the investigators are, the methods or any other aspect, such changes must be submitted as an Amendment for approval by the Committee.

We wish you the best with your research. Yours sincerely

For V Naidoo CHAIRMAN: UP-Animal Ethics Committee

Room 6-13, Arnold Theiler Building, Onderstepoort Private Bag X04, Onderstepoort 0110, South Africa Tel +27 12 529 8483 Fax +27 12 529 8321 Email aec@up.ac.za www.up.ac.za

Fakulteit Veeartsenykunde Lefapha la Diseanse tša Bongakadiruiwa